Changes for the Better



SOFTWARE MANUAL





FX-PCS/WIN-E

Software Manual

Manual number	:	JY992D66501
Manual revision	:	F
Date	:	November 2008

Foreword

- This manual contains text, diagrams and explanations which will guide the reader in the correct installation and operation of the FX-PCS/WIN-E.
- Before attempting to install or use the FX-PCS/WIN-E this manual should be read and understood.
- If in doubt at any stage of the installation of the FX-PCS-/WIN-E always consult a professional electrical engineer who is qualified and trained to the local and national standards which apply to the installation site.
- If in doubt about the operation or use of the FX-PCS/WIN-E please consult the nearest Mitsubishi Electric distributor.
- This manual is subject to change without notice.

Registration

- Microsoft[®] Windows[®], MS, MS-Windows[®] 3.1, Windows[®] 95, Windows[®] 98, Windows[®] Millennium Edition, Windows NT[®] 4.0 Workstation, Windows[®] 2000 and Windows[®] XP are either registered trademarks or trademarks of Microsoft Corporation in the United States and/or other countries.
- The company name and the product name to be described in this manual are the registered trademarks or trademarks of each company.



FAX BACK

Mitsubishi has a world wide reputation for its efforts in continually developing and pushing back the frontiers of industrial automation. What is sometimes overlooked by the user is the care and attention to detail that is taken with the documentation. However, to continue this process of improvement, the comments of the Mitsubishi users are always welcomed. This page has been designed for you, the reader, to fill in your comments and fax them back to us. We look forward to hearing from you.

Fax numbers:			Your name:		
Mitsubishi Electric.					
America (01) 847-478-2253		Your co	ompany:		
Australia	(02) 638-7072				
Germany	(0 21 02) 4 86-1 12	Your lo	cation:		
Spain	(34) 93-589-1579				
United Kingdom	(01707) 278-695				
Please tick the box	of your choice				
What condition did	the manual arrive in?	□Good	□Minor damage	□Unusable	
Will you be using a	folder to store the manual	? □Yes	□No		
What do you think	to the manual presentation	?□Tidy	□Unfriendly		
Are the explanation	ns understandable?	□Yes	□Not too bad	□Unusable	
-	was most difficult to under				
Are there any diag	rams which are not clear?	□Yes	□No		
If so,which:					
What do you think	to the manual layout?	□Good	□Not too bad	□Unhelpful	
	ou would like to see improv				
possible please ide	e information you required entify your experience:		-		
Do you have any c	omments in general about	the Mitsubi	shi manuals?		
	ng the time to fill out this qu				



and this manual easy to use.



Guidelines for the Safety of the User and Protection of the Programmable Controller

This manual provides information for the use of the FX Series Communication Unit. The manual has been written to be used by trained and competent personnel. The definition of such a person or persons is as follows:

- a) Any engineer who is responsible for the planning, design and construction of automatic equipment using the product associated with this manual should be of a competent nature, trained and qualified to the local and national standards required to fulfill that role. These engineers should be fully aware of all aspects of safety with regards to automated equipment.
- b) Any commissioning or service engineer must be of a competent nature, trained and qualified to the local and national standards required to fulfill that job. These engineers should also be trained in the use and maintenance of the completed product. This includes being completely familiar with all associated documentation for the applicable product. All maintenance should be carried out in accordance with established safety practices.
- c) All operators of the completed equipment (see Note) should be trained to use this product in a safe manner in compliance to established safety practices. The operators should also be familiar with documentation which is connected with the actual operation of the completed equipment.
- Note: the term 'completed equipment' refers to a third party constructed device which contains or uses the product associated with this manual.

Notes on the Symbols Used in this Manual

At various times throughout this manual certain symbols will be used to highlight points of information which are intended to ensure the users personal safety and protect the integrity of equipment. Whenever any of the following symbols are encountered its associated note must be read and understood. Each of the symbols used will now be listed with a brief description of its meaning.

Hardware Warnings

1) Indicates that the identified danger WILL cause physical and property damage.



2)Indicates that the identified danger could **POSSIBLY** cause physical and property damage.



3)Indicates a point of further interest or further explanation.

Software Warnings



2)Indicates a special point which the user of the associate software element should be aware of.

1)Indicates special care must be taken when using this element of software.



3)Indicates a point of interest or further explanation.



- Under no circumstances will Mitsubishi Electric be liable responsible for any consequential damage that may arise as a result of the installation or use of this equipment.
- All examples and diagrams shown in this manual are intended only as an aid to understanding the text, not to guarantee operation. Mitsubishi Electric will accept no responsibility for actual use of the product based on these illustrative examples.
- Please contact a Mitsubishi Electric distributor for more information concerning applications in life critical situations or high reliability.

Abbreviations

The following definitions or abbreviations will be used throughout this manual.

- The FX-PCS/WIN-E software will be referred to as the FX-PCS/WIN-E or FXGP/WIN-E.
- The FX0, FX0S, FX0N, FX, FX2C, FX1S, FX1N, FX2N and FX2NC Series PLC may be referred to as the FX Series PLC or FX Series PLC.
- Input/Output may be referred to as I/O.
- Personal Computer may be referred to as PC
- Microsoft[®] Windows[®], Windows[®] 3.1, Windows[®] 95, Windows[®] 98, Windows[®] Millennium Edition, Windows NT[®] 4.0 Workstation, Windows[®] 2000 and Windows[®] XP may be referred to generically as Windows.

CONTENTS

1.	Introduction	1-1
	1.1 Main Software Features	. 1-1
	1.2 Contents of Product Package and System Configuration	
	1.2.1 Contents of product package	
	1.2.2 Operation environment	
	1.3 Compatibility of Data File Types	
	1.4 Comment Control	
	1.4 Comment Control	. 1-7
2.	Installing FXGP/WIN-E	2-1
	2.1 Install Procedure (for Windows 3.1)	
	2.2 Install Procedure (for Windows 95 or after)	
		2-2
3.	Main Operation	3-1
	3.1 Starting method	
	3.1.1 Starting FXGP/WIN-E for Windows 3.1	
	3.1.2 Starting FXGP/WIN-E for Windows 95 or After	
	3.2 Initial Screen Operation	
	3.3 Main screens	
	3.4 Terminating method (Exit)	3-4
4.	Work Windows and their Functions	4-1
	4.1 Construction and Types of Windows	
	4.1.1 Construction of windows and relation between windows	
	4.1.2 Handling of opened FXGP/WIN-E softwares at a time	
	4.1.2 Manualing of opened 1 AGP/Win-L softwares at a time	
	4.1.3 Split window	
	4.1.4 Selecting a window non-too menu	
	4.3 Functions of the Comment Window Group: View \rightarrow Comment View	
	4.4 Functions of the Register Window: View \rightarrow Register View	
	4.5 Functions of the Device List Window Group: View \rightarrow	
	4.6 Functions of the Sampling Trace Window Group: PLC \rightarrow Sampling Trace	
	4.7 Functions of Device Monitor Window: Monitor/Test \rightarrow	4-11
5	Cuidanas of Main Operations	5 1
э.	Guidance of Main Operations	
	5.1 Startup/Edit/window display	
	5.2 Program	
	5.3 Ladder circuit	. 5-3
	5.4 List	. 5-4
	5.5 SFC	. 5-4
	5.6 Instruction input	
	5.7 PLC	
	5.8 Monitor	
	5.9 Comment	
	5.10 Printer	
	5.11 PC	
	5.12 Data compatibility and conversion	. 5-8
	5.13 Telephone line connection	. 5-8
	5.14 Others	

6.	Menu Bar Commands	6-1
	6.1 File Menu Functions	6-2
	6.1.1 New	6-2
	6.1.2 Open	6-3
	6.1.3 Close and Open	6-5
	6.1.4 Save	
	6.1.5 Save As	
	6.1.6 Print	
	6.1.7 Print All	
	6.1.8 Page Setup	
	6.1.9 Print Preview	
	6.1.10 Printer Setup	
	6.1.11 Displaying file use history	
	6.1.12 Exit	
	6.2 Edit Menu Functions	
	6.2.1 Undo	
	6.2.2 Cut	
	6.2.3 Copy	
	6.2.4 Paste	
	6.2.5 Delete	
	6.2.7 Line Insert	
	6.2.8 Block Select	
	6.2.9 NOP Overwrite	
	6.2.10 NOP Insert	
	6.2.11 NOP Remove	
	6.2.12 Device Name	
	6.2.13 Device Name	
	6.2.14 Coil Comment	
	6.2.15 Block Comment	
	6.2.16 Device Registration	
	6.2.17 Edit cancel	
	6.3 Tools Menu Functions	
	6.3.1 Contact	
	6.3.2 Coil	
	6.3.3 Function	
	6.3.4 Wire	-
	6.3.5 Instruction	
	6.3.6 All Clear	
	6.3.7 Convert	
	6.3.8 Default comments for special M/D	
	6.3.9 Comment settings	
	6.3.10 Setting of step of comment	
	6.4 Search Menu Functions	
	6.4.1 Go to Top, Go to End	
	6.4.2 Device Name Search	
	6.4.3 Device Search	
	6.4.4 Instruction Search	
	6.4.5 Contact/Coil Search	
	6.4.6 Go to Step	
	6.4.7 Go to State	
	6.4.8 Character Search	
	6.4.9 Character Exchange	
	6.4.10 Change Device Number	
	6.4.11 Change Bit Device	
	6.4.12 Swap Device Number	
	6.4.13 Tag Set	
	6.4.14 Tag Jump	

6.5 View Menu Functions	6-62
6.5.1 Ladder View	6-63
6.5.2 Instruction View	6-64
6.5.3 SFC View	6-65
6.5.4 Inner Ladder View	6-66
6.5.5 Comment View	6-67
6.5.6 Register View	6-71
6.5.7 Tool Bar Status Bar Function Keys	6-75
6.5.8 Palette	
6.5.9 Contact/Coil List	6-77
6.5.10 Device Used List	6-77
6.5.11 TC Setting List	6-78
6.5.12 Comment display	6-79
6.5.13 Display Setting	6-82
6.5.14 Zoom	
6.6 PLC Menu Functions	6-83
6.6.1 Transfers	
6.6.2 Register Data Transfers	
6.6.3 PLC Memory Clear	
6.6.4 Serial Setting (D8120)	
6.6.5 PLC Keyword Current or Delete	
6.6.6 Runtime Program Changes	
6.6.7 Remote Run/Stop	
6.6.8 PLC Diagnostics	
6.6.9 Sampling Trace	
6.6.10 Ports	
6.7 Monitor/Test Menu Functions	
6.7.1 Start monitor	
6.7.2 Stop Monitor	
6.7.3 Dynamic Monitor	
6.7.4 Entry Device Monitor	
6.7.5 Forced Y Output	
6.7.6 Forced On/Off	
6.7.7 Change Current Values	
6.7.8 Change Setting Values	
6.7.9 Display change of monitor data($\Box\Box\Box$)	
6.8 Option Menu Functions	
6.8.1 Program Check	
6.8.2 Parameter Settings	
6.8.3 Keyword Setting	
6.8.4 PLC's Mode Setting	
6.8.5 Serial Setting (Parameter)	
6.8.6 Printer Title	
6.8.7 Device range Setting	
6.8.8 Comment Move	
6.8.9 PLC Type Change	
6.8.10 Preferences	n-1/4
	6-125
6.8.11 EPROM Transfer	6-125 6-126
6.8.11 EPROM Transfer 6.8.12 Font	6-125 6-126 6-127
6.8.11 EPROM Transfer	6-125 6-126 6-127 6-128

7. Creating Ladder and Instruction Programs	7-1
7.1 Use of Programming Window	
7.2 Creating Ladder Programs	
7.2.1 Main operations necessary for entering ladder circuit symbols	
7.2.2 Instruction type ladder circuit entry	
7.2.3 Essentials for programming	
7.2.4 Creating ladder circuit ladder diagram	7-16
7.3 Creating Instruction List Program	7-20
7.3.1 Main operations necessary for entering instruction words	
7.3.2 Creating instruction list	7-29
8. Creating SFC Programs	8-1
8.1 Description of SFC	
8.1.1 Types of SFC	
8.1.3 Structure of SFC section	
8.2 Creating SFC Programs	
8.2.1 Main operations necessary for entering SFC symbols	
8.2.2 Cursor position and entering symbol	
8.2.3 Main items necessary for creating SFC	
8.2.4 Creating SFC	
0. Demote Meintenenes Lleing e Telenhene Line	0.1
9. Remote Maintenance Using a Telephone Line	
9.1 System Configuration and Functions	
9.1.1 Remote access to PLC	
9.1.2 Sending/receiving files	
9.1.3 Caution	
9.2 Operation Procedures	
9.2.1 Remote access to PLC 9.2.2 File transfer	
9.3 Functions on Remote Menu	
9.3.1 Connect	
9.3.2 Disconnect	
9.3.3 File transfer	
9.3.4 Environment \rightarrow Modem	
9.3.5 Environment $ ightarrow$ Logging File	
9.4 Data Setting on PLC Side for Remote Access	
Appendix A:	
PLC Device Tables	A-1
A-1: FX0/FX0s Series PLC	
A-1: TX0/TX0S Genes FLC	
A-2: FX Series PLC (Version 2.30 or earlier)	
A-4: FX (Version 3.07 or later)/FX ₂ C (All version) Series PLC	
A-5: FX1S Series PLC A-6: FX1N Series PLC	
A-7: FX2N/FX2NC Series PLC	A-11
Appendix B:	
PLC Instruction Tables	R_1
B-1: Basic Program Instruction Table	
B-2: SFC (STL) Program Instruction Table	
B-3: Applied Instruction Table	В-2

Appendiz Eri	c C: or Cord of PLCC-	-1
Appendiz File	c D: e converterD-	-1
Appendi As	c E: sociated Manuals E-	-1



Introduction	1
Installing FXGP/WIN-E	2
Main Operation	3
Work Windows and their Functions	4
Guidance of Basic Operations	5
Menu Bar Commands and their Functions	6
Creating Ladder and Instruction Programs	7
Creating SFC Programs	8
Remote Maintenance Using Telephone Line	9
Appendix A: Device Lists	Α
Appendix B: Instruction List	В
Appendix C: Error Code Tables	С
Appendix D: File Convertor	D
Appendix D: Associated Manuals	Е
	Installing FXGP/WIN-EMain OperationWork Windows and their FunctionsGuidance of Basic OperationsGuidance of Basic OperationsMenu Bar Commands and their FunctionsCreating Ladder and Instruction ProgramsCreating SFC ProgramsRemote Maintenance Using Telephone LineAppendix A: Device ListsAppendix B: Instruction ListAppendix C: Error Code TablesAppendix D: File Convertor

1	Introduction
2	Installing FXGP/WIN-E
3	Main Operation
4	Work Windows and their Functions
5	Guidance of Basic Operations
6	Menu Bar Commands and their Functions
7	Creating Ladder and Instruction Programs
8	Creating SFC Programs
9	Remote Maintenance Using Telephone Line
Α	Appendix A: Device Lists
В	Appendix B: Instruction List
С	Appendix C: Error Code Tables
D	Appendix D: File Convertor
E	Appendix D: Associated Manuals



1. Introduction

This software manual describes the outline connection of the FX-PCS/WIN-E software and the construction of the hardware system.

In addition, this manual describes the connection of various units and utilization of various the software functions, such as data file compatibility. Please read this manual before installing the software.

1.1 Main Software Features

FX-PCS/WIN-E is designed for the creation of programs used on Mitsubishi FX series PLC. This software runs on MS-Windows[®] 3.1, Windows[®] 95, Windows[®] 98, Windows[®] Millennium Edition, Windows NT[®] 4.0 Workstation, Windows[®] 2000, and Windows[®] XP.

Programming:

• Three types of programming windows

Three types of programming tools are available for simultaneous operation. They are the ladder program edit window, list program edit window, and SFC program edit window. You can enter or edit data while switching the active window between the three different methods of programing.

Selection of the operation type in creating ladder diagrams

To create ladder diagrams via the ladder diagram editing window, you can use the conventional keyboard operation method. Using this method, use function keys to enter in ladder symbols. In addition to this conventional operation method, use the mouse operation method. To support the mouse operation method, there is a floating pallet of ladder symbols, menu commands, and function key guides. In addition, use the list editing window, You can create ladder diagrams by directly entering instructions.

In addition, use the list editing window, You can create ladder diagrams by directly entering instr Various tools for selecting instructions from list

This software has various tools that facilitate entry of instructions, such as contact symbols, coil instructions, and application instructions.

Particularly for application instruction entry, the list of instructions will be displayed, and desired instructions can be selected from the list. Such interactive functionality will facilitate instruction entry operation.

Importance placed on keyboard operation

In some work places, the use of a mouse or pointing device may be difficult, however considering such a problem, we have designed this software so that most of operation can be performed using the keyboard.

Entire ladder diagram edit function

Modifying part of the diagram while looking the front and rear of the ladder block currently edited.

Splitting screen into two

Splitting the ladder diagram editing window, ladder circuit monitoring window, comment editing window, etc. into two (upper and lower windows) using the split handle at the lower section of the screen is possible. Each window can display the different sections of the document, and each screen can be scrolled individually. Use this function to display the separate sections of a document concurrently, or to monitor two sections simultaneously.

1

6

F

Monitoring, connection via telephone line, and others:

• Monitoring function

Using the ladder program edit window and SFC program edit window, directly monitor the PLC status.

• Remote access to the FX1S, FX1N, FX2N and FX2NC Series PLC through telephone line

Connect the FX1S, FX1N, FX2N and FX2NC Series PLC to the PC using the modem and telephone line. You can transfer or monitor the programs through the telephone line even if the PLC is away from the PC. This software has a telephone line connection and remote station control function, etc. In addition, if two PCs are connected to each other through the telephone line, data files can be

transferred between PCs.

Activation of two or more application programs

The FX-PCS/WIN can activate two or more application programs at the same time, therefore, the uses can display another program by switching the active window, and move or copy data between the programs using the cut and paste functions

Preview of printed documents

Before printing, preview the images of the printed documents, such as ladder diagrams, command lists, SFC diagrams, or other printable data.

1

2

3

4

1.2 Contents of Product Package and System Configuration

1.2.1 Contents of product package

The FX-PCS/WIN-E software package contains the following items:

- 1) 3.5- inch floppy disks (1.44 MB) (system disks that can activate the SW0PC- FXGP/ WIN- E system), 2 disks
- 2) This operation manual

The cable connecting the PC to the PLC and the interface unit are optional. Prepare a cable and interface unit applicable to the system, while referring to the configuration shown on the next page.

1.2.2 Operation environment

Table:1.1

Table:1.1	T	
Item	Description	
	MS-Windows [®] 3.1 (386 enhance mode) English version Microsoft [®] Windows [®] 95 ^{*1} English version Microsoft [®] Windows [®] 98 ^{*1} English version	5
os	Microsoft [®] Windows [®] Millennium Edition ^{*1} English version Microsoft [®] WindowsNT [®] 4.0 ^{*1*2} Workstation English version (Service Pack 3 or later) Microsoft [®] Windows [®] 2000 ^{*1*2} English version	6
	Microsoft [®] Windows [®] XP ^{*1*2} English version (Home Edition or professional) Operation [®] of each OS shall be assured in the PC to be used.	7
	MS-Windows [®] 3.1: 804 6SX or better one Microsoft [®] Windows [®] 95: CPU i486SX or better one	
PC main body	Microsoft [®] Windows [®] 95: CPU i4865X of better one Microsoft [®] Windows [®] 98: CPU i486DX (66 MHz) or better one Microsoft [®] Windows [®] Millennium Edition: CPU Pentium 150 MHz or better one Microsoft [®] WindowsNT [®] 4.0: CPU i486 (25 MHz) or better one	8
	Microsof [®] t Windows [®] 2000: CPU i486 (133 MHz) or better one Microsoft [®] Windows [®] XP: Pentiumn 300MHz or better one	9
Required memory	MS-Windows [®] 3.1: 8MB or more (16 MB or more is recommended.) Microsoft [®] Windows [®] 95: 8 MB or more (16 MB or more is recommended.) Microsoft [®] Windows [®] 98: 16 MB or more (32 MB or more is recommended.) Microsoft [®] Windows [®] Millennium Edition: 32 MB or more Microsoft [®] WindowsNT [®] 4.0: 16 MB or more	A
	Microsoft [®] Windows [®] 2000: 32 MB or more (64 MB or more is recommended.) Microsoft [®] Windows [®] XP: 128MB or more	В
Hard disk capacity	Free space of 6 MB or more	
Floppy disk unit	3.5-inch (2HD) floppy disk drive \times 1 unit A disk formatted as 1.44 MB shall be able to be read.	С
Display	Video display adaptor whose resolution is VGA or better	
Interface	RS-232C serial interface (COM1 to COM9 shall be able to be changed over.)	[
	Printer interface	
Printer	Printer in accordance with the OS above	
Others	Mouse or other pointing device	
Applicable PLC	FX0, FX0s, FX0N, FX (FX2), FX2c, FX1s, FX1N, FX2N and FX2Nc series PLC	

*1: Long file names cannot be used.

*2: Remote Maintenance cannot support with these operating systems.

1.2.3 System configuration

The cable connecting the PC to the PLC and the interface unit is optional. Select an appropriate cable and interface unit while referring to the following description.



- *1 When using FX1S, FX1N or FX2N PLC, PLC can connect to PC via the following expansion board - FX1S, FX1N PLC: FX1N-422-BD - FX2N PLC: FX2N-422-BD
- *2 When connecting the FX0N-232ADP or FX2NC-232ADP to FX1S, FX1N or FX2N PLC, the following expansion board need to be installed to the FX1S, FX1N or FX2N PLC.
 - FX1S, FX1N PLC: FX1N-CNV-BD FX2N PLC: FX2N-CNV-BD



1

2

3

4

5

6

1.3 Compatibility of Data File Types

Windows Data files

The programs and comments created by FXGP/WIN-E will be stored using the following file names:

Table:1.2

File name	Extension	Data stored in file	
Program file *1	PMW	Parameter, program, file register, and comments	
Comment file *2	COW	Device comments, block comments, coil comments, and device name	
Register file *1	DMW	Data register and RAM file register	
Sampling trace file	STW	Result of sampling trace	
Printer title file	PTW	Printing style	
Entry device monitor file	RMW	Content of Entry device monitor setting	
Sampling trace file Printer title file Entry device monitor file	STW PTW	Data register and RAM file register Result of sampling trace Printing style Content of Entry device monitor setting	

For these data files, the special format for the FXGP/WIN-E is used. The SW1PC-FX/EE (AT compatible machine, DOS version) and the A6GPP/PHP, cannot read these files.

- *1: Only the program and register file data will be stored in the memory of the PLC and the optional memory cassette. Other files will be stored in the peripheral devices.
- *2: The comments stored in the comment file are controlled by the peripheral devices. Refer to Sec. 1.4 "Comment Control", to control the comments to be written to the program memory of the PLC.

Storing data in DOS file

To store data in a DOS file, select File and then Save As. Program files (.PMC) and comment files (.COK) are saved. Up to sixteen characters are allowed as a file name, thus, excessive characters are deleted.

Applicable DOS application program:

• SW1PC-FXGP/ EE (designed for AT compatible machine)

С

Ε

Reading various data files

FXGP/WIN-E has a data file conversion function that converts the data files created by peripheral devices into Windows format.

Applicable software models:

• SW1PC-FXGP/EE (for AT compatible machine, DOS version)

Converted files:

- Program file (PMC)
- Data register file (DMD, DME, DMG)
- RAM file register (DMF)
- Comment file (COK)
- Sampling trace file (STA)
- Printer title file (PTL)

Unconverted files:

- Comment file (COH)
- Block comment file (COL)
- Diversion ladder circuit file (DAT)

Sequentially select File Open Save As Type and DOS Files (*.pmc) to open a file. When opened, the file will be automatically converted. Converted data can be stored as the FXGP/WIN-E file described on the previous page.

1.4 Comment Control

FXGP/WIN-E V2.00 or above has enabled storage of device comments in the parameter-set comment area of the program memory held in the PLC. The number of characters allowed in each comment written to the PLC is 16 maximum.

Comment control using FXGP/WIN-E:

If the FXGP/WIN-E is used, the peripherals can control both types of comments as the device comments; comments to be controlled by peripheral devices and comments to be written to the PLC.

These device comments can be entered using the input device comment window or comment list program edit window (refer to Sec. 6.2.13 and Sec. 6.5.5).

Setting comments to be written to the PLC:

The device comments to be controlled by the peripheral devices can be written in the parameter-set comment area of the program memory in the PLC.

The device comments can be written to the comment area until the comment area is fully loaded (the comment storage capacity is parameter set). The minimum unit for device comment storage is one comment.

- Parameter setting (see Sec. 6.8.2)
- Device comment entry and specifications (see Sec. 6.2.13)
- Setting of comments to be written to the PLC (see Sec. 6.3.9)
- Editing and displaying of device comments using device comment/name window (see Sec. 6.5.5)

Controlling programs including comments:

Programmable controller

• Reading from PLC:

The comments read from the PLC will be transferred to the device comment area of the FXGP/WIN-E. After transfer, the comments will be set as comments to be written to the PLC, and the corresponding device numbers shown in the list window will have an asterisk *.

 Writing to PLC: The device comments to be written in the PLC (comments marked with as asterisk *) will be written in the comment area of the program memory in the PLC.

File

• Reading from program file:

Before opening a file, a confirmation message will appear and prompt whether the comments in the program memory should be transferred to the device comment area of the FXGP/WIN-E.

Click "Yes" to transfer the comments to the device comment area of the FXGP/WIN-E. When clicking "No", the comments will be loaded to FXGP/WIN-E as program data, therefore, device comment area of the FXGP/WIN-E will not be subject to change. Even if comments are loaded to FXGP/WIN-E, transfer the comments into the device comment area later (refer to Sec. 7.8.8).

After clicking "Yes", the comments are transferred to the device comment area, and regarded as the comments to be written to the PLC. In addition, the corresponding device numbers displayed in the list window will be marked with an asterisk *.

• Writing in program file:

The device comments to be written to the PLC (comments marked with an asterisk *) will be written to the program memory, and then stored in the file. All the device comments are stored in the comment file (.COW).

• Importing into comment file:

No comment data will be imported into the device comments if the comments are specified as the comments to be written to the PLC (if the comments are marked with an asterisk * in the device comment/name window).

 Exporting from the comment file: All the device comments in FXGP/WIN-E will be stored in the file.

1

2

3

4

5

6

7

8

9

Α

В

С

D

F

ROM writer

Data will be read/written to/from the ROM writer in the same way as the PLC.



Exporting and Importing a comment file export:

Only a comment file can be saved (expect) with a different name. An existing comment file can be read (import) into opened a program file.

Select the comment window, and execute the following operation.

Exporting : [File (F)] \rightarrow [Export (E)]

Reading : [File (F)] \rightarrow [Import (I)]

The comment type can be selected.

If there are comment types which should not be processed in import or export, remove a check mark in such comment types.

Save as		×
File name C:\ABC.COW		ОК
File <u>t</u> itle		Cancel
File type		<u>H</u> elp
I Device comment	Block comment	
Device name	Coil comment	

When a program file is saved or read, comment files are handled together. The export and import operations described above are required only when independent operations for comment files are required. (Refer to section 6.1.2 and 6.1.4.)



2. Installing FXGP/WIN-E

This section describes the FXGP-WIN-E installation procedure. Before installing FXGP/WIN-E, install the necessary system software, such as MS-DOS and MS-Windows.

2.1 Install Procedure (for Windows 3.1)

- 1) Turn on the power of the PC and start Windows.
- 2) Insert the SW0PC- FXGP/WIN system floppy disk (1.44 MB) into the floppy disk drive.
- 3) Click Icon in the Program Manager dialog box.
- 4) Click "Run".
- 5) Click in the Command Line box, and type "A: SETUP EXE". (A= Floppy disk drive name)
- 6) Click the "OK" button.
- 7) Specify the directory where the FXGP/WIN-E should be installed. The default is C:\FXGPWIN
- 8) Follow the directions shown on the screen to install FXGP/WIN-E.



Uninstallation (for Windows 3.1):

• To delete this software (application program) from the hard disk, delete all the files from the installation directory, and FXGPWIN. INI files from the Windows directory using the file manager, etc. If program files are stored in the directory, be sure to back up the files before deleting this application program. If the program files are not deleted, it will be deleted together with the application program.

Initialize of FXGP/WIN-E:

- 1) FXGP/WIN-E is ended.
- 2) Delete the FXGP/WIN.INI files from the Windows directory.
- 3) Restart FXGP/WIN-E.



1

2

3

2.2 Install Procedure (for Windows 95 or after)



To install the FX-PCS/WIN-E component files from the 2 install disks. Note that this software cannot be run from the install (floppy) disk; this components must be installed onto hard-drive and subsequently run the software from the drive.

Note:

If the PC had been installed with FX-PCS/WIN-E, please uninstall it. If it is uninstalled before the install, the FX-PCS/WIN-E may not operate correctly. For uninstall, refer to below.

To install FX-PCS/WIN-E:

- 1) Restart Windows, and do not start-up any other applications.
- 2) Insert the SW0PC-FXGP/WIN-E bellow disk 1 into floppy disk drive.
- 3) Execute "setup32.exe."
- 4) During the FX-PCS/WIN-E Setup dialog box, click "NEXT" whenever ready thus proceeding to the next panel.
- 5) If the destination folder need to be changed for the FX-PCS/WIN-E component files, click "Browse", and use the browser to locate the appropriate destination.
- 6) Click "NEXT".
- 7) If the Program folders need to be changed for the FX-PCS/WIN-E, enter the program folder name.
- 8) Click "NEXT" to begin the installation. When the process is complete, a message will follow indicating. He successful installation of the FX-PCS/WIN-E software.
- 9) Click "OK" to finish to install.



Uninstallation:

It is possible to remove all of the FX-PCS/WIN-E component files installed on your system with the Install/Uninstall in the Control Panel.

To uninstall FX-PCS/WIN-E:

- 1) Click the "Start Menu", choose "Setting" > "Control Panel", and click it.
- 2) Double-click "Add/Remove Programs" icon.
- 3) Select "FX-PCS/WIN-E" in the "Install/Uninstall" tab, and click "Add/Remove".
- 4) Click "Yes" to begin the uninstall of the FX-PCS/WIN-E component files. When the process is complete, a FX-PCS/WIN-E is successfully uninstalled from your PC.



3. Main Operation

This section describes the main operation of FXGP/WIN-E.

Note that operation methods described in this section are not the special methods used for FXGPWIN-E, but are the common operation methods used in a Windows environment.

For an inexperienced windows operator, it is recommended that the manual for operating system should be used in conjunction with this manual.

3.1 Starting method

3.1.1 Starting FXGP/WIN-E for Windows 3.1

At the completion of install, the MELSEC-F FX application group will be created, and the FXGP/WIN-E start icon will be stored.

The figure below shows the Windows 3.1 screen.



Double- click on this icon to start FXGP/WIN-E.

The FXGP/WIN-E can activate two or more programs at the same time. For this reason, after starting FXGP-WIN-E, if the icon is double- clicked again, another FXGP/WIN-E will be started.

3.1.2 Starting FXGP/WIN-E for Windows 95 or After

This software cannot be run from the install (floppy) disk; this components must be installed onto hard-drive and subsequently run the software from the drive.

To start FX-PCS/WIN-E:

- Click the "Start Menu", choose "Program" ⇒ "MELSEC-F FX Applications", and click the name of the program you want to start.
- It is possible to also double-click a program icon to begin start-up.



3.2 Initial Screen Operation

	Read program from PLC
SwopC +XGP/WIN-E Image: SwopC +XGP/WIN-E File PLC New Ctrl+N Open Ctrl+O Page setup Printer setup LastFile Exit	Remote access to PLC via modem.

1) File menu

New

Use this function to create a new program. Select the PLC type, and then click the "OK" button.

A new file will appear on the display, and data entry is then possible.

Open

Use this function to read out program file from a floppy or hard disk.

- Select a file type from the Save As Type box, and then select a drive, directory, and file name.
- From the Save As Type box, select FXGP/WIN-E type program files (*.pmw) or DOS type program files (*.pmc).

2) PLC menu

Program read

This function reads the programs from the PLC connected to the PC. Before reading the programs, be sure to connect the PC to the PLC using the appropriate cabling and interface unit.

Ports

Use this function to select an RS-232 port from the PC in order to connect the PLC or ROM to the PC.

3) Remote

See.Sec.9.3 "Functions on Remote Menu"

1

3.3 Main screens

This section describes the main screens of FXGP/WIN-E.

Control menu box *1 Tool menu Title of window Tool bar Minimize button *1	2
SWOPC-FXGP/WIN-E'- SFC.PMW Maximize button *1	3
SFC SFC events SFC events of the second seco	
List program edit NO2	4
2 ANI Y001 3 OUT ■ Ladder Ladder Ladder	
su 5 LD 0 10 <td< td=""><td>5</td></td<>	5
Image: state of the state o	6
	7
Ladder (Program step(108/ 2000) Write FX (FX2/FX2C) Status bar (Displays	8
HELP:	L
Function key guidance	9

The Minimize button shrinks the size of the ladder program edit window, list program edit window, etc.

*1: When either Windows 95, Windows 98, Windows Millennium Edition, WindowsNT 4.0 Workstation, Windows 2000 or Windows XP are used, the FX icon will be displayed in place of the control menu box. In addition, the Minimize, Maximum, and Close buttons are different from those shown above.

3.4 Terminating method (Exit)

There are two program termination methods for FXGP/WIN-E. For either method, if there is a non-stored file, the following confirmation dialog box will appear on the display:

- 1) Sequentially select File and then Exit.
- 2) Click in the control menu box of the application window, and then select Close. When either Windows 95, Windows 98, Windows Millennium Edition, Windows NT 4.0 Workstation, Windows 2000 or Windows XP are used, click on the FX icon at the upper left corner of the box, or the Close button (×) at the upper right corner of the box.



The FXGP/WIN-E can open two or more the application windows.

Note that the Exit command is valid only for the application window currently active. If two or more programs are activated, sequentially end the programs by simultaneously activating the application windows.



4. Work Windows and their Functions

FXGP/WIN-E has six work windows that ensures efficient programming, thus, various form of operations can be performed using such windows.

To create a program, open the appropriate window, then create the program. Monitoring is also possible. Various function that can be executed in each window will be shown in the menu bar. The menu bar, therefore, is optimized for each work window.

4.1 Construction and Types of Windows

4.1.1 Construction of windows and relation between windows

The windows of FXGP/WIN-E can be divided into six types as shown below. Each window group can display two or more windows at the same time, and the active window can be switched in each group to proceed to the next Window. Minimize the unused windows to keep it on screen until you use it. The minimized window looks like an icon.



1

4.1.2 Handling of opened FXGP/WIN-E softwares at a time

In the FXGP/WIN-E, up two or more program files can open at a time. When either of the following operations is executed while the FXGP/WIN-E has been already started up and a program file is open, the second FXGP/WIN-E program can be started up.

- Started up a FXGP/WIN-E.
- Select [File (F)] \rightarrow [New (N)].
- Select [File (F)] \rightarrow [Open (O)].

Each of opened FXGP/WIN-E software at a time acts as an independent program file when changing over the active window. Or such programs can be displayed at a time when the window size is adjusted as shown in the figure below.

Among different application windows, programs and comments can be shared by using the cut & paste operation.

SWOPC-FXGP/WIN-E - [TANIGUTI.PMV	V:Ladder] <u>- ×</u>	
Eile Edit Tools Search View PLC	<u>R</u> emote		
Monitor/Test Option Window Help			SW0PC-FXGP/WIN-E - [FNC67_2N.PMW:Ladder] 💶 🗵
	MQ NG W		File Edit Tools Search View PLC Remote
🚟 號 🐝 H23 📰 🖭			Monitor/Test Option Window Help
а <mark>котог</mark> а като (лоо	07200 К/		
	(PL3	MID 1	
	[MOVP 0380	-	
- '' 			
		1	[SE7 M/0]
····] (2010 010 ···]	D'000 K2	∞]	
[xco	010 00	∞]	[][µus αα σz σ∞] [[
[* 072 0000 <u>)</u> [* 072	~ }	-• ->	[248 02 00 010]
[** 02 0'00]]		[FL" 010 [FL"
[+ cz cras ∦-≁ craz	∞) <u> </u>	_' →	[FLT D4 D14]
	יא פיס	∞ j	
(LCP 08	D2 D4	м]	
M2	[SE.1	~mm]	
[NB	D4 D10		
ме 17		-	
			· · · · · · · · · · · · · · · · · · ·
1	A		[rc]
Program step(154/8000) Write	· ·	FX	
	<i>a</i> O- <i>i</i>	£}	
			Program step(116/2000) Write FX

Application window 2

Application window 1

4.1.3 Split window

The ladder program edit, list program edit, comment and register windows can be split. Split the window into two; upper and lower windows.These windows can display the respective sections.

Example of a ladder program edit window (in monitoring mode)





After you edit a ladder diagram in a split window, store the changed data, and then move to another window.

🙏 MITSUBISHI

4.1.4 Selecting a window from tool menu

Select every work window of the FXGP/WIN-E from the menu bar.

Because the menu bar is always optimized according to the contents of work, some menu items cannot be selected on the active work window.

View menu



PLC menu



Monitor/Test menu



4.2 Functions of the Programming Window Group: View

The programming windows displays or edits programs. When a ladder or SFC program edit window is displayed, monitor the operation of the PLC.

Ladder View

-							P/WIN-E - SFO					¥ A
File		Tools	Search			Remote			n <u>W</u> ir	ndow	Help	
		😂 👗	Ba (B)	a 🖻		970 BR	B 🔒 🖬 🗟	8				
					· · ·							
•	M8000	_					Ladder					
0								[IST	X020	S20	\$33	1 🗒
8	S0 - STL -	X013	ү000 И								—(Y002	2
12		×010	¥802 И								(Y000	
15		×012	×007								(Y001	In all In all
		Y001										00
19		×006	×001 1/	×003	¥003						(Y004) <mark>- 1</mark> ≠ ka
24		X011	×005 И	×003	¥004						—(Y003	, <u> </u>
29	S1 STU-	×025								-[SET	S10	1
33	S10 STL -										——(Y002	2
										-[RST	Y000	1
+												
Lado	er			ÍF	roaram	step(14	4/ 2000)	Write	í	FX2	N	
HEL		سے آ ہ	- [.CHV			EO-1			_			
1922	1500	1 2 101			L MALE		164 1	10				

Instruction View

-					GP/WIN-E - SFC					-
File		<u>T</u> ools <u>S</u> earch		PLC Remo			Window	<u>H</u> elp		
		6 1 4 8	20	2 12 m m		8				
•					Instruction				Ŧ	
0		M8000								1
1	IST	×020	S20	S33						Н
8	STL	SO								
9	LD	×013								
10	ANI	Y000								
11	OUT	Y002								
12	LD	X010								
13	ANI	Y002								
14	OUT	Y000								
15	LD	×012								
16	OR	Y001								
17	ANI	×007								
18	OUT	Y001								
19	LD	×006								
20	ANI	X001								
21	ANI	×003								
22	ANI	Y003								
23	OUT	Y004								
24	LD	X011								-
+	1									£,
Instr	uction		(Us	sed steps (1	44/2000)	Insert	[FX2	2N		-
HE	PÍ,		LD	AND OR	ANB OUT					



0							P/WIN-E - S						-
File	<u>E</u> dit	Tools	Search		PLC	Remote				₩indow	<u>H</u> elp		
	20	8 👗	48	ð	a		0 B 🛤	1					
a							SFC					-	
6	8												1
	E)												
	ţ												
	917												
	013												
F.s	, T												Н
Ľ													П
·	- 1												
8	21												
	F	_											
-	t	_t_											
8	22	88											
Ι.	1	1											
6		636											
		μ											
·	+ 1	+											
8	24	827	1		1						1		
+												3	4
SFC	- 1				Dur. (H			Writ	e	(FX	2N		
HEL	P [,	a	°CHA	, 2	• ۲	, ⊽	jladi <mark>, -</mark>	10					_

Main functions:

- Displays ladder circuits.
- Creates or edits ladder circuits.
- Monitors the operation of the PLC.
- Displays or stores comments.
- Writes data during run mode.
- Prints ladder diagrams.

Main functions:

- Displays the instruction list.
- Creates or ediats the instruction list.
- Displays or stores comments.
- Prints the instruction list.

Main functions:

- Displays ladder circuits using the SFC format.
- Creates or edits SFC diagrams.
- Monitors operation of the PLC using the SFC diagram.
- Displays or stores comments.
- Prints the SFC diagrams Execute View (V) and then Inner Ladder View (A) in the SFC program edit window to display the inner state or transition ladder circuit.



4.3 Functions of the Comment Window Group: View \rightarrow Comment View

The comment windows displays, stores, or edits the device names and comments.

Device comment / Device name

ile <u>E</u> c					Monitor/Test	<u>Option</u>	₩indow	Help	
	5 . 46			B # 9	8				
18 小田 一部	50 II EQ								
-				COMMENT	.PMW:Commer	nt			-
Device		De	evice c	omment				Name	
Y000	LAMP No.1						LAMP1		
Y001	LAMP No.2						LAMP2		
Y002	LAMP No.3						LAMP3		
Y003	Ry1						NG.LAMF	,	
Y004	RY2						OK.LAMP	•	
Y005	Ry3						TEST		
Y006	Ry(POWER)						RY1		
Y007	ERROR No.1						RY2		
Y010	ERROR No.2						RY3		
Y011	ERROR.No.3						BUZZER		
Y012	MC1						LED1		
Y013	MC2						LED2		
Y014	MC3						LED.Y		
Y015	MC4						LED.R		
Y016							LED.B		
Y017							LEDG		
Y020									
Y021									
Y022									
ommen	ıt		Devic	e comment	(0/ 0)		FX	2N/FX2NC	
ELP a		- T - T		1					

Main functions:

Device name

- Stores and edits device names.
- Displays device names.
- Prints device names.

A device name can be stored for each device number. Device names are can be used instead of the device numbers when creating a program or displaying a program.

Device comment

- Stores or edits device comments.
- Displays device comments.
- Prints device comments.

Device comments are added to the device numbers. Devise comments are attached to all devices when the appropriate program is displayed.
Block comment

Coil comment

_					IN-E - COMME				•		
e <u>E</u> d					Monitor/Tes	t <u>O</u> ption	₩indow	Help			
38	8 x 10 2 2	DDD	900	<u> 9 m 8</u>	1 8						
Comment											
Step # Block comment											
13]	External display										
14											
15											
16											
17											
18											
19											
20											
21											
22											
23]	Model selection										
24											
25											
26											
27]	Latch clear										
28											
29											
30]	Error reset										
31											
mmen			1			7	EV	2N/FX2NC	1		
un I.a		1	2 4	-r		1		EN/TZZNC	1		

Main functions:

- Stores and edits block comments.
- Displays block comment.
- A block comment can be stored for each circuit block.

SWIPC-FXGPWINE-COMMENT.PMW • Elle Edit Gols Search Mew PLC Bemote MonitorTest Option Window Help • Elle Edit Gols Search Mew PLC Bemote MonitorTest Option Window Help • Image: State of the state of th

Main functions:

- Stores or edits coil comments.
- Displays coil comments.

A coil comments can be stored for each OUT, SET, RST and applied instruction.

4.4 Functions of the Register Window: View \rightarrow Register View

The register window displays data or file registers read from the PLC. Also, the register window edits data or file registers. On the completion of data editing, FXGP/WIN-E writes edited data to the PLC.

Examples of displayed windows

16-bit table type

	8 1	<u>42</u>	900	ng ng ng			8				
-					Registe						× A
	+0	+1	+2	+3	+4	+5	+6	+7	+8	+9	
D0 D10	65 75	66 76	67 77	68 78	69 79	70	71	72	73	74	
D10 D20	/5	/6	87	78	/9	80 90	81 23	82 898	83	84	
D30	св 0	08	87	18	18	210		1126	0	0	
D30	0		······	18	18	210	425	1120	0	0	
D40	0		0	0				0	0	0	
D60	0	0	0	0	0	0	0	0	0	0	
D70	0	0	0	0	0	0	0	0	0	0	
D80	0	0	0	0	0	0		0	0		
D90	0		······					0	°		
D100			0	0	0			0	0	0	
D110				0			0	0		0	
D120		0			0	0		0			
D130	0		·····	·····		0	ö	0	0	ö	
D140	Ň				0		Ň		n n	0	
D150	ů.		0	0	0	0		0	0	0	
D160	0	0	0		0	0	0	0	0	0	
D170	0	0	0		0	0	0	0	0	0	
D180	0	0			0	0	0	0	0	0	
D190	0	0	0	0	0	0	0	0	0	0	
D200	0	0	0	0	0	0	0	0	0	0	•
+					-						<u>.</u>
			,			,		FX21			-

Main functions:

- Displays or edits data or file registers read from the PLC.
- Creates or edits data or file registers, and writes the created or edited data to the PLC.
- Prints register data informat.

32-bit line type

_			SWOPC	-FXGP/WIN-E - [Register]	-
- File E	dit Tools Search	n View PL			\$
	5 1 1 1 2 2		2 B ##	9 T	
	DEC	HEX	ASCII	BIN FLOAT	Пı
DO	4325441	420041	A B	00000000 01000010 00000000 01000001 1.51563275E0	10
D2	4456515	440043	CD	00000000 01000100 00000000 01000011 1.53125799E0	IΓ
D4	4587589	460045	EF	00000000 01000110 00000000 01000101 1.54688823E0	11
D6	4718663	480047	6 H	00000000 01001000 00000000 01000111 1.56250846E0	11
D8	4849737	480049	IJ	00000000 01001010 00000000 01001001 1.5781337E0	11
D10	4980811	4C004B	KL	00000000 01001100 00000000 01001011 1.59375894E0	11
D12	5111885	4E004D	H N	00000000 01001110 00000000 01001101 1.60938418E0	11
D14	5242959	50004F	0 P	00000000 01010000 00000000 01001111 1.62500942E0	11
D16	5374033	520051	QR	00000000 01010010 00000000 01010001 1.64063466E0	11
D18	5505107	540053	ST	00000000 01010100 00000000 01010011 1.65625989E0	11
D20	5636181	560055	UV	00000000 01010110 00000000 01010101 1.67188513E0	11
D22	5767255	580057	HX	00000000 01011000 00000000 01010111 1.68751037E0	11
D24	5898329	5A0059	ΥZ	00000000 01011010 00000000 01011001 1.70313561E0	11
D26	58851351	3820017		00000011 10000010 00000000 00010111 7.64073391E-37	11
D28	0	0		00000000 00000000 00000000 00000000 0E0	11
D30	0	0		00000000 0000000 0000000 00000000 0E0	11
D32 D34	1179648	120000		00000000 00010010 00000000 00000000 1.140625E0	11
D34	13762578	D20012	×	00000000 11010010 00000000 00010010 1.92854794E-38	11
D36	73793961	4660189	o f	00000100 01100110 00000001 10101001 2.70371324E-36	11
D38	0	0		00000000 00000000 00000000 00000000 0E0	11
D40	0	0		00000000 0000000 0000000 00000000 0E0	11
D42	0	Ó		00000000 00000000 00000000 00000000 0E0	11
D44	Ô	Ó		00000000 00000000 00000000 00000000 0E0	11
D46	Ó	Ó		00000000 0000000 0000000 00000000 0E0	11
D48	Ó	Ó		00000000 0000000 0000000 00000000 0E0	11
D50	Ô	Ó		00000000 0000000 0000000 00000000 0E0	IL
052	0	Ó		00000000 0000000 0000000 00000000 0E0	IE
+					۰ſ
Register				FX2N	-
HELP	1. I. I.				
					_

There are two or more optional register display methods. If a different display method is specified, the displayed screen may differ from the example shown in the left figure.

4.5 Functions of the Device List Window Group: View \rightarrow

The device list windows search for devices (X, Y, M, S, T, C, D, P, I, N, V, Z, etc.) incorporated in to the PLC and display the devices that are used in programs.

Contact/Coil list

ile Edit Iools Search View PLC Remote Moniter/Test Option Window Help IPH & X22 & Commune DA & Y	
S Used List	-
Device Y000 Last device Page down Page up Next device	
4 -1/1-	
22 - []-	
LAMP1 28-11-	
Device comment	
LAMP N0.1	
adder Program step(44/ 2000) Write FX2N	

Device used list

-					- NEWS2.PM				¥	-
Eile Edit			LC Remo		iter/Test Op	ntion <u>M</u>	indow <u>H</u> e	elp		_
			a los jus jos							a
•				Used Li	st				▼ ▲	ł
Device	×0	Last device	Page do	IWD	Page up	Next d	evice			
×000	-11-	X010	-11-		X020	-11-				L
X001	-11-	X011	-11-		X021					L
X002	-11-	X012	-11-		X822					
X003	-11-	X013	-11-		X023					L
X004	-11-	×014			×024					L
X005	-11-	X015			×025	-11-				L
X006	-11-	X016			X026					L
X007	-11-	X017			×027					L
										L
										L
										L
										L
										L
										L
Ľ.,									······,	9
Instruction		Us	ed steps (1	56/2000)	Insert		FX2N			-
HELP ,			,		s 10					



- File Ed	ia Taula	Search <u>V</u> iew Pl		VIN-E - NEW.PMV Moniter/Test Op		Help	¥ 4
			ic <u>re</u> note			цевр	
•			Use	ed List			-
Device	TO	Last device	Page down	Page up	Next device		
то	K190	Т8		T16			
T1	K200	Т9		T17			I
T2	K50	T10	K80	T18			
тз	K340	T11	K120	T19			
T4	K500	T12	K380	T20	K60		
T5	K10	T13		T21	K600		
T6		T14		T22			
17		T15		T23			
adder		Proc	ram step(199/2	(000) (Write	(FX2	N	1
HELP		L L L			1		
		<u> </u>					

Main functions:

- Displays step numbers to clarify where the contacts and the output coils of the specified devices are used in the program.
- Displays device names and comments.
- Searches the program for specified contact or output symbols.
- Prints the list.

Main functions:

- Displays whether the specified contacts and output coils are used in the program.
 (Specify a device, and conditions of 24 contacts and coils started from the specified device will be displayed.)
- Prints the list.

Main functions:

• Displays the timer or counter setting values written in the program.

(Specify a device, and the setting values of 24 timers and counters started from the specified device will be displayed.)

• Prints the list.



4.6 Functions of the Sampling Trace Window Group: PLC \rightarrow Sampling Trace

The sampling trace windows to display the device on/ off operation and data change detected by the sampling trace function. A time chart will be used as the display format.

Example of displayed window



Main functions:

• Displays the operation history of the specified device.

The on/ off operation of the bit devices and the numeric values of the word devices will be displayed.

- Displays the detailed operation of the bar position.
- Prints the sampling trace result.

4.7 Functions of Device Monitor Window: Monitor/Test \rightarrow

Use the device monitor window to monitor the operation of the PLC devices on the screen.

Entry device monitor



Main functions:

- Monitors operation of the input relay (X), output relay (Y), auxiliary relay (M), state (S), timer (T), counter (C), data register (D), and index registers (V and Z).
- Up to 48 device numbers can be stored.
- The data register display type can be selected from the following types: 16/32 bit, binary, decimal, hexadecimal, ASCII, and floating point display types.

MEMO

5. Guidance of Main Operations

This section classifies, among explanation of operation methods and menu commands of the FXGP/WIN-E, functions and operations often used by keywords, and summarizes the execution method for each item. The contents described here may require a different operation or prerequisite depending on the actual operation status. Refer to this section as basic operations for the FXGP/WIN-E software.

5.1 Startup/Edit/window display

[Starting up the FXGP/WIN-E]

Choose the "FXGP/WIN-E" icon in the "MELSEC-F FX application" in the "Start Menu \rightarrow program". (Refer to Section 3.1)

[Starting up two or more FXGP/WIN-E programs]

While one or two FXGP/WIN-E programs are running, click again the "FXGP/WIN-E" icon or execute [File (F)] \rightarrow [New (N)] \rightarrow [Open (O)]. (Refer to Sections 4.1.2 and 6.1.1.)

[Terminating the FXGP/WIN-E]

Choose [File (F)] \rightarrow [Exit (X)]. (Refer to Sections 3.4 and 6.1.12.)

[Changing the display size of a ladder circuit or list on the edit window]

Choose [View (V)] \rightarrow [Zoom (Z)]. Or [Option (O)] \rightarrow [Font (F)]. (Refer to Sections 6.5.14 and 6.8.12.)

[Displaying the edit window in the split status]

Adjust the split handle provided at the bottom of the edit window in the vertical direction to split the screen. (Refer to Section 4.1.3.)

[Editing two or more programs at a time]

Start up two or more FXGP/WIN-E applications, then edit them while changing over the application window. Or edit them while collating them displayed side by side on the screen. (Refer to Section 4.1.2.)

[Changing over two or more windows efficiently]

In the Windows, use the key combinations "[GRPH] + [TAB]", "[GRPH] + [ESC]" and "[CTRL] + [TAB]" to change over the windows and change over the active window efficiently. (Refer to the Windows manual.)

[Changing the edit window type]

Choose the edit window type in [View (V)] menu. (Refer to Section 6.5.)

1

D

F



5.2 Program

[Creating a new program]

Choose [File (F)] \rightarrow [New (N)]. (Refer to Section 6.1.1.)

[All-clearing a program]

Choose [Tools (T)] \rightarrow [All clear (A)]. (Refer to Section 6.3.6.)

[Transferring a program to the PLC]

Choose [PLC (P)] \rightarrow [Transfer (T)]. (Refer to Section 6.6.1.)

[Reading a program from a file] Choose [File (F)] \rightarrow [Open (O)]. (Refer to Section 6.1.2.)

[Saving a program to a file]

Choose [File (F)] \rightarrow [Save (S)] (or [Save as (A)]). (Refer to Sections 6.1.4 and 6.1.5.)

[Checking for errors in a program]

Choose [Option (O)] \rightarrow [Program check (C)]. (Refer to Section 6.8.1.)

[Printing a program]

Display the window of item to be printed, then choose [File (F)] \rightarrow [Print (P)]. When printing two or more items at a time, execute [Print All (T)]. (Refer to Sections 6.1.6 and 6.1.7.)

[Setting/changing the program memory capacity]

Choose [Option (O)] \rightarrow [Parameter Settings (P)]. (Refer to Section 6.8.2.)

[Overwriting NOP instructions to a program]

Choose [Edit (E)] \rightarrow [NOP overwrite (W)] from the list program edit window. (Refer to Section 6.2.9.)

[Inserting NOP instructions into a program]

Choose [Edit (E)] \rightarrow [NOP insert (S)] from the list program edit window. (Refer to Section 6.2.10.)

[Removing NOP instructions from a program]

Choose [Edit (E)] \rightarrow [NOP remove (E)] from the list program edit window. (Refer to Section 6.2.11.)

[Displaying the program on the specified step]

Choose [Search (S)] \rightarrow [Go to step (O)]. (Refer to Section 6.4.6.)

[Searching and displaying the specified device number]

Choose [Search (S)] \rightarrow [Device name search (D)] (or [Contact/Coil search (O)]). (Refer to Sections 6.4.3 and 6.4.5.)

[Searching and displaying the specified the instruction, Contact or Coil]

Choose [Search (S)] \rightarrow [Instruction search (I)] (or [Contact/Coil Search (O)]). (Refer to Sections 6.4.4 and 6.4.5.)

[Displaying program on the set position]

Select [Search (S)] \rightarrow [Tag Set (G)] to store an arbitrary program position as the tag position. Select [Tag jump (J)] to display the program portion in the stored tag position. (Refer to Sections 6.4.13 and 6.4.14.)

[Replacing a device No.]

Choose [Search (S)] \rightarrow [Replace Device (V)]. (Refer to Section 6.4.10.)

[Exchanging an a-/b-contact]

Choose [Search (S)] \rightarrow [Change AB Contact (A)]. (Refer to Section 6.4.11.)

[Exchanging a device Number]

Choose [Search (S)] \rightarrow [Exchange Device (E)]. (Refer to Section 6.4.12.)

2

3

4

5

6

7

8

9

Α

B

С

I)

Ε

5.3 Ladder circuit

[Displaying the ladder program edit window]

Choose [View (V)] \rightarrow [Ladder view (L)]. (Refer to Section 6.5.1.)

[Editing a ladder circuit]

Pressing a function key or choosing symbol from [Tools (T)] menu on the ladder program edit window. (Refer to Sections 6.3 and 7.2.)

[Editing a ladder circuit symbol by an instruction word program]

Put the cursor in a symbol input position on the ladder program edit window, then input a instruction word from the keyboard in the same way as inputs to a list. (Refer to Section 7.2.2.)

[Printing a ladder format]

Choose [File (F)] \rightarrow [Print (P)] from the ladder program edit window. When printing two or more items at a time, choose [Print All (T)]. (Refer to Sections 6.1.6 and 6.1.7.)

[Converting a ladder circuit and handling a ladder error]

Pressing the [f4] or choosing [Tools (T)] \rightarrow [Convert (C)] from the ladder program edit window. (Refer to Sections 6.3.7 and 7.2.3.)

When a ladder error occurs, modify the program to establish the ladder circuit. When destroying the contents of change and terminating the conversion mode, first, open the list program edit window, close the ladder program edit window once. (Refer to Sections 6.2.6, 6.2.7 and 7.2.3.)

[Specifying a range assignment by program circuit blocks]

Put the cursor in the range assignment start position, "[SHIFT] + [\uparrow]/[\downarrow]", "[SHIFT] + mouse dragging" or "[Edit (E)] \rightarrow [Block select (B)] \rightarrow [Up (U)]/[Down (D)]" after convert circuit a ladder. (Refer to Sections 6.2.8 and 7.2.4.)

[Specifying a range assignment by ladder symbol]

Put the cursor in the range assignment start position, then drag the mouse to the range assignment end position. (Refer to Section 7.2.4.)

[Copying and pasting a ladder circuit]

Put the cursor on a symbol to be copied or specify a range to be copied, then choose [Edit (E)] \rightarrow [Copy (C)] (or [Paste (P)]). (Refer to Sections 6.2.8 and 7.2.4.)

[Inserting a blank line for new ladder circuit]

Put the cursor in an insertion position, then choose [Edit (E)] \rightarrow [Line insert (I)]. (Refer to Section 6.2.7.)

[Deleting the ladder circuit block]

Put the cursor in a deletion position, select a ladder circuit block to be deleted by operating "[SHIFT] + $[\uparrow]/[\downarrow]$ ", "[SHIFT] + mouse dragging" or "[Edit (E)] \rightarrow [Select Block (B)]", then choose [Edit (E)] \rightarrow [Cut (T)] (, [Delete (D)] or [DEL]). (Refer to Sections 6.2.8 and 7.2.4.)

[Deleting a Instruction symbol]

Put the cursor on a symbol to be deleted, then choose [Edit (E)] \rightarrow [Cut (T)] (or [Delete (D)]). Or the [DEL] and [BS] keys are available also. (Refer to Sections 6.2.2, 6.2.5 and 7.2.4.)

[Switching over the overwrite/insert mode for making program]

Press the [INS] key on the ladder program edit window. The input characteristics are displayed at the lower right corner of the screen. (Refer to Section 7.2.4.)

[Deleting a vertical line symbol]

Put the cursor at the upper right corner of a vertical line, then pressing "[SHIFT] + [f8]", choosing "[Tools (T)] \rightarrow [Line (W)]" or "[| DEL] on the tool pallet". (Refer to Sections 6.3.4, 6.5.8 and 7.2.4.)

[Inputting the pointer "P" or "I"]

Put the cursor on the left side of a bus bar (that is, at the left edge of the screen), then press the [P] or [I] key. (Refer to Section 7.2.1.)



[Inputting horizontal lines/vertical lines continuously]

[GRPH] + [→]/[←]: Inputs horizontal lines. (Refer to Section 7.2.1.) $[GRPH] + [\uparrow]/[\downarrow]$: Inputs vertical lines. (Refer to Section 7.2.1.)

[Deleting horizontal lines/vertical lines continuously]

 $[CTRL] + [GRPH] + [\rightarrow]/[\leftarrow]:$ Deletes horizontal lines. (Refer to Section 7.2.4.) $[CTRL] + [GRPH] + [\uparrow]/[\downarrow]:$ Deletes vertical lines. (Refer to Section 7.2.4.)

[Destroying the contents of ladder circuit edition]

Choose [Edit (E)] \rightarrow [Edit cancel (E)], before converting ladder program. (Refer to Section 6.2.17.)

5.4 List

[Displaying the list program edit window]

Choose [View (V)] \rightarrow [Instruction list view (I)]. (Refer to Section 6.5.2.)

[Editing a list]

Input an instruction word directly on the list program edit window. Or using a function key or [Tools (T)] \rightarrow [Instruction (I)]. (Refer to Sections 6.3.5 and 7.3.2.)

[Printing a list]

Choose [File (F)] \rightarrow [Print (P)] on the list program edit window. When printing two or more items at a time, execute [Print All (T)]. (Refer to Sections 6.1.6 and 6.1.7.)

5.5 SFC

[Displaying the SFC program edit window]

Choose [View (V)] \rightarrow [SFC view (S)]. (Refer to Section 6.5.3.)

[Editing an SFC diagram]

Using function keys on the SFC program edit window. (Refer to Sections 6.3 and 8.2.)

[Converting an SFC program]

Press the function key [f4] or choose [Tools (T)] \rightarrow [Convert (C)] on the SFC program edit window. (Refer to Sections 6.3.7 and 8.2.3.)

[Printing an SFC diagram]

Choose [File (F)] \rightarrow [Print (P)] on the SFC program edit window. When printing two or more items at a time, choose [Print All (T)]. (Refer to Sections 6.1.6 and 6.1.7.)

5.6 Instruction input

[Inputting a program in an device name comment]

In programming, a device No. such as X0 can be input in an device name comment. During the input operation, input to the X0 an device name registered to the X0 while adding the "" symbol on either side. Example: LS1; "LS1" is an example of device name registered to the X0. (Refer to Sections 7.2.1 and 7.3.1.)

[Inputting an applied instruction by selecting it from a list]

On the ladder program edit window, choose [Tools (T)] \rightarrow [Function (F)] \rightarrow [Reference (R)].

In a list program edit window, choose [Tools (T)] \rightarrow [Instruction (I)].

(Refer to Sections 6.3.3 and 6.3.5.)



2

3

4

5

6

7

8

9

Α

В

С

I)

F

5.7 PLC

[Selecting the PLC type]

When creating a new file, when reading (up-loading) a program from the PLC, or when reading a file in the DOS version, the PLC type setting dialog box is displayed upon necessity. (Refer to Sections 6.1.1, 6.1.2 and 6.6.1.)

[Changing the PLC type]

Choose [Options (O)] \rightarrow [PLC type change (H)]. (Refer to Section 6.8.9.)

[Writing (down-loading) a program to the PLC]

Choose [PLC (P)] \rightarrow [Transfers (T)]. (Refer to Section 6.6.1.)

[Writing (down-loading) a register (data register/file register/RAM file register) to the PLC]
Choose [PLC (P)] \rightarrow [Register data transfers (R)]. (Refer to Section 6.6.2.)

[Clearing the program, word device or bit devices in the PLC]

Choose [PLC (P)] \rightarrow [PLC memory Clear (P)]. (Refer to Section 6.6.3.)

[Checking error status in the PLC]

Choose [PLC (P)] \rightarrow [PLC diagnosis (D)]. (Refer to Section 6.6.8.)

[Checking the PLC operation time]

Choose [PLC (P)] \rightarrow [PLC diagnosis (D)]. (Refer to Section 6.6.8.)

[Registering/changing the password for password protection in/to the PLC]

Choose [PLC (P)] \rightarrow [PLC keyboard current, delete (K)]. (Refer to Section 6.6.5.)

[Changing the setting value of a timer/counter in the PLC]

Choose [Monitor/Test (M)] \rightarrow [Change the setting value (C)]. (Refer to Section 6.7.7.) Select one target device on the ladder program monitor window, then click the right button of the mouse to display the shortcut menu. (Refer to Section 6.7.1.)

[Changing the current value of a timer/counter/register in the PLC]

Choose [Monitor/Test (M)] \rightarrow [Change the current values (C)]. (Refer to Section 6.7.8.) Select one target device on the ladder program monitor window, then click the right button of the mouse to (switching) display the shortcut menu. (Refer to Section 6.7.1.)

[Forcing the output to ON/OFF]

Choose [Monitor/Test (M)] \rightarrow [Force Y Output (F)]. (Refer to Section 6.7.5.)

[Forcing a device to set/reset]

Choose [Monitor/Test (M)] \rightarrow [Force ON/OFF (R)]. (Refer to Section 6.7.6.)

Select one target device on the ladder program monitor window, then click the right button of the mouse to display the shortcut menu. (Refer to Section 6.7.1.)

[Changing a program in RUN mode]

Change only one ladder circuit block on the ladder program edit window, then press "[SHIFT] + [f4]" or choose "[Tools (T)] \rightarrow [Runtime program changes (U)]" without converting the ladder circuit block. (Refer to Section 6.6.6.)

[RUN/STOP the FX1s/FX1n/FX2n/FX1nc/FX2nc PLC]

Choose [PLC (P)] \rightarrow [Remote RUN/STOP (O)]. (Refer to Section 6.6.7.)

5.8 Monitor

[Monitoring a ladder program]

Choose [PLC (P)] \rightarrow [Start monitor (M)] on the ladder program edit window. (Refer to Section 6.7.1.)

[Monitoring an SFC diagram]

Choose [PLC (P)] \rightarrow [Start monitor (M)] on the SFC program edit window. (Refer to Section 6.7.1.)

[Monitoring device status]

Choose [Monitor/Test (M)] \rightarrow [Entry device monitor (cursor) (E)]. (Refer to Section 6.7.4.) Select one target device on the ladder program monitor window, then click the right button of the mouse to display the shortcut menu.

[Monitoring the value of the data register, timer or counter]

Choose [Monitor/Test (M)] \rightarrow [Entry device monitor (T, C, D device) (T)]. (Refer to Section 6.7.4.) Select one target device on the ladder circuit monitor screen, then click the right button of the mouse to display the shortcut menu.

[Monitoring the SFC program automatically]

Choose [Monitor/Test (M)] \rightarrow [Start auto monitor (A)] on the SFC program edit window. (Refer to Section 6.7.3.)

[Stopping the monitor]

Choose [Monitor/Test (M)] \rightarrow [Stop Monitor]. (Refer to Section 6.7.2.)

[Saving and reading the registered contents of the Entry device monitor]

Choose [File (F)] \rightarrow [Export (E)] (or [Import (I)]) on the device monitor window. (Refer to Section 6.7.4.)

5.9 Comment

[Displaying a program with comments]

While the ladder/list/SFC program edit window is displayed, choose [View (V)] \rightarrow [Comment display (M)]. (Refer to Section 6.5.12.)

[Printing a list of the device names or device comments]

Choose [View (V)] \rightarrow [Comment view (C)] \rightarrow [Device comment/Device name (D)], and display the device name/device comment edit window.

Then choose [File (F)] \rightarrow [Print (P)] (or [Print All (T)] for printing out two or more items at a time). (Refer to Sections 6.1.6, 6.1.7 and 6.5.5.)

[Printing a program with a comment]

Choose [File (F)] \rightarrow [Print (P)] (or [Print All (T)] for printing out two or more items at a time), then specify comment type to be printed on the print setting dialog box. (Refer to Sections 6.1.6 and 6.1.7.)

[Registering an device name to a device]

While the ladder or list program edit window, etc. is displayed, choose [Edit (E)] \rightarrow [Device name (N)]. Or choose [View (V)] \rightarrow [Comment view (C)], then input device name at a time. (Refer to Sections 6.2.12 and 6.5.5.)

[Registering a device comment]

While the ladder program list program edit window, etc. is displayed, choose [Edit (E)] \rightarrow [Device comment (V)]. Or choose [View (V)] \rightarrow [Comment view (C)], then input device comments at a time. (Refer to Sections 6.2.13 and 6.5.5.)

[Registering a coil comment]

While the ladder program edit window is displayed, put the cursor on a coil for which a comment is to be registered, then choose [Edit (E)] \rightarrow [Coil Comment (O)]. (Refer to Section 6.2.14.)



[Registering a block comment]

While the ladder program edit window is displayed, put the cursor on a ladder circuit block for which a comment is to be registered, then choose [Edit (E)] \rightarrow [Block comment (M)]. (Refer to Section 6.2.15.)

[Inputting a device comment efficiently while making program]

Choose [Options (O)] \rightarrow [Preferences (I)], then add check mark to "Executes consecutively during instruction input." By this setting, the dialog box for device comment input is displayed after instruction was inputted. (Refer to Section 6.8.10.)

[Clearing comments]

Choose [View (V)] \rightarrow [Comment view (C)] to display the device name/device comment edit window. Then choose [Tools (T) \rightarrow [All clear (A)] and then select the cleared comments type. (Refer to Sections 6.3.6 and 6.5.5.)

[Using an existing comment file]

To export a comment file, choose [File (F)] \rightarrow [Export (E)] on the comment edit window. To import the existence comment file, choose [File (F)] \rightarrow [Import (I)] on the comment window. (Refer to Section 6.5.5.)

5.10 Printer

[Printing (active window)]

While an edit window to be printed is displayed, choose [File (F)] \rightarrow [Print (P)]. (Refer to Section 6.1.6.)

[Printing (two or more windows)]

While an arbitrary edit window is displayed, choose [File (F)] \rightarrow [Print All (T)]. (Refer to Section 6.1.7.)

[Previewing the print result]

While an edit window to be previewed is displayed, choose [File (F)] \rightarrow [Print preview (V)]. Or click "Preview (V)" button on the print setting dialog box. (Refer to Sections 6.1.9 and 6.1.6.)

[Selecting print items]

Choose [File (F)] \rightarrow [Print (P)] (or [Print All (T)] for printing out two or more items at a time), then specify an item (or items) to be printed. (Refer to Sections 6.1.6 and 6.1.7.)

[Limiting the print range]

Choose [File (F)] \rightarrow [Print (P)] (or [Print All (T)] for printing out two or more items at a time), then specify the range to be printed. (Refer to Sections 6.1.6 and 6.1.7.)

[Attaching the front cover or the parameter setting]

Choose [File (F)] \rightarrow [Print (P)] (or [Print All (T)] for printing out two or more items at a time), then add check mark to check box of the "Title (I)" or "Parameter settings list (E)." (Refer to Sections 6.1.6 and 6.1.7.)

[Changing the print size]

Choose [File (F)] \rightarrow [Print (P)] (or [Print All (T)] for printing out two or more items at a time), then choose size from "Scale" pull down menu. (Refer to Sections 6.1.6 and 6.1.7.)

[Setting the margin, header and footer of the print paper]

Choose [File (F)] \rightarrow [Paper setup (G)]. (Refer to Section 6.1.8.)

[Setting the paper size, direction and paper feed position]

Choose [File (F)] \rightarrow [Printer setup (G)]. (Refer to Section 6.1.10.)

[Printing according to the paper size]

Choose "Auto" from "Scale" pull down menu on the print setting dialog box. By this setting, printing is executed according to the lateral width of the specified paper. (Refer to Section 6.1.6.)

3

1

2

5	

6	

7	
1	

В

5.11 PC

[Connecting the PC to the PLC]

As the RS-232C/RS-422 conversion interface, use the FX-232AWC or FX-232AW. The connection cable depends on the PC and connected PLC model; Refer to the corresponding explanation item. (Refer to Section 1.2.3.)

[Connecting the PC to the ROM writer]

Connect the COM port (RS-232C) connector in the PC to the RS-232C port in the ROM writer with a cable FX-232CAB-1. (Refer to Sections 1.2.3 and 6.8.11.)

[Selecting an RS-232C port in the PC]

Choose [PLC (P)] \rightarrow [Ports (S)]. (Refer to Section 6.6.10.)

[Changing the password registered with a program in the PC]

Choose [Options (O)] \rightarrow [Entry Code Setting (Y)]. (Refer to Section 6.8.3.)

5.12 Data compatibility and conversion

[Reading data in the DOS version]

Choose [File (F)] \rightarrow [Open (O)], then select "*.PMC" from "File Type (T)" pull down menu. (Refer to Sections 6.1.2 and 3.1.3.)

[Saving data in a DOS format file]

Choose [File (F)] \rightarrow [Save As (A)], then select "*.PMC" from "File Type (T)" pull down menu. (Refer to Sections 6.1.2 and 3.1.1.)

[Handling comments in the program memory]

Comments can be transferred to device comments. In this case, comments to be written to the PLC need to set. (Refer to Section 1.6.)

5.13 Telephone line connection

[Downloading, uploading and monitoring with the PLC through the telephone line] Use the remote maintenance of the PLC. (Refer to Section 9.1.1.)

[Sending and receiving data through the telephone line]

Use the file transfer function. (Refer to Section 9.1.2.)

[Connecting to the telephone line]

For remote maintenance refer to Section 9.2.1. For file transfer, refer to Section 9.2.2.

[Modem setting]

Choose [Remote (R)] \rightarrow [Environment (E)] \rightarrow [Modem (M)]. (Refer to Section 9.3.4.)

[Connecting the telephone line]

Choose [Remote (R)] \rightarrow [Connect (C)]. (Refer to Section 9.3.1.)

[Setting the FX1s/FX1N/FX2N/FX2NC PLC for connection to the telephone line] Set the PLC for remote maintenance. (Refer to Section 9.4.)



Note

The remote maintenance cannot be supported on the following operating systems. Windows NT 4.0 Workstation Windows 2000 Windows XP



5.14 Others

[Open recent file]

Displays the history of the file opened in the past. Select the arbitrary file from the displayed history and clicking. The file that has been opened in the past can be opened again. (Refer to Paragraphs 6.1.11 and 3.1.1.)

[Tracing the device status in PLC]

The FX/FX_{2C}/FX_{2N}/FX_{2NC} PLC is equipped with the sampling trace function which traces the device status in the PLC. For the condition settings and result display of the sampling trace function, choose [PLC (P)] \rightarrow [Sampling trace (M)]. (Refer to Paragraph 6.6.9.)

[Downloading the data register value to PLC from PC]

Choose [PLC (P)] \rightarrow [Register data transfers (R)] to download the value of a data register and file register created in the PC to the PLC. Because a file register is stored in the program memory, it can be transferred also by [Write (W)]. (Refer to Paragraphs 6.5.6 and 6.6.2.)

[Setting the parameter]

Choose [Options (O)] \rightarrow [Parameter setting (P)] to set parameters including the area to hold devices in the program step/file register area in the PLC. (Refer to Paragraph 6.8.2.)

[Setting the file register area]

Choose [Options (O)] \rightarrow [Parameter Setting (P)]. (Refer to Paragraph 6.8.2.)

[Specifying a display font]

Choose [Options (O)] \rightarrow [Font (F)]. Specify "MS Sans Serif" usually because something may be wrong with the display if an alphabetic font is selected. (Refer to Paragraph 6.8.12.)

[Using help file]

Select the [Help (H)] menu or pressing the function key (f1) to display help file. (Refer to Paragraph 6.10.)

[Checking the version]

Choose [Help (H)] \rightarrow [About SW0PC-FXGP/WIN-E (A)]. (Refer to Paragraph 6.10.)

MEMO

6. Menu Bar Commands

This chapter describes the functions of the commands shown on each of the pull-down menus. The commands shown on the pull-down menu depend on the edit window currently active and the operation status. In addition, the function of the same command may be changed when the active window is switched.

Description of this section

This section describes all the commands of each pull-down menu. The command-valid windows and the operation status will be shown first for each command.

Showing valid commands:

L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
~		~	✓		~	~	~	✓	✓

Window or operation status

Check mark showing valid command____

OPN: Opening

- L/V : Ladder View
- L/ M : Ladder Monitor
- I/V : Instruction View
- S/ V : SFC View
- S/ M : SFC Monitor
- C/V : Comment View
- R/ V : Register View
- D/L : Contact/Coil list, Device used list, TC setting list
- D/ M : Device Monitor
- SMP : Sampling trace

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
1	1	~	1	1	1	1	1	1	1	~

6.1 File Menu Functions

6.1.1 New

Use this command to create new program files.

PLC type setting		×
C FX <u>0</u> / FX0S	C FX1 <u>S</u>	ОК
C FX0 <u>N</u>	C FX <u>1</u> N	Cancel
C FX (FX <u>2</u> /FX2C)	FX2N / FX2NC	Help

- Choose "File" and then "New". The PLC Type Setting dialog box will appear on the display.
- Select a PLC type, and then click the "OK" button.



 Commands, internal devices, and number of steps usable for programming depend on the PLC type specified in the PLC Type Setting dialog box.



2

3

4

5

6

7

8

9

Α

B

С

D

Ε

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
1	1	1	1	1	1	1	1	1	1	1

6.1.2 Open

File <u>N</u>ame:

sfc.pmw sfc_lan.pmw

List Files of <u>Type</u> WIN Files (*.pm

WIN Files (*.pm WIN Files (*.pm DOS Files (*.pm All Files (*.*)

*.pmw lad1.pmw new.pmw news.pmw news2.pmw orikaesi.pmw s-p38-41.pmw sfc.pmw

Use this command to read out a program file from a floppy or hard disk.

OK

<u>H</u>elp

Network

ŧ

Cancel

- Choose "File" and then "Open". The Open dialog box will appear on the display.
- From the File Type box, select the FXGP/WIN-E program (*.pmw) or DOS files (*.pmc).

- Check the items set in the window, and then click the "OK" button.
- Any file types shown in the File Type box can be read.The dimmed file types, cannot be read as they do not exist.
- The data in the DOS file will be converted into the FXGP/WIN-E data, and read.
- Any file types shown in the File Type box can be converted.
 - The dimmed file types, however, cannot be converted as they do not exist.
- Check the items set in the window. Then, after that, select the PLC type from the PLC Type box, and click the "OK" button.
- When a program is opened, if the program includes comments stored in the program memory, the message shown in the left figure will appear.

In the dialog box, click "Yes" to transfer comments to the FXGP/WIN-E device comment area. In addition, the comments transferred will be set as the device comments to be written to the PLC. (See Sec. 1.4 and Sec. 6.3.9.)

If "No" is clicked in the message window, comments will be loaded in FXGP/WIN-E as program data, therefore the device comment area will not be subject to change. However, after loading the comments as program data, converting them into the device comments is an option. In this case, convert the comments into device comments with reference to Sec. 6.8.8.

FXGP/WIN-E file open _ <u>F</u>ile name <u>0</u>K C:\EXGPWIN\NEW.PMW File title <u>C</u>ancel PLC type: FX (FX2/FX2C) <u>H</u>elp File type Program(PMW) Benister(DMW) Comment(COW) Print title(PTW) DOS file open DOS file open Eile name C:\FXGPWIN\NEW.PMC <u>0</u>K <u>C</u>ancel File title TEST-1 <u>H</u>elp ⊢File type Program(PMC) Comment(COK) Data register(DMD.DME.DMG) Sampling trace(STA) RAM file register(DMF) Print title(PTL) PLC type FX (FX2/FX2C) FXUN FX (FX2/FX2C)

File Open

<u>D</u>irectories c:\fxgpwin

🗁 c:\ 🎦 fxgpw 🛅 usr

Drive

+



To open a file including comments:

_		SW0PC-FXGP/WIN-E
	2	Program areas comment move to comment file. comment data will be overwritten.Are you OK?
		<u>Yes</u> <u>N</u> o



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
1	1	1	1	1	1	1	1	1	1	✓

X



When opening the file for DOS application format, specify the PLC type on the following dialog box.

OS file open			
ile name			ОК
A:******** .PMC			
ile title			Cancel
			Halp
File type			<u>H</u> elp
Program(PMC)	Comment(COK)		
Data register(DMD.DME.DMG)	Sampling trace(STA)	
RAM file register(DMF)	Print title(PTL)		
FX1S FX1N FX2N / FX2NC			
W0PC-FXGP/WIN-E	×	When the	•
PLC type error. Continue	program read?	between f application,	left dialog
Yes No		If clicking "Y the range o type (The	of the spe
		excluded.).	

When the specified the PLC type is disagreement between FXGP/WIN-E and FXGP/AT-EE(DOS) application, left dialog box is appeared. f clicking "YES" button, the program file is opened within he range of the specification of the specified the PLC

type (The device number and the instruction are excluded.). Therefore, the program of the area where the program capacity is exceeded is deleted, and the change in the instruction and the device number is needed.

Files read simultaneously

- Open a program file (*.PMW type file) by executing the Open and Close commands. All the related files will be read.
- Files to be read:



As described above, specify TEST. PMW and execute Open, four files shown above will be read simultaneously. For this reason, always specify the *.PMW type file even if an a alternate file other than the program file, such as the comment or register file is needed.

If the *.COW, *.DMW, or *.PTW file is deleted for some reason, only the existing files will be read. Note that if the *.PMW type file is deleted, no files will be read even if some files are stored.

To read the sampling trace file (*.STW), choose [PLC (P)] \rightarrow [Sampling trance (M)] \rightarrow [Read from result file (R)].

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1		1	1		1	1	1	1	~

6.1.3 Close and Open

2

Yes

Use this command to close the current program file and open another existing file.

- 1) If the program is not changed:
 - Execute File and then Close and Open. The current file will be closed immediately.
 - After closing the file, the File Open dialog box will appear on the display (see Sec.6.1.2).
- 2) If the program is changed:

Execute File and then Close and Open. One of the following dialog boxes will appear on the display. The dialog box will depend on whether the file name is specified.

_	SWOPC-FXGP/WIN-E							
0	Save Changes to C:¥FXGPWIN¥NEW.PMW?							
	Yes <u>No</u> Cancel							

SWOPC-FXGP/WIN-E

Save Changes to untitl01?

No

- If the file name is already specified, the message in the dialog box will prompt whether the data should be overwritten on the specified file.
- If the data is new and has no file name, the message shown in the dialog box will prompt whether the data should be stored in the untitled file.

_	File Save As	
File Name: Thitubi.pmw comment.pmw key_word.pmw komen.pmw ladi.pmw news.pmw news.pmw	Directories: c:\fxgpwin c:\ c:\ frgpwin usr	Cancel
Save File as <u>T</u> ype:	Dri <u>v</u> es:	

Cancel

To store the program data, specify the directory and the file name, then click the OK button.

If no file name is specified, the data will be stored to a temporary file, Untitl**.pmw.

For a detailed description of Save As dialog box, refer to Sec.6.1.5.

3	
4	
5	
6	
7	
8	
9	
A	
В	
С	
D	
E	

1

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1		1	~		~	1	1	1	✓

6.1.4 Save

Use this command to overwrite the currently-displayed program data in the specified file.

1) When an existing file is open

Execute File and then Save. The program data will be overwritten immediately continuous. At the completion of overwrite, the program will not be closed, therefore, edit of the program is possible.

2) When a new file is open

Execute File and then Save As. The Save As dialog box will appear on the display.

	File Save As		Specify the directed		
File <u>N</u> ame: untitl01.pm w	Directories:	OK Cancel	name, then click th If no file name is s		
c0mment.pmw comment.pmw key_word.pmw komen.pmw lad1.pmw new.pmw news.pmw news2.pmw	↑ C:\ P fxgpwin □ usr	transformed by the second	will be stored to a Untitled**.pmw.		
Save File as <u>T</u> ype: WIN Files (*.pmw)	Dri⊻es: ➡ c: ms-dos_6	<u>+</u>			
WIN Files (*.pmw) DOS Files (*.pmc) – All Files (*.*)					
		Stores data in a	FXGP/ WIN-E type file.		
		- Stores the data i	n a DOS type file.		

ory and the file OK button. ecified, the data temporary file,



Files to be stored at the same time

- Store the program data in a *.PMW type file by executing the Close and Open, Save, and Save As commands. All the related files will be created or updated.
- Files to be stored



As described above, specify TEST.PMW and then execute Save, four files shown above will be created or updated at the same time. For this reason, always specify the *.PMW type even if you want to store a file other than the program file, such as the comment or register file.

To save the sampling trace file (*.STW), choose [PLC (P)] \rightarrow [Sampling trace (M)] \rightarrow [Write to result file (W)].



2

3

4

5

6

7

8

9

Α

В

С

Ε

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1		1	1		1	1	1	1	1

6.1.5 Save As

Use this command to store current program data displayed in the specified file.





How to assign a file name

 Input a file name in up to 16 characters. The FXGP/WIN-E does not support long file names available in the Windows.

Due to the restriction in the operating system, the following characters are not allowed in file names.

. , : ; " < > | / \ * ?

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		~	~		~	√	~	~	~

6.1.6 Print

Clicking the Print command, enables the data in the currently-active window to be printed. Therefore, if you click the Print command when the ladder program edit window is active, the ladder diagram will be printed. If you print two or more items, such as the ladder diagram, instruction list, device use list, etc., click Print All.

Relation between active window and items to be printed



2

Ε

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		1	~		1	1	1	1	~

Printing ladder diagram

This section describes how to print programs using the format of ladder diagrams. When the ladder program edit window is active, execute File and then Print.

Set the comment type to be printed.



Click Copy Settings on the previous screen to display this screen. This screen displays the comment display data set in the program edit window.



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		~	~		~	~	~	~	✓

Printing instruction list

This section describes how to print programs using list program format. When the list program edit window is active, execute File and then Print.

Block comment





2

С

Ε

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		✓	~		~	~	~	~	✓

Printing SFC diagram

This section describes how to print programs using the format of SFC diagrams. When the SFC diagram edit window is active, execute File and then Print.



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		~	~		<	<	~	~	✓



Click Copy Settings on the previous screen to display this screen. This screen displays the comment display data set in the SFC program edit window.



Click Copy Settings on the previous screen to display this screen. This screen displays the comment display data set in the inner ladder program edit window.

2

Α

В

С

Ε

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		1	1		~	~	1	~	~

Printing Device comment / Device names

This section describes how to print the list of Device comments/Device names. When the device name window is active, click File and then Print.



the specified printing

sheet.

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		~	1		1	1	1	1	1

Printing contact/coil list

This section describes how to print the contact/coil list. The list shows the program steps that use contacts or coils of devices.

When the contact/coil list window is active, execute File and then Print.

	Contact/Coil list	
Specify the printing → device range.	Range OK OAll range Cancel Setting range - Y - P - M - 1 -	 Starts printing.
	S - N - Preview T - - Z - C - V -	Displays the image of the printed document on the preview screen.
prir	Device name Device comment Only print used devices Title Scale Parameter settings list AUTO ± Click this button to print the parameter setting page. Before printing, set the parameter setting page by clicking Option and Parameter Setting. Ck this button to print the title setting page. Before https://www.automatical.com/option/automat	 Click the desired button in this box to print the comments added to the respective device numbers. Specify the printing size. Specify "Auto" or a percentage between "50% and 150%" (in increments of 5%). Specifying Auto, adjusts the number of characters in one line automatically depending on the width of the specified printing sheet.

Printing device use list

This section describes how to print the device used list. The list shows whether the contacts and coils of devices are used in the program.

When the device use list window is active, click File and then Print.

Device used list	
Range Image Setting range X - Y - M - S - T - N -	OK Cancel Help Preview
Comment Device name Device comment Dipole comment Dipole comment Title Scale Parameter settings list AUTO	

To set data in the device used list printing dialog box, refer to "Printing contact/coil list" as described above.



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	✓		1	1		1	1	1	1	1

Printing TC setting list

This section describes how to print the TC setting list. The list shows the timer/counter setting values used in the program.

When the TC setting list window is active, click File and then Print.

TC setting list	
Range (All range) Setting range T	OK Cancel
Comment Device name Device comment Donly print used devices	Help Pre <u>v</u> iew
☐ Title Scale ☐ Param <u>e</u> ter settings list AUTO <u>↓</u>	

To set data in the TC setting list printing dialog box, refer to "Printing contact/coil list" as described above.

Printing Register list

Print the list of current data register values.

To print this list, activate the data register window. While displaying the data register window, click File and then Print.

-	Data register	
Print form I able Line Print mode Binary	Data size	OK Cancel Help
Deci <u>m</u> al He <u>x</u> adecimal <u>A</u> scii Eloating point	O <u>d</u> d	Pre <u>v</u> ie w
Range ● All <u>r</u> ange ○ Setting range D D	to	
☐ T <u>i</u> tle ☐ Param <u>e</u> ter settings	<u>S</u> cale list AUTO <u>+</u>	

To set the printing data in this window, refer to the data setting method as described in Sec. 6.5.6 "Register View".

D

Ε

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		~	~		~	~	~	~	✓

6.1.7 Print All

The Print command described in the previous section is valid only for the active window.

The Print All command described in this section, however, can print all the specified items in the specified order while ignoring the active window.

Execute File and then Print All. Execute these commands in any edit window.



- To store a printing item to the print list, position the cursor on the desired item in the Item List box.
 Then, click → to move as item to the Print List box.
- To delete an item from as print list, position the cursor on the item in the Print List box.
 Then, click ← to delete as item from the print list and move it to the Item List box.

2

С

D

Ε

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
1	~		1	1		1	~	1	1	✓

6.1.8 Page Setup

Before printing use this command to set page settings. For example, set the margin, header, footer, and page number printing position.

Use the Printer Setup command to set the paper size and printing direction.



Set the data for the header and footer without their own appropriate windows, then click Option and Printer Title. The header and footer will be printed at the upper and lower sections of the page as specified in the respective windows.

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		1	1		1	1	1	1	1

6.1.9 Print Preview

Use this command to display the image of the printed document on the preview screen. The image shown on the preview screen depends on the data set in the printing condition setting dialog box, Page Setup window, and Printer Setup window.

To view the preview screen, open the related window, then click File and Print Preview.

Click this button to open the printing condition setting dialog box of the corresponding work window.



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		~	~		~	~	~	~	✓



Note

- Clicking print and print all will allow for the preview screened to be viewed. Click Print and then Print All.
 - If the preview screen reveals that the paper size is too small to contain all the data, adjust the page data as follows:

Click Page Setup to open the Page Setup window, then adjust the margins.

Click Printer Setup to open the Printer Setup window, then adjust the paper size and printing direction.

Open the printing condition setting dialog box of the corresponding item, and then adjust the zoom rate.

1

2

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
1	1		1	1		1	1	1	1	1

6.1.10 Printer Setup

Use this command to set the printer type, paper size, and printing direction.



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
✓	~	~	~	~	~	~	✓	~	~	~

6.1.11 Displaying file use history


1

2

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
✓	~	~	~	~	~	~	~	~	~	✓

6.1.12 Exit

Use this command to exit FXGP/WIN-E.

Select File and Exit to leave FXGP/WIN-E.

	<u> </u>			GP/WIN-E -						•				
<u>File E</u> dit <u>T</u> ools <u>N</u> ew Open			<u>R</u> emote	Moniter/Test	t <u>O</u> ption	n <u>W</u> indov	w <u>H</u> elp							
Closed and ope	en			Ladder	Сто	ко	КЭ	H0000	K1	 ▼ ▲ ↓ 			7	
<u>S</u> ave Save <u>a</u> s	Ctrl+S	СОТО	ко	К4	210	K10000		К1	112					
Print	Ctrl+P				[T0	ко	К6	КО	К1	E	ľ	f a file is	not store	ed.
Prin <u>t</u> All Page setup					[TO	КО	K15	K100	K1	Э			1	
Print pre <u>v</u> iew Printer setup					(FROM	ко	K28 K29	K9M20 D20	к1 к1					
1 NEW.PMW		- 			2				—с мо	2				
2 SFC2.PMW 3 LAD1.PMW								<u>+++++</u>	—С М1	>				
4 COMMENT.PM	1₩						(111 141 111 141	—С M2 —С M3	2				
73 M8000							[<u>아마</u> 	C M7	2				
	M120						l	<mark>≁ 00.</mark>	C M12	2				
M50 80 80		[DTOP	ко	K26		КО		K1	C M50	2				
		-LDION	NU	N20		ΝŬ		K1		-				
x007										+				
×007		G		/ 100/0000				/// I		+				
X007	<mark> → -, vu3., </mark>			p(199/2000)	jWr	rite	P	2N						
×007 ←	<mark> .cnv </mark>				[Wr	rite	<mark>a]]</mark>	<u>(2N</u>						
x007 Ladder HELPHH-		<mark></mark>			Wr	rite	(리)	<u>(2N</u>						
×007 ←		<mark></mark>			<mark>/Wr</mark>	rite	<mark>a]]</mark>	2N					V	
x007 Ladder HELPHH-		<mark></mark>			[Wr	rite	역	2N			WOPC-F	XGP/WIN-	▼ E	
all the files	are stor	<mark></mark>			Wr	rite	(-]	©N 	, Sav	S			♥ E IN¥NEW.PM	1w?
x007 Ladder HELPHH-	are stor	<mark></mark>			[Wr	rite	(व		Sav	S	iges to	C:¥FXGPW	IN¥NEW.P№	1₩?
all the files	are stor	<mark></mark>			Wr	rite	데]) Sav	S		C:¥FXGPW		1w?
all the files	are stor	<mark></mark>			jwr	rite	9) Sav	ve Chan	iges to	C:¥FXGPW	IN¥NEW.P№	1₩?
all the files	are stor	<mark></mark>			Wr	rite	(-)		Sav	ve Chan	iges to	C:¥FXGPW	IN¥NEW.P№	1w?
all the files	are stor	<mark></mark>			jwr	rite			Sav	ve Chan	iges to	C:¥FXGPW	IN¥NEW.P№	1₩?
all the files	are stor	<mark></mark>			Wr	rite			Sa	ve Chan	iges to	C:¥FXGPW	IN¥NEW.P№	1w?
all the files	are stor	<mark></mark>			jwr	rite			Sav	ve Chan	iges to	C:¥FXGPW	IN¥NEW.P№	1₩?
all the files	are stor	<mark></mark>			Wr	rite			Sa	ve Chan	iges to	C:¥FXGPW	IN¥NEW.P№	1₩?
Ladder HELP J_HH- J_HH- All the files	are stor	<mark></mark>			<u>(</u> wr	rite			Sav	ve Chan	iges to	C:¥FXGPW	IN¥NEW.P№	1w?

Note that the Exit command is valid only for the active application window. If two or more programs are activated, end the programs by activating the application windows.



6.2 Edit Menu Functions

The Edit menu has various commands as shown below. Note that the function of the same command may be changed when the active window is changed.

- (1) Cut-and-paste functions used for ladder diagrams, instruction lists, etc.
- (2) Cut-and-paste functions used for comments in character strings.
- (3) Line insert and delete functions used for ladder diagrams
- (4) All edit functions for NOP instruction used in editing lists
- (5) Comment storing and edit functions



Note

<How the cut & paste function works in the Windows environment>

- In the Windows environment, various editions are allowed using the cut & paste, copy and deletion functions. The outline of such operations are as shown below.
- 1) \dashv \vdash -1/-2) 3) 4) ᅪᡘᆜ 5) 6) ┥ᢩᠰ –|↓⊦ 7) -|↓|-8) -l↑|--[9) ()} 10)
- 1) a-contact [LD, AND]
- b-contact [LDI, ANI]
 Parallel a-contact [OR]
- 4) Parallel b-contact [ORI]
- 4) Paraller D-contact [ORI]
- 5) Load/And pulse contact [LDP, ANP]
- 6) Load/And falling pulse contact [LDF, ANF]
- OR pulse contact [ORP]
- OR falling pulse contact [ORF]
 - 9) OUT coil [OUT]
 - 10) Special coil [SET, RST, or applid instruction, etc.]
 - 11) Horizontal line
 - 12) Vertical line
 - 13) INV instruction [INV]
 - 14) Vertical line deletion

* Because data transferred to the clipboard is expressed in a special format, its contents are not displayed even if the clipboard is opened.



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1		1	1		1	1			

6.2.1 Undo

Undo the last operation to return to the previous condition.

6.2.2 Cut

Cuts the selected program or characters. The cut program or characters will be stored to the clip board, therefore, paste the cut program or characters to another place.

6.2.3 Copy

Copies the selected program and characters to the Windows clip board. The copied data can be pasted to another place using the Paste command.

6.2.4 Paste

Pastes the programs and characters to the cursor position only if the data is stored to the Windows clip board via the Cut or Paste command.

6.2.5 Delete

Deletes the selected program and characters. The deleted data will not be stored to the Windows clip board. For this reason, the Paste command will not be valid for deleted data.



Note

The following command will not be valid in various windows.

- Undo : Invalid in the SFC diagram edit windows.
- Copy : Invalid in the device name comment and SFC diagram edit windows.

Cut : Invalid in the SFC diagram edit window.

Paste : Invalid in the SFC diagram edit window.



1

В

С

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~									

6.2.6 Line Delete

Use this function to delete the right side of the cursor position.

After deleting, enter a new program into the null section, or edit the ladder circuit so that the ladder circuit block is valid. After that, click Convert.

Note, this command cannot be used for any one-line ladder circuit block in the program step-completed status. To delete such one-line ladder circuit block, two or more ladder circuits, use the Block Select command described later, and then execute Delete.

-					b	adder						•
171	M8000	M500								-C 1050 >		+
		MSOL								_C Y021 2		
		M502								-(Y022)		
		M503										
		MS04								-C Y029 D	41-44	
		\rightarrow								-C Y024 >	4144	
		MSOS								-C Y025 >	***	
190	M43	1901								-C Y010 2	00	
	X022		V002							_C Y008 3	<u>−</u> 1	
	H32											' II
196	HS06								EALTP	H506 3		
200	-17 M8000									-C M507 >		
202	-11-				 		[CMP	K2	DO	H270 3		
210	X025		M42							-C M510 >		
	H510	T200							-OPLS	1511 7		
218	H510								CT200	K100		
218	T200	X026	Y000	H348						1		
222	-11-	117	W.	W	 			_		-C Mds 刘		+

		Final Alexandrian Control of Cont			
171	M8000	Ladder	(Y020 3	•	+
		HEOL	C Y021 2		
		N502	(Y022)		
		NS03	-C Y029 2	41 41	
		N504	(Y021 >	41 41	
		N505	C Y025 >	重度	
190	M43	Not Contract of the second sec	—C Y010 >	<u>•</u> •	
	X022			<u>- 18</u>	
196			M506]		
200	H506		(H507 >		
202	M8000	маз [Сяр кс D0	N270 3		
210	M510	7200 JF	-C M510 >		
	H510	PLS [PLS	M511]		
218	T200				
222 •	-11-			Ŀ	•

➡

The ladder circuit will be determined after conversion.



Remedy for Ladder conversion error

If a circuit error is detected after executing the ladder circuit conversion, correct the circuit, then convert the circuit to determine the Ladder program.

To delete the circuit data you edited and to cancel the ladder diagram editing mode, click Edit and then Edit Cancel. Or execute Edit Cancel using the right button on the mouse. Note that after executing Edit Cancel, all the circuit data that was edited in the ladder diagram window will be deleted if the data is not converted.



Execute Edit and then Line Delete.

In this condition, execute Tools and then Convert if the intention to only delete are line. If the circuit is not complete, complete the circuit first, then execute Tools and then Convert.

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~									

6.2.7 Line Insert

Use this command to insert a null line above the specified position.

	0						Ladder						-	
	134	H56	M515									-C Y011 >		+
	187									-DHUND	K2	DO 3		
	143	MS01	M502	MS09	MS04	MSOS						(M500)		
	149							-ESFTUP	H500	MSOL	KS	кі з	alad	
	160	H31	M50					[SFTRP	H500	M501	K5	кі з	41 44	
	171	M8000	M500									-C 4050 >		
			MSOL									-C Y021 3	00	
			M502									-C Y022 3	<mark>≁ ke</mark>	
100 101 102 101 101 101 102 102 102 102 102			M503									-(Y029)		
100 101 102 102 103 <td></td> <td></td> <td>HHF-</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>-C Y024 ></td> <td></td> <td></td>			HHF-									-C Y024 >		
139-1			M505									-C Y025 >		
	190	M42	1001									-(Y010)		
196-11		×032		V002								-C Y009 3		
200 17 (1507)	196	H32									-CallTP	H506 3		
*		MS06										-C M507 2		
	ŧ												-	•

Position the cursor on the line. A null line will be inserted just above the cursor-positioned line.

					Ladde	a'					
4H56	MSIS									-C Y011	>
								DUNP	K2	DO	3
MS01	M502	M509	MS04	MSOS						-C M500	
-	- ⁰										Del .
190	M51										11-44
9						{%FTLP	M500	M501	KS	KI	
-TH-						-CIFTRP	H500	M501	K5	К1	3 11 11
1 M8000	MS00										3 <mark>9 9</mark>
	MSOL									-C Y021	> 7 ke
	M502									-C Y022	1
	M508										1
	M504										1
	H H H									-C Y024	1
1413	1001									-C Y025	2
-i-i-										-C Y010	2
X032		Y002								-C Y009	2
M32									-CALTP	H506	
1											7
X032	21	VODP JA							-Cal.TP	-C Y009	1

Execute Edit and then Line Insert.

A null line will be inserted directly above the cursorpositioned line. To insert two or more null lines, simply repeated Line Insert.

Enter a necessary circuit in the null line, then execute Tools and then Convert. The circuit will be determined after conversion.



Line insert into step 0

If a null line is in front of step 0 or at the top line of the Ladder program, be sure to enter new program there, then convert the ladder diagram of the circuit.

If there is a null section, execution of conversion will cause a circuit error. If a circuit error is displayed after the conversion, click the control menu box at the upper left corner of the window to close the window, then open the window to display the ladder diagram again.

Be careful! When you close the ladder diagram window, all the edited data will be erased if it is not converted.



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~									

6.2.8 Block Select

Use this command to specify a ladder circuit block above or below the cursor position. This command can also be executed using the (1) + (1)/(1) keys or (1) + (1)/(1) keys + mouse dragging.

For the specified block, the following commands will be valid:

- Edit \rightarrow Cut : Cuts out the specified block, and then stores it in the clip board.
- Edit \rightarrow Copy : Copies the specified block in the clip board.
- Edit \rightarrow Delete: Deletes the specified block.



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~									



- Position the cursor on a line to specify a block.

Execute Up.

The cursor-positioned block will be specified.

		Ladder					*
					-OPLS	M55 ;	3
537 H52 H54						-C Y041	>
			-ECMP	кэ	DO	M160	41-44 41-44 41-44
MECCO	міет маг			-(MOVP	Кđ	DO	
555	M164 H36		-[049	K4 —(MOVP			
M8000			-COMP	K12	K12	D0	
	H167 H38			(MOVP	К4	DO	3
	Liji		-ECMP	EMOVP K13	K19 D0	D0 M169	3
	M170 M156 V001			-(MOVP	K12	DO 3	

Execute Up again.

The block just above the currently-specified block will be added to the currently-specified block. Repeatedly execute the Up command, to increase the specified block area in the upper direction (lower direction if the Down command is executed).



At the completion of block specification, execute Cut, Copy, or Delete command. When the Cut or Delete command is executed, the cut or deleted result will be stored in the program software, therefore, it is not necessary to convert the program.

v •

★ 1 ○ 載寺 左 ÷

K4 D0 K13 D0 D0 M16



T8 T8

M16

M170 M156 Y001

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
			1							

6.2.9 NOP Overwrite

- NOP overwrite	
Range O All range	OK Cancel
● Setting range 5 to 41	<u>H</u> elp

Use this command to overwrite the NOP instructions in the specified range of steps at a time. All the instructions currently stored in the specified range will be replaced with the NOP instructions.

6.2.10 NOP Insert

- NOP insert	
Range	<u>QK</u>
O <u>A</u> ll range	<u>C</u> ancel
● Setting range 31 to 67	<u>H</u> elp

Use this command to insert NOP instructions to the specified range of steps. When NOP instructions are inserted, the instructions currently set in the specified range will be shifted in the lower direction depending on the specified number of steps.

In addition, the block and coil comments will be shifted the instructions.

If the number of steps exceeds the specified upper limit number, the insertion of NOP instructions will not be possible.

1

2

3

4

5

6

7

8

9

Α

В

С

D

Ε

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
			1							

6.2.11 NOP Remove

NOP remove	
Range	<u>Q</u> K
● <u>All range</u>	<u>C</u> ancel
○ Setting range 67 to 67	<u>H</u> elp

Use this command to erase all the NOP instructions from the entire program area or the specified range of the steps. The non- erased instructions will be shifted in the upper direction.

The step numbers of the non-erased instructions will be reduced depending on the number of erased steps.

In addition, the block and coil comments will also be shifted following the shifted instructions.



Note

Specify the step range using the cursor on the list edit screen, thus, the specified step range will be automatically displayed in the Setting Range box of the corresponding dialog box.

-			Instruction	•
67	ZRSTP	D0 D4	99	+
72	ZRST	C0 C1	99	
77	MOVP	KO V		
82	MOVP	KO Z		
87	MOVP	K2 Di		
92	LD	M405		
93	MOVP	<u> </u>	NOP overwrite	
98	LD	N _{rRange}		
99	ANI			
100	OUT	○ <u>A</u> ll range	Cancel	
103	OUT	 Setting ran 	ge 72 to 87	
106	LD	1	<u>Help</u>	
107	ANI			
108	MPS			+
+ <u> </u>				→

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~			~						

6.2.12 Device Name

Use this command to store the name of the selected device. To store a device name, position the cursor on the device and then execute Edit and then Device Name.

	Input device nam	е
Device :X025		
START.SW		
OK	<u>Cancel</u>	<u>H</u> elp

- Number of characters: Up to 8 characters can be used for the device name.
- Character type: Alphanumeric characters and symbols (+ * / = .? # \$ % & :; _)
- Applicable devices: X, Y, M, S, T, C, D, P, and I.

In programming mode, device names can use in place of the device numbers, such as X000 and M10. For this reason, use of the same names for two or more devices will not be possible. You can also store the device name by sequentially executing View, Comment View, and then Device Name.



Device comment and Device name

• Add device comments to the devices held in the PLC, and use the device comments when they are displayed in the programs or printed.

Example: Device comments displayed Device comments Device numbers (Y002) START SWITCH OUTPUT Device comments

• The device names can be used in the programs in place of the device numbers.



Device names and device comments are added to the device numbers. For this reason, when a step is subject to change or the cut-and-paste command is executed, device names and comments will be erased or moved together with the respective device numbers.

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~			1						

6.2.13 Device Comment

Use this command to store the device comments for the selected device. To store a device comment, position the cursor on the device in the ladder diagram display window, and then execute Edit and then Device Comment.

-	Input device comme	ent
Device :X025		
Start switch		
<u>O</u> K	<u>C</u> ancel	<u>H</u> elp

- Number of characters: Up to 50 characters can be used for the device comment.
- Applicable devices: X, Y, M, S, T, C, D, P, and I.
- After inputting ladder circuit symbols on the ladder circuit edit screen, the device comment input window can be displayed consecutively. (Refer to Section 6.8.10.)
- Device comments can be edited in a list by choosing [View (V)] → [Comment (C)] → [Device Comment (D)].
- A device comment executed by this command is controlled on the peripheral equipment side. When making a comment to be written to the PLC, observe the following cautions, and refer to section 1.4 and section 6.3.9.
- Cautions on inputting a comment to be written to the PLC
 - When data should be shared by the A6GPP, A7PHP and programming software in the DOS version, input a comment in the following range. Number of characters :15 Character type : Numbers, alphabets and some symbols (., -, +, ? and =)
 - 2) If a comment is displayed in peripheral equipment not supporting characters of the comment, such characters may be displayed abnormally. If a comment exceeds 16 characters, only the portion up to 16 characters is written to the PLC, and the 17th character and later are ignored.

5 6 7 8 9 A B C D

E

1

2

3

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~									

6.2.14 Coil Comment

Use this command to store coil comments for the selected output and application instructions. To store a coil comment for an output or application instruction in the ladder program edit window, position the cursor on the instruction or in front of the instruction, and then execute Edit and then Coil Comment.



- To store a coil comment, use any number of characters and any character type.
- Enter coil comments at the completion of a ladder circuit conversion.

If the ladder circuit conversion is not complete, no coil comment can be entered.

 To copy a coil comment, specify the copy range and then execute Ctrl+C (cut). After that, execute Ctrl+V (paste) in the copy-to comment dialog box.

6.2.15 Block Comment

Use this command to store block comments for the cursor-positioned ladder circuit block. To store a block comment for a ladder circuit block in the ladder program edit window, position the cursor at any position of the ladder circuit block, and then execute Edit and then Block Comment.

Input block comment					
Step :23					
Counter coincidenc	e signal				
<u>O</u> K	<u>C</u> ancel	<u>H</u> elp			

- To store a block comment, use any number of characters and any character type.
- Block coil comments at the completion of ladder circuit block conversion.

If ladder circuit block conversion is not complete, no coil comment can be entered for the ladder circuit block.

 To copy a block comment, specify the copy range and then execute Ctrl+C (cut). Then, execute Ctrl+V (paste) in the copy-to comment dialog box.



Coil and block comments can be added to step numbers if the steps include the corresponding coils or blocks. For this reason, if a step is subject to change due to a coil or block insertion or deletion, the corresponding coil or block comment will be also inserted or deleted. In addition, if an instruction step or ladder circuit block itself is deleted, the comments stored in the step or block will be deleted at the same time.



1

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
									~	

6.2.16 Device Registration

Use this command to store device numbers in the device store/monitor screen. To store a device number, position the cursor at a desired position on the device monitor screen, and then execute Edit and then Device Registration. The device number will be stored in the cursor position on the screen.

Enter the device number to be stored, changed, or deleted.



You can store two or more device numbers starting from the number set in the Device box. Data entry range: 1 to 48. Click this button to store the device number in the monitor screen.

Deletes the device number from the Device box. The data set in the Number of Devices box, however, will not be deleted.

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	✓									

6.2.17 Edit cancel

Use this command to delete the ladder circuit data that was edited in the ladder program edit window.

Before editing

(The Ladder circuit is already converted, therefore the Ladder circuit determined.)

-	LAD_MONI.PMW:Ladder			-	•
	0	[RST	DO	Э	+
		-[RFT	Dl	Э	
	10002 	K 500	ро	Э	
	[nov	K350	Dl	Э	
	[INO/ K572620542	D10		3	+
+				+	



Editing ladder diagram

0	LAD_MONI.PMW:Ladder			•	•
		-[RST	во	Э	+
		-[RST	Dl	Э	
	10002 [Hav	K 500	DO	Э	
	[INCV X250	Dl		Э	
	[IMOV K572626542	D10		Э	¥
+				+	·



Click Edit and then Edit Cancel.

After edit cancel

(The ladder diagram will return to the pre-editing condition.)





6.3 Tools Menu Functions

Use the commands in the Tools menu to enter symbols in ladder or SFC diagrams or to convert ladder circuit blocks in the diagrams.



Note

For preferential keyboard operation, use functions keys that has the same functions as the Tools menu commands.

Circuit symbols, instructions, and SFC diagrams are allocated to the functions keys, and these functions keys depend on the edit window.

_					SWOPC-	-FXGP/V	VIN-E - O	COM	MENT.PN	1W				•						
		ols <u>S</u> e					Moniter/	ſest	<u>O</u> ption	<u>W</u> indow	<u>H</u> elp									
	98	200	ðC		80	9 66 6	8													
							Ladder							•						
c	X000	X002	¥0	20									C Y013	1 1						
	Y013										 ₩	-CT10	K200							
														1				otina	tool bo	
	V013												020	1			FIO	aung	tool bo	x
13	¥013										_0 ात	K1800	D1	3						
											- I 7 M	-[SET	M30	3						
												-[PLS	M95	3						
													C Y000	~						
	X002										Ewo.		-	1						
23	X005	¥000									[MC	NO	M50	-						
27														>						
- 30												[MCR	NO	3						
92	X006											-[C]	PO	3						
36	M8000												C Y020							
													C V021							
														1						
													C V022							
+													C 8000	+						
Ladde		uL 1	1.1.				(44/2000)	Writ	e	FX	2N								
<u>Jacur</u>	<u>an </u> a			<u>]</u> 777]?	<u>(rja</u> r		.0													
		7																		
Con	vert	kov					K		Circ	suit o	umh		ntry ke	~~						
COL	vert	кеу					/		Ond	Juit 5	ynno			∍y						
							•													
			Г																	
										1				1 .	_		-			
SHIF	T +	1		ł	' <u> H</u>	#	RINC	5	<u>4 P</u>	<u> 41</u>	<u>ہ</u> ا		,1 DEL	,		10				
	_																			

Function keys in ladder program edit window

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1									

6.3.1 Contact

Use this command to enter the a-contact and b-contact. Note that this command has the same function as function keys F5 (-I I-), F6 (-I/I-), F2 (-I1-), and F3 (-I1-).

_	SWOPC	-FXGP/WIN-E	
<u>File</u> <u>E</u> dit	<u>Tools</u> <u>S</u> ear	rch <u>V</u> iew <u>P</u> L	Click this command to enter the a-contact.
Option M	Co <u>n</u> tact C <u>o</u> il		Click this command to enter the b-contact.
	Eunction Wire	-P-	Click this command to enter the rise contact.
	All clear	└─┉─┐◀〜 ┨	Click this command to enter the fall contact.
	<u>A</u> ll clear	-	
	<u>C</u> onvert		

Applicable devices

X (input relay) M (auxiliary relay) T (timer) Y (output relay) S (state) C (counter)

	Input device
⊣⊢	\Box Keep this dialog box
XO	
<u>0</u> K	Reference Cancel Help

Entering a-contact

Enter the device mnemonic and device number.

_	Input device
**	\Box Keep this dialog box
XO	
<u></u> K	Reference <u>C</u> ancel <u>H</u> elp

Entering b-contact

Enter the device mnemonic and device number.

_	Input device
-#1-	□Keep this dialog box
XO	
<u>O</u> K	Reference Cancel Help

Entering rise contact

Enter the device mnemonic and device number.



Entering fall contact

Enter the device mnemonic and device number.

9 A B C D

1

2

3

4

5

6

7

8

• In making program, an device name can be used instead of a device No. (Refer to Section 7.2 and "Note" in the next page.)



С	DPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
		1									



Entering contacts using Reference command

Click the Reference command. The following dialog box will appear on the display. In this dialog box, the device names already stored will be displayed. Click a device name using the mouse button to enter the device.





1

2

3

4

5

6

7

8

9

Α

В

С

D

Ε

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~									

6.3.2 Coil

Use this command to enter output coil instructions. Note that this command has the same function as function key [F7] (-()-).

Applicable devices: Y (output relay) M (auxiliary relay) S (state) Input device Input device Keep this dialog box Entering timer and counter Applicable devices: T (timer) C (counter)	Input device	Entering bit device
► □Keep this dialog box Applicable devices: T (timer)		M (auxiliary relay)
T (timer)		
	Input device	Entering timer and counter

- Use device names in place of the device numbers (see Sec.7.2 and the following Note).
- Use the Function command (as described in Sec.6.3.3) to enter the following special coil instructions: SET/RST PLS/PLF MC/MCR STL/RET END

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	✓									



Entering coils using Reference command

Click the Reference command. The following dialog box will appear on the display. In this dialog box, the device names already stored will be displayed. Click a device name using the mouse button to enter the device.





OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~									

6.3.3 Function

Use this command to enter application and special coil instructions. Note that this command has the same function as function key [F8] (- [] -).

_	Input instruction
-{}-	□Keep this dialog box
SET Y5	
<u></u> K	Reference Cancel Help

Entering special coil Applicable instructions: SET/ RST PLS/ PLF MC/ MCR STL/ RET END
Formats for data entry: SET[space] Y5 PLS[space] M200 MC[space] N0[space] M30 STL[space] S20 For the following instructions, it is not necessary to enter device numbers: RET, END
Entering application instructions Formats for data entry: MOV[space] K200[space] D0 (16-bit MOV instruction)
MOVP[space] K200[space] D0 (16-bit MOV instruction of pulse execution type)
DMOV[space] K40000[space] D0 (32-bit MOV instruction)

<u> </u>	Input instruction				
ፈን	\Box Keep this dialog box				
MOV K200 D0					
<u></u> K	Reference Cancel Help]			

_	Input instruction					
-{}-	□Keep this dialog box					
MOVP K200 E	10VP K200 D0					
<u></u> K	Reference Cancel Help					

	Input i	nstruction				
ፈን	-CF CKeep this dialog box					
DMOV K40000	DO					
<u>O</u> K	<u>Reference</u>	<u>C</u> ancel	<u>H</u> elp			

9 A B C

1

2

3

4

5

6

7

-{}-

FNC12 K200 D0

Ok

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~									

	Input instruction
-{}-	\Box Keep this dialog box
DMOVP K40	000 D0
<u>K</u>	Reference Cancel Help

Input instruction

<u>Cancel</u>

<u>H</u>elp

□Keep this dialog box

<u>R</u>eference

DMOVP[space] K40000[space] D0 (32-bit MOV instruction of pulse execution type)

FNC12[space] K200[space] D0 (16-bit MOV instruction)

FNC12P[space] K200[space] D0

(16-bit MOV instruction of pulse execution type)

Input instruction
 Input instructin
 Input instruction
 Input instruction
 Input instruction

_	Input in	nstruction					
-{}-	-C- CKeep this dialog box						
FNCD12 K400	00 D0						
<u></u> K	<u>R</u> eference	<u>C</u> ancel	<u>H</u> elp				

_	Input instruction					
-(}-	\Box Keep this dialog box					
FNCD12P K4	40000 D0					
<u></u> K	Reference Cancel Help					

FNC12[space] K40000[space] D0 (32-bit MOV instruction)

FNC12[space] K40000[space] D0 (32-bit MOV instruction of pulse execution type)

1

2

	OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
1		1									



in this box.

Entering coils using Reference command:

an instruction in this box. In addition,

Click the Reference command. The following dialog box will appear on the display. Using this dialog box, you can select instructions from the list, or look up the functions of operands.



🖈 MITSUBISHI

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	✓									

6.3.4 Wire

Use this command to enter vertical and horizontal lines, INV instruction, or delete vertical lines.

Note that this command has the same function as function keys (Ishift) + F9 (I), F9 (-), (Ishift) + F7 (-/-), and (Ishift) + F8 (I Delete).





Entering vertical line

A vertical line will be entered at the lower left section of the cursor.



Entering horizontal line

A horizontal line will be entered at the cursor position.

C Y010)
C Y010 >

Entering INV instruction

An INV instruction will be entered at the cursor position.



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	✓									



Deleting vertical line

A vertical line will be deleted from the lower left section of the cursor.

2
3
4
5
6
7
8
9
Α
В
С
D
E

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
			~							

6.3.5 Instruction

From the list edit screen, directly enter instructions and device numbers using the keyboard. Use the Instruction command to edit programs using lists or references displayed in the dialog box.



Describes the necessary operand for each instruction type. The system enters the data entry waiting status.

Specify operands, or enter device names.

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1		~			1	~		1	

6.3.6 All Clear

Use this command to clear the active work window.

- Ali clear	Clearing program area
Program clear <u>(OK)</u>	Applicable windows:
<u>C</u> ancel	Ladder program edit window
<u>H</u> elp	List program edit window
Comment all clear □ Device name QK ☑ Device comment Cancel □ Coil comment Lep ☑ Block comment Help	Clearing comment area Applicable windows: Device name comment window Device comment window
	To erase coil comments and block comments, open the device name
Specify the item to be	comment window or device comment window.
erased by clicking the	Note that the set comment data to be written in the PLC will be erased
button.	also.
Register all clear Data register RAM file register RAM file register Each file register RAM file register Help Specify the item to be erased by clicking the button.	Clearing register area Applicable windows: Register window When data registers are displayed When file registers are displayed When RAM registers are displayed

Ε

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1			~						

6.3.7 Convert

Use this command to convert ladder or SFC program data into list program.

Note that this command has the same function as function key F4 (Convert).

• If a ladder or SFC program created on screen is not converted, it is only a diagram, therefore it is not valid.

To make the diagram valid, the diagram should be converted into a program (list program). If the list edit function is used for programming, thus, conversion will not be needed. This is due to the list edit mode, that requires a direct creation of instruction list by entering instructions.

- Convert the diagram when:
 - 1)The created data of the ladder or SFC diagram is checked.
 - 2) After creating a ladder or SFC diagram convert the diagram into instruction list, then edit the list.
 - 3) Close the ladder or SFC program edit window when minimizing the window, conversion will not be needed.
 - 4) Finish programming with FXGP/WIN-E by executing Close or Exit.
- Be careful! Non-converted ladder or SFC diagrams may be erased if an operation error is made.
- When "Convert and run time program changes" is selected via "Option → Preferences", "Run time program changes" is executed after "CNV F4" or "Tools → convert" is executed.



Conversion of ladder or SFC diagram into instruction list:



OP	Ν	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
							1				

6.3.8 Default comments for special M/D

Use this command to store default comments in the special auxiliary relay comment area (M8000 and onwards) or the Diagnostic register comment area (D8000 and onwards) using the device comment/name window.

If device comments are already stored in the specified area, executing this command will overwrite the default comments, therefore, device comments previously stored will be erased.



Example:Comments stored in special/diagnostic relay comment area

-	LAD_MONLPMW:Comment		- -
Device	Device comment	Name	1
M8000	RUN monitor NO contact		
M8001	RUN monitor NC contact		
M8002	Initial pulse NO contact		
M8003	Initial pulse NC contact		
M8004	Error occurrence		
M8005	Low battery voltage		
M8006	Low battery voltage latch		
M8007	Momentary power failure detected		
M8008	Power failure		
M8009	24V DC OFF		
M8010			
M8011	10 msec clock		
M8012	100 msec clock		
M8013	1 sec clock		
M8014	1 min clock		
M8015	Inner clock		
M8016	Inner clock		
M8017	Inner clock		
M8018	Inner clock		

2
3
4
5
6
7
8
9
Α
В
С
D
E

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
						~				

6.3.9 Comment settings

This command is used in the device comment/name window to specify the device comments that are to be written to the PLC.

Start <u>E</u> nd	Setting of com Y000 Y025	ment move	ed to PLC Entry Clear Close	 Sets the specified device comments as the data to be written to the PLC. After executing, this button, an asterisk * will appear next to the specified device numbers.
				Possts the specified device comments. This means that

Resets the specified device comments. This means that the specified device comments will be regarded as the comments not be written to the PLC.

- To write the device comments to the PLC, 1 block or more of the comment area should be secured. To secure the comment area, click Option and then Parameter Settings.
- Enter the same type of device in the From and To boxes. To specify two or more types of devices, such as input (X) and output relays (Y), select the devices for each type of device.
- If the same device number is entered in the From and To boxes, only one device comment will be set.
- To specify non-continuous device numbers, execute this command repeatedly. Or in the device comment/ name window, click the right button of the mouse on each device.
- Specifying a device that does not have a device comment, allows for the number of set device comments to be increased, and the free storage area will be reduced for the device comments. For this reason, do not specify any unnecessary devices.
- Cautions on inputting a comment to be written to the PLC
 - When data should be shared by the A6GPP, A7PHP and programming software in the DOS version, input a comment in the following range. Number of characters :15
 Characters transmission of the state and examples (see a character).
 - Character type : Numbers, alphabets and some symbols (., -, +, ? and =)
 - 2) If a comment is displayed in peripheral equipment not supporting characters of the comment, such characters may be displayed abnormally. If a comment exceeds 16 characters, only the portion up to 16 characters is written to the PLC, and the 17th character and later are ignored.

Operation in device comment/device name window:

To specify device comments to be written to the program area of the PLC, click Tools and then, the setting of Comment Move to PLC. Or in the device comment/name window, click the right button of the mouse at the cursor position or after specifying the range.

After specifying the device comments to be written to the PLC an asterisk * will appear next to the corresponding device numbers.

When comments are transferred from the program area (where comments are read from the PLC or files are stored) to the device comment area, an asterisk * will appear next to the corresponding device numbers. (see Sec. 6.1.2 and Sec. 6.6.1).

- Eile	Edit Tools Search View PLC Remote Monitor/Test	<u>Option Window H</u> elp :
] 6 () 42 () 4 () 2 () 10 () 1	
Device	Device comment	Name
Y000*	LAMP No.1	LAMP1
Y001*	LAMP No.2	LAMP2
Y002*	LAMP No.3	LAMP3
Y003*	By1	NG.LAMP
Y004*	RY2	OK.LAMP
Y005*	Ry3	TEST
Y006*	Ry(POWER)	RY1
Y007*	ERROR No.1	RY2
Y010*	ERROR No.2	RY3
Y011*	ERROR.No.3	BUZZER
Y012*	MC1	LED1
Y013*	MC2	LED2
Y014*	MC3	LED.Y
Y015*	MC4	LED.R
Y016		LED.B
Y017		LEDG
Y020		
Y021		
Y022		
Y023		
Y024		
Comme	ent Device comment(14/ 50)	FX2N / FX2NC

Indicates the device comment storage capacity. (Unit: comments)

Indicates the number of set comment devices that is to be written to the PLC. (Unit: comments)



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
						~				

6.3.10 Setting of step of comment

Use this command to correct the comment step indication in the block comment window or coil comment window (see Sec. 6.5.5).

After adding or deleting a program, the comment step numbers will be automatically adjusted, but the brackets showing the block/coil comment steps will not move automatically together with the step numbers.

For this reason, to enter comments in the block or coil comment windows after changing a program, be sure to execute this command to correct the bracket positions.

6.4 Search Menu Functions

The command functions of the Search menu can be classified into the following types:

- (1) Commands used for jumping to another place in the work window, such as Go to Top and Go to end and Tag jump
- (2) Commands used for searching comment character strings
- (3) Commands used for searching programs using device names, device numbers, and instruction symbols
- (4) Commands used for replacing character strings and device numbers
- (5) Commands used for inverting contact logic or device numbers



Relation between convert and search

Convert the diagram before using a search command in the ladder or SFC diagram editing status. If the ladder or SFC diagram is not converted, the diagram will not be valid as the program. For this reason, if a search command is preformed, a message will appear on the display to request a conversion.



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	>	~	>	>	>	>	>			

Тор

End

Тор

6.4.1 Go to Top, Go to End

Use this command to move the cursor to the top or end of the document.

Example: Going to top or end of ladder diagram



Example: Going to top or end of instruction list

0	LD	X000		
1	OR	Y013		
2	ANI	X002		
3	OUT	Y013		
4	OUT	T10	K200	
7	OUT	C5	D20	
10	LD	Y013		
11	MOV	K1800	D1	
16	SET	M30		
17	PLS	M35		
19	LD	X002		

39	END			
40	NOP			
41	NOP			
42	NOP			
43	NOP			
44	NOP			
45	NOP			
46	NOP			
47	NOP			

End



1

2

3

4

5

6

7

8

9

Α

В

С

I)

Ε

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	✓	~	1	~	~					

6.4.2 Device Name Search

Use this command to search the specified device name in the program.

Enter a device name with single quotation marks in this box



Specify the search direction.

• Even if device names are not displayed on the screen, the corresponding device number will be searched.

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~	~	~							

6.4.3 Device Search

Use this command to search the specified device number in the program.

Enter the device number.

Applicable devices: X, Y, M, S, T, C, D, V, Z, P, and I

Click this button	to select a device number from the list.
Search Device X0	
Coption □Bit device group □Double word	
Direction I op to end Under the cursor Upper the cursor	——————————————————————————————————————

Even if the device numbers are not displayed on the screen, the device corresponding to the set device number will be searched upon.



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~	~	~							

6.4.4 Instruction Search

Use this command to search the specified instruction symbol.

Enter an instruction in this box.

Example: LD, AND, OUT, SET, MOV, DMOVP, FNC12, FNCD12P, etc.



To search the program by specifying an instruction and device number, use the Contact/Coil Search command as described on the next page.

6.4.5 Contact/Coil Search

Use this command to search the program for the specified contact symbol and device number.



Enter a device number in this box.


OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1	1	~							

6.4.6 Go to Step

Use this command to go to the specified program step.



Enter a step number in this box. Click this button to select a device number from the list.

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
				~	~					

6.4.7 Go to State

Use this command to go to the specified state in the SFC diagram edit window.



Enter the state number in this box.

Click this button to select a state number from the list.



1

7	
8	

В

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
						~				

6.4.8 Character Search

Use this command to search the specified character string.



6.4.9 Character Exchange

Use this command to find and replace the specified character string.



Click the Select button to sequentially find and replace the character strings. The dialog box will appear each time, and ask if replacement of the characters string is required. Click the Replace All button to replace all the character strings at one time.

Enter the character string you want to find.

2

3

4

5

6

7

8

9

Α

В

С

Ε

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1		1	1						

6.4.10 Change Device Number

Use this command to replace the specified device numbers to other numbers. Applicable devices: X, Y, M, S, T, C, and D





Devices cannot be replaced if the number of steps differ from each other.

- 1) M0 to M1535 ↔ M1536 to M3071 (For FX2N, FX2Nc series PLC family PLC only)
- 2) 16-bit counter↔32-bit counter (Common to the FX family PLC)

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		~	~						

6.4.11 Change Bit Device

Use this command to invert the logic of the contact instructions for the specified device numbers. Applicable devices: X, Y, M, S, T, and C



6.4.12 Swap Device Number

Use this command to swap the specified device numbers with each other. Applicable devices: X, Y, M, S, T, C, and D



device replacement to the same type. Examples:

 $X{\leftrightarrow}X,\ Y{\leftrightarrow}Y,\ M{\leftrightarrow}M,\ T{\leftrightarrow}T,\ D{\leftrightarrow}D,\ etc.$



Devices cannot be replaced if the number of steps differ from each other.

- 1) M0 to M1535 \leftrightarrow M1536 to M3071 (For FX2N, FX2NC series PLC controller only)
- 2) 16-bit counter↔32-bit counter (Common to the FX family PLC)



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~	~	~	~						

6.4.13 Tag Set

This command is used to set the tag for Tag Jump. In the ladder program edit window, the first step number in the cursor-positioned ladder circuit block will be stored as the tag. In the list program edit window, the cursor-positioned step number will be stored as the tag.

Stores step 75 as the tag.



Stores step 74 as the tag.

		Instruction	•	
70	OUT	M2		t
71	LDI	X003		
72	OUT	M3		
73	LDI	M8000		
- 74	OUT	M7		
75	LD	X007		
76	OR	M50		
77	ANI	M120		
78	OUT	M12		
79	OUT	M50		
80	LD	M8002		
81	OR	X007		
82	DTOP	KO		
99	LD	M8000		
100	MOVP	K20		
105	OUT	M8039		Ŧ
+			+	Ċ

6.4.14 Tag Jump

Use this command to jump to the tag position.

Select a tag from the list. Up to 10 tags can be stored in the list. If the eleventh tag is stored, then, the oldest tag is stored, then will be deleted.



6 7 8 9 A B C D E

1

2

3

4

5

6.5 View Menu Functions

The commands on the View menu enables the user to switch work windows or display the selected items.

Windows that can be switched by View menu commands



2

3

4

5

6

7

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		1	~		~	~	~	~	~

6.5.1 Ladder View

Use this command to display the ladder program edit window.



Functions of ladder program edit window:

- 1. Displaying and editing existing programs (see Sec.7.2.4)
- 2. Creating and editing new ladder circuits. (see Sec.7.2.4)
- 3. Storing and displaying comments (see Sec.6.2.12, 6.2.15, and 6.5.12)
- 4. Displaying ladder diagrams so that operator can monitor the operation of the PLC (see Sec.6.7.1)
- 5. Writing a program in run mode (see Sec.6.6.1)
- 6. Printing ladder diagrams (see Sec.6.1.6 and 6.1.7)

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		~	~		✓	√	~	~	✓

6.5.2 Instruction View

Use this command to display the instruction list program edit window.



Functions of list program edit window:

- 1. Display and edit existing programs (see Sec.7.3.2)
- 2. Creating and editing new instruction lists. (see Sec.7.3.2)
- 3. Displaying comments (see Sec.6.5.12)
- 4. Printing instruction lists (see Sec.6.1.6 and 6.1.7)

2

3

4

5

6

7

8

9

Α

В

С

I)

Ε

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	>		>	>		>	~	>	>	✓

6.5.3 SFC View

Use this command to display the SFC diagram edit window. The inner ladder program edit windows for ladder blocks, states, transitions can be also displayed.

SFC diagram

 $\overline{\cdot}$



HELP];], ,,CHV], ♀], ▼], ▽ JLADI], —],

Press the Shift key to display the sub-guidance.

	Inner ladder				
			—C 1000 >		
	 	CT0	К100 🔿		
			-C M100 >		
M100 			-C Y001 >		
X001 X002 Y000 X003			-C M101 >	<u> </u>	
2003			-C M102 >	<u>++++</u>	
				111 144	
M101 M102				**	
M101 M102			-C Y002 >	<u>• •</u>	
			—C Y003 🔾	≁ k a	
M102 M100 X005			-C Y004 >		
M0		[PLS	M0 3		
1		ESET	Y005]		
-0-		[PLF	м1 Э		
M1		[RST	Y005 🕽		
X006	 	[RST	C0]		
X007		(C0	К10 🔿		

Functions of SFC diagram edit window:

- 1. Displaying and editing SFC diagrams for the existing programs (see Sec.8.2.4)
- 2. Creating and editing new SFC diagrams (see Sec.8.2.4)
- 3. Displaying, creating, and editing inner ladder diagrams (see Sec.8.2.3 and 6.5.4)
- 4. Displaying comments (see Sec.6.5.12)
- 5. Displaying SFC diagrams so that operator can monitor the operation of the PLC (see Sec.6.7.2 and 6.7.3)
- 6. Printing SFC diagrams (see Sec.6.1.6 and 6.1.7)

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
				✓						

6.5.4 Inner Ladder View

Use this command to display the inner ladder program edit windows. In this window, the inner ladder program of ladder blocks, states, transitions are displayable.

Position the cursor on a SFC symbol in the SFC diagram edit window, then execute View and Inner Ladder View.

For a detailed description of the inner ladder program, refer to Sec.8.2.3 "Creating SFC diagrams" and Sec.7.2 "Creating Ladder program".

Execute View then the inner ladder view to display the inner ladder diagram of the cursor positioned symbol.



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		~	~		~	~	~	~	~

6.5.5 Comment View

Use this command to display the device name or comment edit window. Using these windows, store and edit the device names and comments.

•	N
L)
	/

Saving and reading comment files exclusively

A comment file can be saved with a different name. An existing comment file can be read into the program file that is currently opened.

When the comment window is selected, perform the following operation:

Saving a comment file: File \rightarrow Export

Reading an existing comment file: File \rightarrow Import

When importing or exporting a comment, the comment type to be processed can be selected. If there are comment types which should not be imported or exported, uncheck the check boxes of corresponding comment types.

Save as		×
File name C:\ABC.COW		ОК
File title		Cancel
		Help
File type		<u> T</u> eib
Device comment	Block comment	
Device comment	Coil comment	
V Device name	Coil comment	

When a program is saved or read, a comment file is usually processed together. The export or import operation is required when a comment file is needed to be processed independently. (See Sec.6-1-2, 6-1-4)



1

2

3

4

8

B

D)
F	

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1		1	1		1	1	1	1	~

Device Comment/Device name

The list of Device comments and names will be displayed in the window. Using this window, store, edit, or print the Device comments and names. For a detailed description of Device comments and names, refer to Sec.6.2.12 and 6.2.13.

Enter the device number to be displayed first in the list.



Click this button to display all the devices while ignoring the PLC model. The unstored device numbers in the PLC will be displayed in red.



Functions of Device comment and name edit window:

- 1. Display list of device comments already stored
- 2. Storing and editing device comments (In the ladder program edit window, click Edit and then Device Comments to individually store the device comments.)
- Storing and editing device names (In the ladder program edit window, click Edit and then Device Name to individually store the names.)
- 4. Printing device comments (see Sec.6.1.6 and 6.1.7)



Specifying comments to be written to PLC:

The device comments specified can be stored in the parameter-set program area of the PLC (see Sec. 1.4, Sec. 6.2.13, and Sec. 6.3.9).



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		 Image: A set of the set of the	1		1	~	1	1	~

Block Comment

The list of block comments will be displayed in the block comment window. For a detailed description of block comments, refer to Sec.6.2.15.

Enter the first device number to be displayed from the list.



The block comment storage positions are controlled by the first step numbers of the Ladder ladder circuit blocks, therefore use the first step numbers to display or store the block comments. After inserting or deleting a program, the step numbers will be automatically corrected. The brackets [] of the block comment step numbers, however, will not move automatically. To correct the bracket positions, click Edit and then Setting of Step of Comment (see Sec. 6.3.10).

Functions of block comment edit window:

- 1. Displays the list of block comments already stored.
- 2. Stores and edits block comments. (If you use the ladder program edit window, you can individually store each block comment by clicking an Edit and then Block Comment.)

1

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1		1	1		1	1	1	1	~

Coil Comment

The list of coil comments will be displayed in the coil comment window. For a detailed description of coil comments, refer to Sec.6.2.14.

Enter the device number to be displayed first in the list.



The coil comment storage positions are controlled by the step numbers of the output control instructions, therefore, use the step numbers of the instructions to display or store the coil comments.

After inserting or deleting a program, the step numbers will be automatically corrected. The brackets [] of the coil comment step numbers, will not move automatically. To correct the bracket positions, click Edit and then Setting of Step of Comment (see Sec. 6.3.10).

Functions of coil comment edit window:

- 1. Displays the list of coil comments.
- 2. Stores and edits coil comments.



2

3

4

5

6

7

8

9

Α

В

С

Ε

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		1	1		1	1	1	1	✓

6.5.6 **Register View**

Use this command to display the register edit window.



Click this button to display all the devices while ignoring the PLC model. The unstored device numbers in the PLC will be displayed in red.

Select 32 bit to display and edit data of from a 32-bit register.

To display 32-bit data, select Numbers. The 16-bit register device numbers of the selected type (even or odd numbers) will be determined as the low-

If the selected low-order side differs from that specified in the program, wrong values will be displayed on the screen.

Functions of the register edit window:

1. Display and editing data register data or file register data read out from PLC

2. Creating and editing data or file register data to be written to the PLC

OF	N	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
		1		1	1		1	1	1	1	~

Example: 16bit Register view

Binary data, table type

	-			Register			•
		+0	+1	+2	+3	+4	+
Double-click the device	DO	00000000 01000001	00000000 01000010	00000000 01000011	00000000 01000100	00000000 01000101	
	D10	00000000 01001011	00000000 01001100	00000000 01001101	00000000 01001110	00000000 01001111	
number column.The	D20	00000000 01010101	00000000 01010110	00000000 01010111	00000000 01011000	00000000 01011001	1.4
	D90	00000000 00000000	00000000 00000000	00000000 00000000	00000000 00010010	00000000 00010010	¶
display softing dislog	D40	00000000 00000000	00000000 00000000	00000000 00000000	00000000 00000000	00000000 00000000	1.4
display setting dialog	D50	00000000 00000000	00000000 00000000	00000000 00000000	00000000 00000000	00000000 00000000	¶
hav will annoar an the	D60	00000000 00000000	00000000 00000000	00000000 00000000	00000000 00000000	00000000 00000000	1. P
box will appear on the	D70	00000000 00000000	00000000 00000000	00000000 00000000	00000000 00000000	00000000 00000000	1. P
	D80	00000000 00000000	00000000 00000000	00000000 00000000	00000000 00000000	00000000 00000000	1-1
display.	D90	00000000 00000000	00000000 00000000	00000000 00000000	00000000 00000000	00000000 00000000	
	D100	00000000 00000000	00000000 00000000	00000000 00000000	0000000 00000000	00000000 00000000	1-1-1-
	D110	00000000 00000000	00000000 00000000	00000000 00000000	00000000 00000000	00000000 00000000	1-9 I
\backslash	D120	00000000 00000000	00000000 00000000	00000000 00000000	00000000 00000000	00000000 00000000	1-1-1
	D130	00000000 00000000	00000000 00000000	00000000 00000000		00000000 00000000	- <u>-</u>
	D140	00000000 00000000	00000000 00000000	00000000-00000000	00000000 00000000	00000000 00000000	•••
	+						+

You can edit the numeric values of the data cell.

Decimal data, table type

-					Re	gister					•
	+0	+1	+2	+3	+4	+5	+6	+7	+8	+9	t
DO	65	66	67	68	69	70	71	72	73	74	
D10	75	76	77	78	79	80	\$1	82	83	84	-
D20	85	86	87	88	89	90	23	898	0	0	
D90	0	0	0	18	18	210	425	1126	0	0	
D40	0	0	0	0	0	0	0	0	0	0	
D50	0	0	0	0	0	0	0	0	0	0	
D60	0	0	0	0	0	0	0	0	0	0	
D70	0	0	0	0	0	0	0	0	0	0	
D80	0	0	0	0	0	0	0	0	0	0	
D90	0	0	0	0	0	0	0	0	0	0	
D100	0	0	0	0	0	0	0	0	0	0	
D110	0	0	0	0	0	0	0	0	0	0	
D120	0	0	0	0	0	0	0	0	0	0	
D130	0	0	0	0	0	0	0	0	0	0	L
D140	0	0	0	0	0	0	0	0	0	0	ŧ
+				-							+

Hexadecimal data, table type

-							Registe	er		
	+0	+1	+2	+3	+4	+5	+6	+ /	+8	+9
DO	41	42	43	44	45	46	47	48	49	4A
D10	4B	4C	4D	4E	4F	50	51	52	53	54
D20	55	56	57	58	59	5A	17	382	0	0
D30	0	0	0	12	12	D2	169	466	0	0
D40	0	0	0	0	0	0	0	0	0	0
D50	0	0	0	0	0	0	0	0	0	0
D60	0	0	0	0	0	0	0	0	0	0
D70	0	0	0	0	0	0	0	0	0	0
D80	0	0	0	0	0	0	0	0	0	0
D90	0	0	0	0	0	0	0	0	0	0
D100	0	0	0	0	0	0	0	0	0	0
D110	0	0	0	0	0	0	0	0	0	0
D120	0	0	0	0	0	0	0	0	0	0
D130	0	0	0	0	0	0	0	0	0	0
D140	0	0	0	0	0	0	0	0	0	0
+ ÎÎ		~	Ŷ	~	<u>^</u>			^	^	^

ASCII data, table type

_									Ro	giste
_	0		1.0	1.13	1.1	1.16	1.6			<u>gisic</u> 1 +9
	+0	+1	+2	+3	+4	+5	+6	+7	+8	+9
DO	. A.	В	C	D	E	F	G	H	I	J
D10	K	L	M	N	0	P	Q	R	I S	T
D20	U	U V	M	X	Ι Y	Z				
D30				1		8		f		
D40										
D50										
D60										
D70										
D80										
D90										
D100										
D110										
D120				l	l	l		l		
D130					1		1			
D140			[
+				1	1	1	r	r	P	r

Line type

-					Register
	DEC	HEX	ASCII	BIN	
DO	65	41	A	00000000 01000001	
D1	66	42	В	00000000 01000010	
D2	67	43	С	00000000 01000011	
D3	68	44	D	00000000 01000100	
D4	69	45	E	00000000 01000101	
D5	70	46	F	00000000 01000110	
D6	71	47	G	00000000 01000111	
D7	72	48	н	00000000 01001000	
D8	73	49	Ι	00000000 01001001	
D9	74	4A	J	00000000 01001010	
D10	75	4B	K	00000000 01001011	
D11	76	4C	L	00000000 01001100	
D12	77	4D	M	00000000 01001101	
D13	78	4E	N	00000000 01001110	
D14	79	4F	0	00000000 01001111	
D15	80	50	P	00000000 01010000	
D16	81	51	Q	00000000 01010001	
D17	82	52	R	00000000 01010010	
D18	83	53	S	00000000 01010011	
D19	84	54	T	00000000 01010100	
D20	85	55	U	00000000 01010101	
D21	86	56	V	00000000 01010110	
D22	87	57	М	00000000 01010111	
+					→



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		~	~		~	~	~	~	✓

Example: 32bit Register view Binary data, table type

	Register	
+0	+2	+4
00000000 01000010 00000000 01	000001 00000000 01000100 00000000 01000011	0000000 01000110 000000
00000000 01001100 00000000 01	01011 00000000 01001110 00000000 01001101	00000000 01010000 000000
0000000 01010110 0000000 01	010101 0000000 01011000 0000000 01010111	0000000 01011010 000000
00000000 0000000 00000000 00	000000 0000000 00010010 00000000 0000000	00000000 11010010 000000
00000000 0000000 00000000 00	000000 0000000 0000000 0000000 0000000	00000000 00000000 000000
00000000 0000000 0000000 00	000000 0000000 0000000 0000000 0000000	0000000 0000000 000000
00000000 0000000 00000000 00	000000 0000000 0000000 0000000 00000000	00000000 00000000 000000
00000000 0000000 0000000 00	000000 0000000 0000000 0000000 0000000	00000000 00000000 000000
00000000 00000000 00000000 00	000000 0000000 0000000 0000000 00000000	00000000 00000000 000000
00000000 0000000 0000000 00	000000 0000000 0000000 0000000 00000000	00000000 00000000 000000
00000000 00000000 00000000 00	000000 0000000 0000000 0000000 0000000	00000000 00000000 000000
00000000 0000000 0000000 00	000000 0000000 0000000 0000000 0000000	00000000 00000000 000000
00000000 0000000 0000000 00	000000 0000000 0000000 0000000 0000000	00000000 00000000 000000
00000000 0000000 0000000 00	0000000 00000000 00000000 00000000 00000	00000000 00000000 000000
00000000 00000000 00000000 00	0000000 00000000 00000000 0000000 000000	00000000 00000000 000000
1 00000000 0000000 0000000 00		· ···································
		7
	'	
You car	edit the numeric value	es of the data
	call and marrierio value	

Double-click the device number column. The display setting dialog box will appear on the display.

Decimal data, table type

-			R	egister		•
	+0	+2	+4	+6	+8	
DO	4325441	4456515	4587589	4718663	4849737	
D10	4980811	5111885	5242959	5374033	5505107	
D20	5636181	5767255	5898329	58851351	0	
D30	0	1179648	13762578	73793961	0	
D40	0	0	0	0	0	
D50	0	0	0	0	0	
D60	0	0	0	0	0	
D70	0	0	0	0	0	
D80	0	0	0	0	0	
D90	0	0	0	0	0	
D100	0	0	0	0	0	
D110	0	0	0	0	0	
D120	0	0	0	0	0	
D130	0	0	0	0	0	
D140	0	0	0	0	0	
+		~				

Hexadecimal data, table type

-				Re	egister
	+0	+2	+4	+6	+8
DO	420041	440043	460045	480047	4A0049
D10	4C0048	4E004D	50004F	520051	540053
D20	560055	580057	5A0059	3820017	0
D30	0	120000	D20012	46601A9	0
D40	0	0	0	0	0
D50	0	0	0	0	0
D60	0	0	0	0	0
D70	0	0	0	0	0
D80	0	0	0	0	0
D90	0	0	0	0	0
D100	0	0	0	0	0
D110	0	0	0	0	0
D120	0	0	0	0	0
D130	0	0	0	0	0
D140	0	0	0	0	0
÷	AL				

1
2
3
4
5
6
7
8
9
Α
В
С
D
E
L

OP	NL	_/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
		✓		~	~		~	~	~	~	✓

ASCII data, table type

-					
	+0	+2	+4	+6	+8
DO	AB	СD	EF	GΗ	ΙJ
D10	KL	MN	0 P	QR	ST
D20	UV	WХ	ΥZ		
D30			x	⇒ f	
D40					
D50					
D60					
D70					
D80					
D90					
D100					
D110					
D120					
D130			[
D140					
4					

Floating-point data, table type

-			Register			-	ŀ
	+0	+2	+4	+6	+8		T
DO	1.51563275E0	1.53125799E0	1.54688323E0	1.56250846E0	1.5781337E0		h
D10	1.59375894E0	1.60938418E0	1.62500942E0	1.64063466E0	1.65625989E0		h
D20	1.67188513E0	1.68751037E0	1.70313561E0	7.64073391E-37	0E0		
D30	0E0	1.140625E0	1.92854794E-38	2.70371324E-36	OEO		
D40	OEO	0E0	0E0	OEO	0E0		
D50	OEO	0E0	0E0	OEO	0E0		
D60 D70	OEO	0E0	0E0	OEO	0E0		
D70	OEO	0E0	0E0	0E0	0E0		
D80	0E0	0E0	0E0	0E0	OEO		
D90	0E0	0E0	0E0	OEO	0E0		
D100	OEO	0E0	0E0	0E0	0E0		
D110	0E0	0E0	0E0	0E0	OEO		
D120	OEO	0E0	0E0	OEO	0E0		
D130	OEO	0E0	0E0	OEO	0E0		
D140	OEO	0E0	0E0	OEO	0E0		t
+ C	050	050	000	050	000		t

Line type

_	DEC	HEX	ASCIT	BIN	FLOAT	
DO	4325441	420041	AB	0000000 01000010 0000000 01000001	1.51563275E0	
D2	4456515	440043	CD	00000000 01000100 00000000 01000011	1.53125799E0	
D4	4587589	460045	EF	00000000 01000110 00000000 01000101	1.54688323E0	
D6	4718663	480047	GH	00000000 01001000 00000000 01000111	1.56250846E0	
D8	4849737	4A0049	ΙJ	00000000 01001010 00000000 01001001	1.5781337E0	
D10	4980811	4C004B	KL	00000000 01001100 00000000 01001011	1.59375894E0	
D12	5111885	4E004D	MN	00000000 01001110 00000000 01001101	1.60938418E0	
D14	5242959	50004F	0 P	00000000 01010000 00000000 01001111	1.62500942E0	
D16	5374033	520051	QR	00000000 01010010 00000000 01010001	1.64063466E0	
D18	5505107	540053	ST	00000000 01010100 00000000 01010011	1.65625989E0	
D20	5636181	560055	UV	00000000 01010110 00000000 01010101	1.67188513E0	
D22	5767255	580057	WX	00000000 01011000 00000000 01010111	1.68751037E0	
D24	5898329	5A0059	ΥZ	00000000 01011010 00000000 01011001	1.70913561E0	
D26	58851351	3820017		00000011 10000010 00000000 00010111	7.64073391E-37	
D28	0	0		00000000 0000000 00000000 00000000	0E0	
D30	0	0		00000000 00000000 00000000 00000000	0E0	
D32	1179648	120000		00000000 00010010 00000000 00000000	1.140625E0	
D34	13762578	D20012	k	00000000 11010010 00000000 00010010	1.92854794E-38	
D36	73793961	46601A9	>f	00000100 01100110 00000001 10101001	2.70371324E-36	
D38	0	0		00000000 00000000 00000000 00000000	0E0	
D40	0	0		00000000 0000000 00000000 00000000	0E0	
D42	0	0		00000000 00000000 00000000 00000000	0E0	
D44	0	0		00000000 0000000 00000000 00000000	0E0	
D46	0	0		00000000 0000000 00000000 00000000	0E0	

Functions of register edit window:

1. Display and edit the contents of the read data registers and file registers from the PLC.

2. Create and edit the contents of data registers and file registers to be written to the PLC.



2

3

4

5

6

7

8

9

Α

В

С

Ε

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~	~	~	~	~	~	√	~	~	✓

6.5.7 Tool Bar Status Bar Function Keys

Use these commands to change the application window display conditions.

Click Tool Bar to erase or redisplay the tool bar.



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~									

6.5.8 Palette

Use this command to erase or redisplay the tool palette in the ladder program edit window.



Various symbols, such as contacts, lines, and output coils, used for creating ladder diagrams are located in the tool palette. Click the mouse button on a symbol in the tool palette to select the symbol.

This palette is a floating type. The palette, therefore, can be moved to another position so that the user can easily use the mouse.

These symbols (contact, line, or coil symbols) can be also entered if the Tools command or function keys in place of the tool palette is used. The advantage of the tool palette is that changes to the ladder circuit symbols without erasing the device entry dialog box is possible.

<Functions of the tool pallet>

- 1) |-||-|/[-]/[-] 2)
- 3) [4] [4] 4)
- 5) ⊣↑⊢ ⊣↓⊢ 6)
- 7)
 └┥↑└│ └┥↓└│ 8)
- 9) -()- -()- 10)
- 11) ____ | 12)
- 13) 🗡 |DEL 14)
- 1) a-contact [LD, AND]
- 2) b-contact [LDI, ANI]
- 3) Parallel a-contact [OR]
- 4) Parallel b-contact [ORI]
- 5) Load/And pulse contact [LDP, ANP]
- 6) Load/And falling pulse contact [LDF, ANF]
- 7) OR pulse contact [ORP]
- 8) OR falling pulse contact [ORF]
 - 9) OUT coil [OUT]
 - 10) Special coil [SET, RST, or applid instruction, etc.]
 - 11) Horizontal line
 - 12) Vertical line
- 13) INV instruction [INV]
- 14) Vertical line deletion

2

3

4

5

6

7

8

Ε

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1		1	1		1	~	1		

6.5.9 Contact/Coil List

This command displays the associated step number for the selected device.



6.5.10 Device Used List

Use this command to display the contact/coil conditions of 24 devices. Specify the first device number to determine the 24 devices.



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	✓		~	~		<	<	~		

6.5.11 TC Setting List

Use this command to display the timer and counter values that are set in the program.

Use these buttons to scr up and down the screen	~		<u>``</u>			//	$_{\gamma}$ Click this button to switch the device.
Enter the device	- Device	TO	Last device	Page down	Page up	Next device	* *
number.	ТО	K2	T8	К7	T16	K50	
	T1	К7	Т9	K1	T17	K60	
	T2	K1	T10	K20	T18	K70	
	T3	КЗ	T11		T19	K80	
	T4	К5	T12	K10	T20	K90	
	T5	K27	T13	K20	T21	K100	
	T6	K10	T14	K30	T22	K3	
	17	K20	T15	K40	T23	K6	

Displays the values set in the program.

The timer and counter values that are set in the program by constants will be displayed on the screen above. To check the values indirectly set by the data register, etc., use the monitoring function or the data register display function.

2

3

4

5

6

7

8

9

Α

В

С

Ε

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1	1	1	1	1					

6.5.12 Comment display

Use this function to specify the comments to be displayed in the ladder, instruction list, or SFC program edit windows.

Ladder program edit window





🙏 MITSUBISHI

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~	~	~	~	~					

List program edit window



Example: If all types of comments are specified







OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~	~	~	~	~					

SFC diagram edit window



Example: If all types of comments are specified



1
0
2
3
4
5
6
7
8
9
Α
В
С
D

Ε

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
									~	

6.5.13 Display Setting

Use this command to specify the data register display size and device comment display style for the device store/monitor screen.

Select the data display method.

	Display change	
Display mode Binary Decimal Hexadecimal Ascii Eloating point	The device comment	OK Cancel Help

Click this box to display device comments on the device store/monitor screen.

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1		~	1		1	1			1

6.5.14 Zoom

Use this command to enlarge or reduce the characters displayed in the window.

	Zoom	
Percent 0 150% 0 125% 0 100% 0 75% 0 50% 0 Custom	100% ±	OK Cancel
	100% ↑ 95% 90% 85% ↓	_

- Click Custom to specify the font size in the range of 50% to 150% in increments of 5%.
- To enlarge or reduce characters displayed in the window, change the font size by clicking Option and then Font.
- The font size specified by the Zoom command will not affect the printed character size. To change the printed character size, click Scale in the printing data setting dialog box.



6.6 PLC Menu Functions

Use the commands on the PLC menu to execute online functions, such as data transfer to/from the PLC, PLC operation check, and setting/ displaying sampling trace data.

Connection to PLC:

Correctly connect the PC to the PLC while referring to Sec.1.2.3 "System configuration".

Solving communication error:

- Check that the power source is supplied to the PLC.
 If no power is supplied, turn on the power of the PLC, and then restart the operation.
- 2) Check the model names of the cable and the RS-232C/RS-422 interface unit connecting the PC to the PLC while referring to Sec.1.2.3 "System configuration".
 If the wrong cable or interface unit is used, replace it with the right one. If the cable or interface unit is incorrectly connected please correct the correction.
- 3) Check the looseness of all the connectors.To prevent problems caused by looseness, tighten all the connectors, and then restart operation.
- 4) Check the communication port setting conditions for the communication port for the PC side. For a detailed description of the communication port, refer to Sec.6.6.10 "Ports".
- 5) If you have another peripheral unit of the FX family PLC, connect it in place of the current unit, and then communicate with the PLC.
 If the peripheral unit provides proper communication, ascertain if the PC main body, cable, or interface unit, is defective. In this case, please contact the nearest Mitsubishi electric distributor.

E

1

2

3

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		~	~		~	~			

6.6.1 Transfers

Use this command to read programs from the PLC, write programs to the PLC, or collate program data.



Writing conditions

Memory: Internal memory, RAM/ EEPROM (protect switch off) cassette PLC operation mode: Run or stop mode

Note: If an instruction uses two or more steps, there may not be any difference in the data of an error step between the PC and the PLC.

Verify conditions

LD X003

Memory: Internal memory, RAM/ EPROM/ EEPROM (protect switch on or off) cassette PLC operation mode: Run or stop mode



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		1	1		1	1			



Program transfer operation

- Use Parameter and Program (File Register, comment) to transfer all the programs at one time. Note that it takes 3 to 4 minutes to transfer a program of 1.6 K steps to the FX-EEPROM-16.
- If a keyword is stored, the Keyword dialog box will appear on the display.

Enter the keyword. If the wrong keyword is entered, then the entire operation will be stopped.

⇒ <mark>K</mark> €	eyword
Keyword	<u>O</u> K
	<u>C</u> ancel
	Help

When a program is transferred from the PC to the PLC, program verification will be automatically executed after has been written the program.
 To skip the program verification step, click Preference (see Sec. 6.8.10). The Preference window will appear. In the Automatically Verify box of this window, click the "No" button.

• If PLC type in PC is in discord with PLC, the following dialog boxes are displayed. When downloading/uploading the program data to/from PLC in the force, click "YES" button. If the download/upload would not be executed correctly, the program should be downloaded/ uploaded to/from the PLC after matching the PLC type by the PLC type change.

SW0PC-FXGP/WIN-E	SW0PC-FXGP/WIN-E
PLC type error. Continue program write?	PLC type error. Continue program read?
Yes No	

Control of parameter-set comments

- If a program has parameter-set comments, FXGP/WIN-E will read these comments as the device comments. In addition, these device comments will be regarded comments to be written to the PLC. (See Sec. 1.4 and Sec. 5.5.5.)
- The device comments regarded as the comments to be written to the PLC will be written to the program area of the PLC. (See Sec. 1.4, Sec. 6.2 13, and Sec. 6.3.9.)



1

2

3

4

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	✓		~	~		~	~			

6.6.2 Register Data Transfers

Use this command to read register data from the PLC, write register data to the PLC, or collate register data.



PLC operation mode: Stop mode (writing is impossible in run mode)

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		1	~		~	~			



Note

• If the keyword is stored, the Keyword dialog box will appear on the display. Enter the keyword. If a wrong keyword is entered, operation will be stopped.



• To use file registers, set the correct parameter to ensure enough free area. To use the RAM file register data, turn on the special auxiliary relay (M8074) on the PLC side. The file register and RAM file register will be valid only if the PLC containing these register functions is selected.

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		~	~		1	~			

6.6.3 PLC Memory Clear

Use this command to clear the memory of the PLC via the PC in the online mode. Click this button to clear the program memory of the PLC.

(Programs + file registers + parameters)

Command execution conditions

Memory: Internal memory, RAM/ EEPROM (protect switch off) cassette PLC operation mode: Stop mode (execution impossible in run mode)



Click this button to turn off all the bit devices (X, Y, M, S, T, and C). In addition, clicking this button will reset the current timer and counter values to zero.

Command execution conditions

Memory: All memories

PLC operation mode: Stop mode (execution impossible in run mode)

Click this button to clear all the data registers (D), file registers (D), and RAM file registers (D).

Command execution conditions

impossible in run mode)

Memory: Internal memory, RAM/ EEPROM (protect switch off) cassette PLC operation mode: Stop mode (execution

6.6.4 Serial Setting (D8120)

Use this command to set the communication format in the special diagnostic register (D8120) if the PLC uses a communication adapter or expansion board.

	Serial setting (D8120)	V
Protoco <u>l</u>	RS instruction 🛓	OK
<u>D</u> ata bits	7 ±	<u>Cancel</u>
<u>P</u> arity	None 🛃	Liele
Stop <u>b</u> its	1 🛓	<u>H</u> elp
Trans <u>f</u> er speed (bps)	<mark>.</mark>	Cle <u>a</u> r
H <u>e</u> ader	OFF 🛃	·
Te <u>r</u> minater	OFF 🛓	
Control l <u>i</u> ne	None 👱	
Hard <u>w</u> are	Nomal/RS-232(<mark>±</mark>	
Control <u>m</u> ode	None 👱	
S <u>u</u> m check	No 👱	
Con <u>t</u> rol procedure	Format 1 👱	

Click this button to set the data.

- If the communication data is still not set in D8120, the values shown in the left figure will appear on the Serial Setting screen.
- If communication data is already set in the D8120, the set values will appear on the Serial Setting screen.
- Click OK to directly write the set values in the D8120 of the PLC.
 In addition, the same data will be written in the data register area on the PC side. For this reason, when you use the Register Data Transfer command to transfer all the register data to the PLC, the D8120 data set on this screen will be written in the PLC again.

Command execution conditions

Memory: Internal memory, RAM/ EPROM/ EEPROM (protect switch on or off) cassette PLC operation mode: Run or stop mode



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		1	1		1	1			

6.6.5 PLC Keyword Current or Delete

Use this command to store the keyword to the program memory of the PLC. Store the keyword if you want to protect the programs from being erased by mistake (due to overwrite) or stolen.

Click this button to store the keyword.

	Keyword setting	Y
<u>O</u> ld keyword		<u>O</u> K
<u>N</u> ew keyword		<u>Cancel</u>
<u></u> o		Help

Store

Enter a keyword in the New Keyword box to store the keyword.

Change

Enter an old keyword in the Old Keyword box, and then enter a new keyword in the New Keyword box. Then, press the OK button to change the old keyword to the new keyword.

To delete a keyword, enter the keyword in the Old Keyword box, and then enter 8 characters of spaces. Then, press the OK button.

Command execution conditions

Characters used for keyword: Alphanumeric characters of A to F and 0 to 9

Number of characters: 8 characters (An error will be detected if the number of characters is less than 8.) Memory: Internal memory, RAM/ EPROM/ EEPROM (protect switch on or off) cassette PLC operation mode: Run or stop mode



When the registered keyword is unknown

In the case of an unknown keyword, the keyword cannot be changed or canceled. When deleting an entire program including the keyword, execute "PLC \rightarrow PLC memory clear".

Setting a keyword protection level for peripheral online units

If the peripheral units enable online operation with the PLC FX-10P, FX-20P, etc., set the protection level for the keyword with the appropriate device. There are three levels of protection as described below. Protect the programs from being monitored or modified by online units, via setting keyword of the optimum protection level.

All protected (No peripheral unit enables operation.):
 ADDDDDDDOR
 DDDDDDD

Enter an alphanumeric character other than A, B, or C.

- Protection from copying (Monitoring, data modify, etc.will be inhibited.):
- Protection from overwriting (Reading, monitoring, data modify, etc.will be inhibited.):

In the empty boxes \Box , enter alphanumeric characters to determine the keyword. Use alphanumeric characters from A from F and 0 to 9.



Ε

ſ	OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
		✓		√	>		>	>			



Keyword check and entry

- If a keyword is set, enter the keyword at the start of the next operation.
 The system checks the entered keyword. If a wrong the keyword is entered, a dialog box will appear on the display, and request the correct keyword.
 The keyword will be checked,
 - 1) When programs are read from the PLC, when programs are written to the PLC, or when programs in the PLC are collated, a check of the keyword is preformed.
 - 2) When programs are read from the memory cassette installed on the ROM writer, or when programs in the memory cassette are collated. The system will not check the keyword, when file data stored on floppy disks or hard disk are used, a check of the keyword not needed.
- Enter the keyword,
 - 1) When a change or delete to keyword using a PLC is preformed.
 - 2) When using a the PLC or ROM writer for data reading or writing.

If a keyword is set for the PLC or ROM cassette, the keyword entry dialog box will appear on the display for when programs are read from the PLC or ROM cassette. If the right keyword is entered, the programs will be read.

If a keyword is set for the PLC, the keyword entry dialog box will appear on the display when data is written to the PLC. If the wrong keyword is entered, no data will be written to the PLC.

To write data to the ROM writer, data in the EPROM cassette should be erased using ultraviolet ray. For this reason, the keyword entry dialog box will not appear on the display.

If the keyword stored in the PLC differs from that of the PC, the keyword in the PLC will have priority. For this reason, if the keyword entry dialog box appears an writing data (to transfer programs), enter the keyword for the PLC.

 In the offline mode, click Option and then Keyword Setting to store a keyword in the PC only, to change the keyword stored in the PC, or to delete the keyword from the PC. When using Option and Keyword Setting, the password held in the PC will be changed, but the password in the PLC will not be changed.

2

3

4

5

6

7

8

9

Α

В

С

 \square

Ε

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~									

6.6.6 Runtime Program Changes

When the PLC is in the run or stop mode, use this command to change programs in the ladder program edit window. Changed programs will be directly written to the program memory of the PLC.

Data writing method

- Using this command, modifying the existing ladder circuit in the ladder program edit window (add, insert, or delete units of ladder symbols), and inserting or adding ladder blocks (but ladder block delete is not possible) is possible. Such a modification range is limited to one ladder block. If two or more blocks are modified at a time, the program cannot be written in the program memory in the run mode. For data writing conditions in the run mode, refer to the next page.
- 2) Press the (ishift) and F4 keys (RUN CNV) or PLC and Runtime Program Change to write the changed data in the program memory in the run mode.





Caution

- If a program is changed in the run mode, the PLC will change the control sequence. Carefully change programs in the run mode for the safety reasons.
 - For a detailed description, refer to the operation manual of the PLC that describes how to write data in the run mode.

Before starting the operation, be sure to check that the same programs are stored in both the PLC and the PC.

If the operation is started with differing programs, then the programs stored in the PLC may be destroyed.

- When "Convert and run time program changes" is selected by "Option → Preferences", "Run time program changes" are executed also after "CNV F4" or "Tools → convert" is executed.
- Carefully write a pulse application instructions (with a P) in the ladder diagram. This is because the instruction written in the ladder circuit will be executed only if the input condition is ON.



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~									



Condition for program change

Hardware

Model: FX series (version 2.1 or above) FX2C series FX0N series (version 1.10 or above) FX1S, FX1N series FX2N, FX2NC series

Memory type:

FX_{2N}, FX_{2NC}; Internal RAM, optional RAM cassette, or optional EEPROM cassette (protect switch off)
FX_{1S}, FX_{1N} ; Internal EEPROM or optional EEPROM cassette (protect switch off)
FX, FX_{2C} ; Internal RAM or optional RAM cassette
FX_{0N} ; Internal EEPROM

PLC operation mode: Run or stop mode

Program

Items common to the FX, FX₂C, FX₀N, FX₁S, FX₁N, FX₂N and FX₂NC series PLC:

- When a ladder circuit is changed, only one ladder circuit block at a time is possible. In addition, before and after changes, the number of program steps included in the block should be 127 or less (including the NOP just after the ladder circuit block, but for the final block, excluding the NOP just after the block). Changes to programs, use of the Transfers command to transfer all the programs.
- 2) When modifying a ladder circuit block requires the label P or I should not be added, deleted, or changed.
- 3) If a ladder circuit block before or after a modification includes a high-speed counter C235-C255 output instruction (OUT) or application instruction FNC 90-98, the modification to the block will not be possible.
- 4) When modifying a ladder circuit block, no 1 ms timers can be inserted.

Notes regarding use of the FX (version 3.07 or above) or FX₂C series PLC:

- 1) For new instructions, inserting substitute instructions in the program, or replacing new instructions with substitute instructions is prolisted. If substitute instructions are located in the program, they will not function in the run mode.
- 2) For the above PLC, the CMP, ZCP, and MOV commands (FNC10 through FNC12) and ADD, SUB, MUL, DIV, INC, and DEC commands (FNC20 through FNC25) ensure high-speed operation only if the digits of word devices are specified or if no indexes are used for modification. However, these commands, will not ensure high-speed operation when data is written or inserted in the run mode.
- 3) For the FX (version 3.07 or more) and FX2C series, the application OFF command time is reduced. However, if data is written or inserted in the run mode, this time reduction function will not be valid, and the command OFF time will be increased by approximately 60% for the FX (version 2.30 or less) series. At the completion of data editing in the run mode, change the mode of the PLC to stop and then run again. The above functions 1) through 3) will be valid again.
| OPN | L/V | L/M | I/V | S/V | S/M | C/V | R/V | D/L | D/M | SMP |
|-----|-----|-----|-----|-----|-----|-----|-----|-----|-----|-----|
| | ~ | | | | | | | | | |



Notes regarding use of the FX series PLC (version 2.1 to version 2.30):

- 1) Application instructions or device numbers added by the FX series PLC (version 3.07 or more) will be written to the program memory of the PLC. At this moment, the system will not check these instructions, and the PLC will continue operating. However, when the mode of the PLC is switched to stop and then run again, the system will check the programs. When the PROG-ELED indicator lamp flashes, correct the corresponding instructions or device numbers, and then transfer all the acceptable programs at one time.
- 2) Even if a device number invalid for the FX series (version 3.07 or more) is specified by the operand of the application instruction, the instruction will be written to the program memory of the PLC. At this moment, an operation error (wrong device number) will be detected for this instruction. The PLC will continue operate while ignoring the error.

When the mode of the PLC is switched to stop and then run again, the system will check the programs, and the PROG-ELED indicator lamp will flash. Correct the operand of the application instruction, and then transfer all the programs at are time.

Notes regarding use of the FXON series PLC (version 1.10 or above):

- 1) If a device number invalid for the FX_{0N} is written, the message "Ladder diagram creation error" will appear on the display, and the program will not be changed.
- The FXoN cannot execute any pulse type application instructions. If such an instruction is used, "Writing error" will appear on the display, and the program will not be changed.
- 3) Even if an invalid application instruction for the FX_{0N} series is inserted, the instruction will be written to the program memory of the PLC. At this moment, the system will not check the instruction, and the PLC will continue to operate.

However, when the mode of the PLC is switched from stop and then run again, the system will check the programs. When the PROG-E LED indicator lamp flashes, correct the corresponding instruction, and then transfer all the acceptable programs at one time.

4) If an invalid device number for the FXoN series is specified by the operand of the application instruction, the instruction will be written to the program memory of the PLC. At this moment, an operation error (wrong device number) will be detected for this instruction. However, the PLC will continue operate while ignoring the error.

When the mode of the PLC is switched from stop and then run again, the system will check the programs, and the PROG-ELED indicator lamp flashes, correct the operand of the application instruction, and then transfer all the programs at one time.

For a detailed description, refer to the hardware manual of the PLC.

Writing data in run mode

• Even if the number of program steps is reduced by a change in the program, the PLC will not reduce the total number of steps, but will add NOP instructions to balance the number of steps.



• If the number of program steps is increased by change in the program, programs will be inserted to balance the number of steps.



OR X1 will be inserted in step 1.

1

3

4

5

6

7

8

9

Α

В

С

D

Е



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	✓	~	~	~	~	~	~	~	~	✓

6.6.7 Remote Run/Stop

Use this command to run or stop the FX1s, FX1N, FX2N, FX2NC series PLC using the PC.

For these operations (remote run and stop), the special auxiliary relays M8035 (forced run mode), M8036 (formed run), and M8037 (forced stop) will be controlled. Note that remote stop will be valid even if the RUN/ STOP switch of the PLCs is at the RUN position.



Select Run or Stop, and then click the OK button.

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	✓		✓	~		~	✓			

6.6.8 PLC Diagnostics

Use this function to display the scan time (operation cycle) and error detected on the PLC connected to the PC.

	PLC diagnostics		
Error item	Diagnostics result display Operation error E(6706) S(145)		lick this button to vecute PLC diagnostics.
Error code		<u>H</u> elp	0
Error step number			
Displays the data stored in the special diagnostic registers of the PLC.	Display status Scan time Current value 2 ms Maximum value 4 ms Minimum value 2 ms		
Current value : D8010 Maximum value : D8011 Minimum value : D8012	Displays the PL	C operation status	

For a detailed description of error codes, refer to the APPENDIX.



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1		~	~		~	~			1

6.6.9 Sampling Trace

Use this command to sample data at the specified time under the appropriate trigger conditions. Sampling trace will be executed by the PLC. The FXGP/WIN-E will set the sampling conditions and display the result of trace.

Applicable PLCs: FX, FX2c and FX2N, FX2Nc series





OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1		~	~		~	~			✓

Setting Parameters

Use this command to specify the devices selected for a sampling trace, and also specify the tracing time and the number of tracing times.

In addition, use a device as the trigger for tracing.



<u>, pphoable devices.</u>

Contact	: X, Y, M, S, T, C	٦	Lin to 10 devices
Coil	: T,C (Y,M*1)		Up to 10 devices
Current Value	: T, C, D, Z, V	Ξ	Up to 3 devices

*1: Devices Y and M can be stored in the Coil list. However, depending on the operation of the PLC, the coil on/off operation may not be sampled. For this reason, store devices Y and M in the Contact list.



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		~	~		~	~			✓

Run

Use this command to transfer the sampling conditions to the PLC, or read the tracing results from the PLC. In addition, use this command to start or stop the tracing operation.

Click this button to transfer the set sampling conditions to the PLC connected to the PC.

Displays the tracing status.



Click this button to read the results of sampling trace from the PLC. Use the Display Result command to display the results.

Operation while specifying a trigger:

Sampling will be performed while observing the specified tracing time until the trigger conditions are satisfied. When satisfied, sampling will be performed while observing the specified tracing time, and then completed when the number of tracing times equals the specified number (specified by Trace Count).

1

2

3

4

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		~	~		~	~			1

Display result

Use this command to display the results of the sampling trace read from the PLC.



Read from Result File

Use this command to read the results of sampling trace and the set conditions from the file.

-	Read data file	
File Name: c:\fxgpwin\sampling.stw	Directories: c:\fxgpwin c:\ fxgpwin usr	OK Cancel <u>H</u> elp
List Files of <u>Type</u> : Sampling trace(*.st w) <u>+</u>	Dri <u>v</u> es:	<u>+</u>

Write to Result File

Use this command to store the results of sampling trace and the set conditions in a file.

	Save data file
File Name: c:\fxgpwin\sampling.stw	Directories: OK c:\fxgpwin Cancel fxgpwin usr Usr Usr Cancel Help
Save File as <u>T</u> ype: Sampling trace(*.stw) <u>+</u>	Dri⊻es:



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
1	~		~	~		~	~			

Select the port from COM1 to COM9 considering the PC system

6.6.10 Ports

This "Ports" dialog box set the communication COM port (COM to COM9) and transfer speed (band rate) for connecting to PLC by RS-232C on the PC.

"19,200 bps" is available only in the FX1N/FX2N/FX2NC Series PLC. In any other PLC, the transfer speed is automatically set to "9,600 bps"

configuration.

Set the transfer speed to "9,600 bps" when using REMOTE MAINTENANCE FUNCTION.

Ports	×
Ports	OK Cancel
─Trans <u>f</u> er speed — © 9600bps © 19200bps	<u>H</u> elp
Ports	×
Ports COM1 COM1 COM2 COM2 COM3 Pd	OK Cancel

 \bigcirc

Ν	0	te

In a notebook PC equipped with infrared communication ports, the connection of the COM1 port
may be set to an infrared port, not to an RS-232C connector.
In such a case, in addition to the setting described in this paragraph, change over the COM1 port
from "infrared port" to "RS-232C connector" in the setting menu of the PC.
While the COM1 port is connected to an infrared communication port, the FXGP/WIN-E cannot
communicate with the PLC.

1	
2	
3	

D

6.7 Monitor/Test Menu Functions

On the Monitor/Test menu, there are two types of commands; monitoring commands and testing commands. The monitoring commands enable the user to monitor operation of the PLC displayed in the ladder diagram, device, or SFC diagram window. The testing commands directly control the devices in the PLC and numeric values.

Connection to PLC

To execute commands of the Monitor/Test menu, the PC should be connected to the PLC. For a detailed description of connection, refer to Sec.6.6 "PLC Menu Functions".

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1			1						

6.7.1 Start monitor

This command to monitors operation of the PLC using the ladder or SFC diagram edit window.

- ¥001 1∕1 -EALTP M506 M506 M50 M800 К2 DO M270 -ECNE M43 X02 T200 M510 -[PLS M511 M51 K100 T200 Y003 M348 M44 JPI S -FPLS M45 -FSET S0 -FSET S40
- In the edit window, click Monitor/Test and then Start Monitor to begin monitoring operation of the PLC.

- ladder program edit window

• Before monitoring operation of the circuits, check that the programs stored in the PC are the same as those of the PLC. If the programs are different, wrong monitoring results may be shown on the display.

$\text{Monitor/Test} \rightarrow \text{Start Monitor}$





Contacts connected and coils activated will be green.

Double-click the mouse button on an output instruction, such as a timer, counter, or application instruction, to modify the set value (see Sec.6.7.8).

1

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~			~						

Displaying current values on ladder circuit monitor



If the following devices are included in the OUT instruction or application instruction, the current values will be displayed on the monitor screen. (For the indirectly specified timers, counters, and devices included in the RST instruction, the current values will not be displayed.)

- . Timer (T)
- . Counter (C)
- . Data register (D)
- . Index register (V, Z)
- The values of index-modified devices in application instructions will be displayed as follows:
 - Only the current values of index-modified devices will be displayed.
 - The current values of index-modified devices will not include the index modification result.
 - The current values of index registers (V and Z) will be displayed only if one index register (V or Z) is specified in the operand.
- The current values will be displayed using decimal numbers or hexadecimal numbers ("H" will be added to the top of the value).

To switch the current value display style between decimal and hexadecimal numbers, click Monitor/Test and then Display Change of Monitor Data (_____)(see Sec. 6.7.9).

Laidder (H1 100 (H2 100) (H2 100 (H2 100 (H2 100 (H2 100 (H2 100 (H2 100 (H

Monitoring ladder circuits and operation of each device

Use Entry Device Monitor (described later) to monitor respective devices. Select a device, and you can monitor the following items at the same time:

- Operation of the contact and coils of the selected device
- Timer/counter setting value and current timer/ counter value
- Current data register value

0			Ladder			*	•
	M510 T200			-(PLS	M511	з	1
218	M510			 -CT200	K100	5	
222	T200 2026		Automatically entry device monitor				
230	M43 M506		Force ON/OFF Change the current values	-(PLS	N44	з	
239	N44 M508		Change the setting values	(SET (SET	50 540	2	
249	N43 M506	SII			K150	ļ	+

To monitor a device, click the right button of the mouse on the desired device in the ladder diagram. The shortcut menu will appear on the display as shown in the figure.

For a detailed description, refer to the description of each command.



С	OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
			~			1					

6.7.2 Stop Monitor

Use this command to stop monitoring of the PLC operation. The ladder program edit window used for monitoring will return to the edit mode.

-					Ladde	r					•	
190	N43 X032	V001	Y002							C Y010	1	Ī
196	M32 M32 M506		1002						CALTP	<mark>C 14003</mark> M506	3	
200 202	M8000 X025		M43				ECNP	К2	D0	M270	3	
210	M510 M510	T200	M43						[PLS	⊂ M510 M511	с З	
218 222	T200	X026	1003 J.Y	H348					CT200	K100	2	
230	M43	M506							(PLS	M44	з	
239	₩44 	M508							EPLS (SET	M45 S0	3	
	-1								ESET.	S40	3	→

If the ladder program edit window is in the monitoring mode. The ladder circuit cannot be edited.

 $\text{Monitor/Test} \rightarrow \text{Stop Monitor}$



0					Ladde	er					•	•
190		001 1/								—C Y010	2	Ť
	X032		¥002 ↓Ť		 					—C Y003	þ	
196	M32				 				(ALTP	M506	з	
200	M506				 					—C M507	þ	
202	M8000						(CMP	K2	D0	M270	з	
210	X025		M43							—C M510	þ	
		200							[PLS	M511	з	
218	M510								CT200	K100	þ	
222		<u>الا</u>	¥003 ↓/	M348						—C M43	þ	
		110 1/7										
230		506 							[PLS	M44	Э	
		507 							[PLS	M45	Э	
239		508 1/T							ESET.	S0	з	
	M45 M	508 							ESET	S40	З	Ŧ
+											-	•

Editing can be performed, once monitoring has stopped.

1

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
				~						

6.7.3 Dynamic Monitor

This command automatically reads out the operating state data, and monitors operation of the states displayed in the SFC program.

The screen will be automatically scrolled up, down, rightward, or leftward as the active state is switched.

- The special diagnostic auxiliary relay M8047, must be on to enable automatic monitoring using the SFC program.
- If two or more states are active in the activate multiple flows simultaneously, use the $\begin{bmatrix} Page \\ Ue \end{bmatrix}$ and $\begin{bmatrix} Page \\ Dewn \end{bmatrix}$ keys to switch the active state to be automatically monitored.

Automatic monitoring of translation branches:

To monitor the triple activate multiple flows simultaneously shown in the ladder diagram, press the $\frac{Reg}{Dam}$ key once to monitor the devices of the first branch, twice to monitor the devices of the second branch, and three times to monitor the devices of the third branch. (Up to 8 branches can be monitored.)



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1	1	1	1	1	1	1		1	

6.7.4 Entry Device Monitor

This command can be set to monitor a specified device. To use: Enter a device number to specify the device. The following can be monitored.

- Operation of the contacts and coils of the specified devices
- Timer/counter setting values and current timer/ counter values
- Current data register values

The device store/ monitor window can overlap the ladder program edit window.



Indicates that the contact is closed.

Applicable devices

X, Y, M, S, T, C, D, V, Z

Number of storage points

48 points, maximum

Saving and reading the information on registered devices

The element symbols registered in "Entry device monitor" can be saved as a registration monitor file, and can later be read to the current program file.

When saving a registration monitor file: File \rightarrow Export When reading an existing registration monitor file: File \rightarrow Import

Display style:

- The data register (D) values can be displayed using binary numbers, hexadecimal numbers, ASCII numbers, or floating-point representation. In addition, the display size can be switched between the 16-bit and 32-bit sizes. For the other types of devices, decimal numbers will be used.
- Device comments can be displayed together with the registered devices (see Sec. 6.5.13). To display
 device comments, click View and then Display Setting, or click the right button of the mouse in the device
 store/monitor window, and then click Display Setting.

1



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~	✓	~	~	~	~	~		1	

New device storage

To store a new device, position the cursor at the desired position, and then open the device storage dialog box using one of the following methods.

- 1) Directly enter the device name using the keyboard.
- 2) When the cursor is at the desired position, press the $\begin{bmatrix} Enter \\ \leftarrow \end{bmatrix}$ key.
- 3) Double-click the mouse button on the desired position.

Device storage dialog box



Stores new devices after erasing all the devices currently stored when marked.

Stored device change or delete

To change or delete a device once stored, position the cursor on the desired device, and then open the devise storage dialog box using one of the three methods described above New device storage.

Positioning the cursor on the desired device, and then pressing the best key will also delete a device.

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1	~	1	1	1	1	1		1	

Other operations using the device store/monitor screen

The data register having character "L" will be displayed using the 32-bit size.

Double-click the current value of a data register. The display size of the data register will be switched between the 16-bit and 32-bit sizes.

-			Device Monito	r/		-	•
Contact	Coil	Set	Current				t
D 1			(350)	Lower limit			
V DO			500/	Upper limit			
(D10L) C237	-	9999999999	572638543	Data			
Y000	-	9999999999	35	High speed signal Output			
C0		500	500	Execution			
TO		350	46	Wait			
		A					¥
		A				+	
т	ha fall	owing ch	artaut ma	au will appear wher	the rig	ht	h

The following shortcut menu will appear when the right button of the mouse is clicked on an item displayed on the screen.



Entering and monitoring devices using ladder diagram monitoring screen:

When the ladder program monitor window is open, the Entry Device Monitor (T,C,D device) and Entry Device Monitor (Cursor) commands will be displayed in addition to the Entry Device Monitor command described before.

The functions of these commands are described below:

Function of Entry Device Monitor (T,C,D device):

Started from the cursor position, up to 10 devices (data register : D, timer : T, counter : C) will be copied from the ladder diagram monitor window, and then pasted at the cursor position of the Entry Device Monitor screen.

• Function of Entry Device Monitor (Cursor):

The device selected inverted by the cursor will be copied from the ladder diagram monitor window, and then pasted at the cursor position of the Entry Device Monitor screen.

To execute this function, you can also use the shortcut key by clicking the right button of the mouse in the ladder program monitor window.

Ε



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1	1	1	1	1				~	

6.7.5 Forced Y Output

This command performs forced on/off operation of the output relay (Y) in the PLC.





Displays the operation history.

Execution conditions

Applicable device: Y

PLC operation mode: Run or stop mode

- This command is executed for only one operation cycle of the PLC. In the run mode, therefore, forced on/off will be valid for the latch circuit and SET/RST circuit.
- If the PLC is in the stop mode, or if the output unused in the program is specified, the forced output command will be held.
- This command can turn on the specified output relay even if no programs are stored in the PLC. This command, therefore, can be used for operation check, etc.at the completion of wiring.

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	✓	~	~	~	✓				~	

6.7.6 Forced On/Off

This command performs forced on/off of the specified bit device in the PLC. This command can also reset the current value of the specified word device to zero.

Enter a device number.



Displays the operation history.

Execution conditions

Applicable devices: X, Y, M, S, T, C (forced on/ off) T, C, D, Z, V (current value clear)

- PLC operation mode: Run or stop mode
- This command is executed for only one operation cycle of the PLC. In the run mode, therefore, forced on/off will be valid for the latch circuits and SET/RST circuit, and current value clear will be valid for the T, C, D, Z, and V.
- If the PLC is in the stop mode, or if the output unused in the program is specified, the forced on/off command will be held.
- The current value of the specified file register cannot be reset to zero if the program memory in the PLC is the EPROM cassette. If the EEPROM cassette is used, set the protect switch to off before executing this command.



1

2

3

4

5

6

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~	~	~	~	1				1	

6.7.7 Change Current Values

This command modifies the current value of timers (T), counters (C), or data registers (D) of the PLC



The current value of the specified file register cannot be modified if the program memory of the PLC is the EPROM cassette.

If the EEPROM cassette is used, set the protect switch to off before executing this command.

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~	1	~	~	1				1	

Current value setting range and applicable characters

Table:6.1

Device	Setting range and applicable characters
Т	K0 to K32,767 (decimal data)
C0-C199	K0 to K32,767 (decimal data)
High Speed Counter	K-2,147,483,648 to K2,147,483,647 (decimal data)
D(16 bits)	K-32,768 to K32,767 (decimal data) H0000 to HFFFF (hexadecimal data) B0000000000000000 to B11111111111111 (binary data) A (character string consisting of 8 ASCII characters or less)
D(32 bits)	K-2,147,483,648 to K2,147,483,647 (decimal data) H00000000-HFFFFFFF (hexadecimal data) A (character string consisting of 8 ASCII characters or less)
Z,V	K-32,768 to K32,767 (decimal data) H0000 to HFFFF (hexadecimal data)

Binary data entry : B**** Decimal data entry : K**** Hexadecimal data entry : H**** ASCII data entry : A****

The first character is an alphabetic character to show the data type.

3 4 5 6 7 8 9 Α В С D Ε

1

2

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~	✓	~							

6.7.8 Change Setting Values

This command changes the timer (T) and counter (C) setting values, constants (K, H) used as the operands of the instructions, or device numbers of the data registers (D) or index registers (V, Z).

Command execution conditions

Applicable devices	: T and C (change in constant K or device number of indirectly specified data register) K and H in application instruction (constant change)
	D, V, Z in application instruction (change in device number)
Memory	: Internal memory, RAM/ EEPROM (protect switch off) cassette
PLC operation mode	: Run or stop mode

- If the specified instruction in the PC is different from the instruction in the PLC, writing data will be inhibited. Correct the step in the program, and then execute this command.
- No value can be changed if the new value specifies a different type or device. Specify the same type and do not change the total number of steps.
- When the ladder program edit window is in the monitoring mode, position the cursor on an output instruction, and then click Monitor/Test and then Change Setting Values, or double-click on the output instruction. The following dialog box will appear on the display.

Changing timer/counter setting value





OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
		~								

6.7.9 Display change of monitor data(

This command changes the current value display style for the ladder diagram monitor window. Each time the command is executed, the current value display style will be changed between decimal number display style and hexadecimal number display style (see Sec. 6.7.1).

Decimal numbers displayed in ladder diagram (initial values)



Hexadecimal numbers displayed in ladder diagram ("H" is added to the top of each value.)



3

4

9

Ε

6.8 Option Menu Functions

On the Option menu, there are various commands that are not directly related to the active window or current work. Connect to ROM writer, if a command related to it is used.



Two similar commands

The PLC menu and the Option menu have similar commands as described below

$\mathsf{PLC} \quad \rightarrow \mathsf{PLC} \text{ Diagnostics}$

When a program is started with the PLC in the run mode, the PLC will check the program. The PC will display the detected error and operation status.

Option → Program Check The PC checks the programs stored in its own memory. If an error is detected, the PC will display the error portion.

PLC	ightarrow PLC Keyword Current or Delete
	This command can change or delete the keyword stored in the PLC.
Option	\rightarrow Keyword Setting
	This command can change the keyword stored in the PC only. The keyword stored in the PLC will not be changed even if the controller is connected to the PC.



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		1	1		1	1		1	

6.8.1 Program Check

This command checks the programs stored in the PC for syntax errors, double coils, and ladder circuit errors. If an error is detected, the step with an error will be displayed on the screen.

Check	Description	Program check
Syntax fo	Checks the instruction codes and instruction ormats. The check result will be displayed on he screen.	Syn <u>t</u> ax error check <u>OK</u> <u>D</u> ouble coil check <u>C</u> ancel Check device
Double o coil P	Checks the device symbols and device numbers in the program for double use of output instructions (OUT,SET,PLS, PLF,MC,RST) or labels (P,I). The check result will be displayed on the screen.	
Circuit di di	Checks the instructions in programs for circuit structural errors. The check result will be displayed on the screen. The error message displayed on the screen depends on the error ype as described in the following table.	Result 94 Ladder error.

Step number Description

Error message	Description	Remedy
Ladder error	 There is a circuit that cannot be converted. Examples: A wrong instruction is used as the first instruction of the circuit. The first instruction should be LD,LDI, or STL. Over-stacking Wraparound circuit 	Check the ladder diagram, and correct it.
Label error	Label instructions (P) are not used properly. Examples: CALL and P0 are used, but label P0 does not exist?	Check the labels, and correct label instructions.
Step ladder Program error	The number of STL instructions was not appropriate, or the structure of the STL circuit is not appropriate. Examples: Nine consecutive STL instructions or more are set.	Check the STL instruction circuit, and correct the ladder diagram.
Subroutine Program error	 The subroutine program was not created properly. Examples: The subroutine program is created at a wrong position. The subroutine is over-nested. An restricted instruction is detected in the subroutine. 	Check the CALL and SRET instruction circuits, and correct the ladder diagram.
Interrupt Program error	 The interrupt program was located not created properly. Examples: The interrupt program is located at a wrong position. No IRET instruction is set. An restricted instruction is detected in the interrupt program. 	Check the IRET,EI, and DI instruction circuits, and correct the ladder diagram.
FOR-NEXT Program error	The FOR-NEXT program was not created properly. Examples: • Syntax error • Over-nesting	Check the loop instruction and correct the circuit.

3 4 5

1

2

6 7

8

9

A

B

D

Ε

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1		1	1		1	1		~	

6.8.2 Parameter Settings

This command sets the parameters for the PLC. The parameter-set items and parameter setting ranges depend on the type of the PLC as described below.

Parameter settings	
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	The numeric value of 0 to 4 can be entered

Change these values to change the devices in the power failure preventive range. The setting range depends on the PLC type. Refer to the instruction manual of your PLC. If the manual describes that the power failure preventive range should be fixed to the specified range, do not change the device numbers initially set.

The programs are stored in the program memory of the PLC. The program memory can be classified into two types; internal memory incorporated in the PLC and optional memory installed whenever necessary. You can allocate the memory area for various purposes by setting parameters. Specifically, the memory area can be allocated to comments^{*1}, file registers, etc. for the PLC FX_{0N}, FX, FX_{2C} and FX_{2N} series.

In one block comment area, comments for 50 devices can be written to the PLC.

However, the allowable number of characters per comment is up to 16. If a comment consists of 17 or more characters, the 17th character and later are ignored when the comment is written to the PLC.

Setting range

Table:6	6.2						
				Memory type	•		
U	nit : step			Optional memory	cassette or board		
			4K step	8K step	8K or 16K step	16K step	
Model	Setting data type	Initial memory	1)FX-EEPROM-4 2)FX-EEPROM-4C 3)FX2NC-EEPROM-4C	1)FX-EEPROM-8 2)FX-EEPROM-8C 4)FX-RAM-8C 5)FX1N-EEPROM-8L	1)FX-EPROM-8 6)FX-RAM-8	7)FX-EEPROM-16 3)FX2NC-EEPROM-16 3)FX2NC-EEPROM-16C 7)FX2N-ROM-E1 3)FX2NC-ROM-CE1	
	Program	0.8K					
FX0	File register	—					
FX0S	Comment	—	—	—	—	—	
	Total memory capacity	Fixed parameter					
	Program	0-2K	0-2K	0-2K	0-2K		
	File register	0-1.5K	0-1.5K	0-1.5K	0-1.5K]	
FXON	Comment	0-1.5K	0-1.5K	0-1.5K	0-1.5K		
	Total memory capacity	2K	2К	2K	2К		
	Program	0-2K	0-4K	0-8K	0-8K		
FX	File register	0-1.5K	0-2K	0-2K	0-2K]	
FX2C	Comment	0-1.5K	0-3.5K	0-4K	0-4K] —	
	Total memory capacity	2K	2K/4K	2K/4K/8K	2K/4K/8K		
	Program	0-2K		0-2K			
	File register	0-1.5K		0-1.5K			
FX1S	Comment	0-1.5K	—	0-1.5K	_	—	
	Total memory capacity	2K		2К			
	Program	0-8K		0-8K			
	File register	0-7K		0-7K			
FX1N	Comment	7.5K	—	0-7.5K	_		
	Total memory capacity	2K/4K/8K		2K/4K/8K			
	Program	0-8K					
	File register	0-7K					_
FX1NC	Comment	0-7.5K		_			
	Total memory capacity	2K/4K/8K					
	Program	0-8K	0-4K	0-8K	0-16K	0-16K	
EV.	File register	0-7K	0-3.5K	0-7K	0-7K	0-7K	
FX2N	Comment	0-7.5K	0-3.5K	0-7.5K	0-15.5K	0-15.5K	
	Total memory capacity	2K/4K/8K	2K/4K	2K/4K/8K	2K/4K/8K	2K/4K/8K/16K	
	Program	0-8K	0-4K			0-16K	
	File register	0-7K	0-3.5K			0-7K	
FX2NC	Comment	0-7.5K	0-3.5K			0-15.5K	
	Total memory capacity	2K/4K/8K	2K/4K			2K/4K/8K/16K	

- 1) The FX-EEPROM-4, FX-EEPROM-8 and FX-EPROM-8 are available in the FX2C, FX0N and FX2N Series.
- 2) The FX-EEPROM-4C and FX-EEPROM-8C are available in the FX, FX_{2C} and FX_{0N} Series. The clock function is available also in the FX, FX_{0N} and FX_{2C} Series. When using the clock function in the FX_{0N} Series, however, an optional battery FX_{0N}-40B is required.
- 3) The FX2NC-EEPROM-4C, FX2NC-EEPROM-16, FX2NC-EEPROM-16C and FX2NC-ROM-CE1 are available only in the FX2NC Series.
- 4) The FX-RAM-8C is available in the FX and FX_{2C} Series. The clock function is available also in the FX and FX_{2C} Series.
- 5) The FX1N-EEPROM-8L is available in the FX1s and FX1N Series.
- 6) The FX-RAM-8 is available in the FX, FX2c and FX2N Series.
- 7) The FX-EEPROM-16 and FX2N-ROM-E1 are available only in the FX2N Series.

Default parameters setting

Table: 6.3

	ltem	Default setting								
	nem	FX0,FX0S	FX0N,FX1S	FX,FX2C	FX1N	FX2N,FX2NC				
	Memory capacity	2000 steps	2000 steps	2000 steps	8000 steps	8000 steps				
	File register ^{*1}	—	0	0	0	0				
Memory capacity	Comment area ^{*2}	—	0	0	0	0				
	Program (only view)	2000 steps*3	2000 steps	2000 steps	8000 steps	8000 steps				
	File register (only view)	0 words ^{*4}	0 words	0 words	0 words	0 words				
	Auxiliary relay [M]			500-1023		500-1023				
	State [S]			500-999	_	500-999				
Latch range ^{*5}	Counter (16bit) [C]			100-199	_	100-199				
	Counter (32bit) [C]			220-255		220-255				
	Data register [D]			200-511	_	200-511				

*1 File register is set 500 words/block. A block consumes 500 steps from memory capacity.

*2 Comment area is set 50 comment/block. A block consumes 500 steps from memory capacity.

*3 Actual program capacity is 800 steps for FX0 and FX0S series PLC.

*4 FX0 and FX0S series PLC cannot support File resister.

*5 FX, FX2C, FX2N and FX2NC series PLC can set the following range for each device.

Table:6.4

Device	Setting range
Auxiliary relay	0-1023
State	0-999
Counter (16bit)	0-199
Counter (32bit)	200-255
Data register	0-511

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1		~	~		~	~		~	

6.8.3 Keyword Setting

This command can be used to store, change, or delete the keyword for the program stored in the PC. The set keyword will not affect the keyword stored in the PLC.

	Keyword setting	
<u>O</u> ld keyword		<u>O</u> K
New keyword		<u>C</u> ancel
INCA KCYMOLU		<u>H</u> elp

For a detailed description of the keyword, refer to Sec.6.6.5 "PLC Keyword Current or delete".

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1	1	~	1	1	1	1	1	1	

6.8.4 PLC's Mode Setting

This command is used to set parameters for the battery-less mode, modem, and RUN terminal.

Click ON button to disable the memory backup battery on the FX2N, FX2NC series PLCs.

Enter data in this box to use a modem.

None, User entry mode, AIWA (PV-AF288), OMRON (ME3314B)

Batteryless mode	OON	iode setting		<u>0</u> K
<u>M</u> odem initialize	None	V	¥	<u>C</u> ancel
RUN from input	O Yes	● No		<u>H</u> elp
RUN terminal input number	χΟ		l	

Specify the a device in this box to which the external run/ stop signal is input.

X0 to X17 (X0 to X7 for FX2N-16M and FX2NC-16M Type)

Click the Yes button to enable the external run/stop signal to an FX1S, FX1N, FX2N and FX2NC series PLC.

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~	~	~	~	~	~	~		~	

6.8.5 Serial Setting (Parameter)

Serial communication settings will be stored in the parameter memory of the FX1S, FX1N, FX2N and FX2NC series PLC, and copied to the special diagnostic registers D8120 (communication format), D8121 (station number setting), and D8129 (overtime judgment value).



_: Initial condition

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1		1	1		1	1		~	

6.8.6 Printer Title

Use this command to set the title to be printed on a document.





OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1		~	~		~	~		~	

6.8.7 Device range Setting

Device ranges are specified for the PLC, however device ranges can be specified using this command.



After setting device ranges, any out-of-range devices cannot be entered even if the PLC stores the device numbers.

This function is valid when you enter or check programs.

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	~		~	~		~	~		~	

6.8.8 Comment Move

This command converts the DOS type comments stored in the program area to the FXGP/WIN-E type comments. The converted comments are stored in files (*.COW). In addition, the converted comments can be displayed on the screen and edited.

For a detailed description of the DOS type comments controlled by the FXGP/WIN-E, refer to Sec.1.4.

Flowchart for comment conversion



The comments stored in the comment area will not be changed or erased.

After being transferred to the PLC, the parameter-set comments still exist in the parameter area.

- If a device has a device comment already stored, the new comment will be overwritten by the previous comment when comment conversion is executed.
- After comment conversion, the comments stored in the program area may be unnecessary. If they are
 unnecessary, click Parameter Setting and then Comment Area. After that, set the number of blocks to "0"
 to delete the comments from the program area.
- After conversion, the converted device comments will be regarded as comments not to be written to the PLC.

To write device comments to the PLC, specify the range and refer to Sec. 6.3.9.

С

I)

Ε

1

6.8.9 PLC Type Change

Change the model recognition in a program. The interface to select a change destination model is displayed.

PLC type change Change destination – C FX <u>0</u> / FX0S C FX0 <u>N</u> C FX (FX2/FX2C)	C FX1 <u>S</u> © FX <u>1</u> N C F <u>X</u> 2N / FX2NC	Cancel	 Click this button to change the PLC type. Select a model to change the current model to the selected mode.
(The parameter setting ca	ed to the specified model. Il data in accordance with th an be changed depending o		X OK Cancel
PLC type [Memory capacity [Program [File register [Change source FX (FX2/FX2C) 2K 2000 steps 0 blocks 0 blocks	Change destination FX1N 2K 2000 steps 0 blocks 0 blocks	
	Click "C	IK" button	

When [Yes (\underline{Y})] is selected, all device ranges are available in the change destination model (newly selected model).

When [No (<u>N</u>)] is selected, inputs to only a device range specified for the previously selected model or a device range set through [Options (<u>O</u>)] \rightarrow [Assign Device (<u>D</u>)] are enabled in the change destination model. When special restriction is not required in devices available in the change destination model, select [Yes (<u>Y</u>)]. (Select [Yes (<u>Y</u>)] usually.) When applying the device range available in the previously selected model to the change destination model, select [No [<u>N</u>]].



Change from an upper-level model to a lower-level model

- Changing from an upper-level to a lower-level PLC will not automatically adjust the current program for incompatibilities with newer instructions or devices. The program can be corrected before or after the PLC type has been changed. If the program is not corrected, a program error will occur when the program is transferred to the PLC.
- Parameters can also cause problems when trying to change the PLC type. If a parameter setting is not compatible with the PLC that has been selected, the message "parameter error" will be displayed and the change will be aborted. In this case, the parameter setting should be adjusted to a setting compatible with the new PLC type.



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	✓		~			~	✓		✓	

6.8.10 Preferences

This command sets the operation environment for the FXGP/WIN-E.



- 1) Selecting the Yes option will cause the device comment entry dialog box to appear when a circuit symbol is entered in the ladder diagram edit window. Selecting the No option will not allow the device comment entry dialog box to appear automatically when editing the ladder diagram. The Edit command or the comment window should be used to enter device comments in this case.
- 2) Click the Yes button in the Automatically Verify box.
 After that, if you write programs in the PLC, the programs will be automatically collated.
 If you click the No button, programs can be written in the PLC, but no collation will be performed. (see Sec.6.6.1)
- 3) Selecting Yes will synchronize the ladder and list views so that the cursor position in the ladder diagram will indicate the step to display at the top of the list view window, and vice versa. Selecting No will disable this feature but selecting Refresh from the Window menu will synchronize the two views again.
- 4) Select the line type (Regular or Bold) for the ladder circuit display. The line type selected here is valid for display of Ladder view and printout of ladder circuit diagrams. (see Sec.6.9)
- 5) Selecting "Convert and runtime program changes" will write the program changes to the PLC during RUN status when CNV (F4) or "Tools → Convert" is executed as described in "5.6.6 Runtime Program Changes." "Convert only" should be selected when write in RUN status should not execute after a circuit has been changed or if a PLC is not connected.
- 6) If the current file has comments in the program, a confirmation message will appear and ask the user whether or not the comments should be transferred to the device comment area. The comments in the device comment area will be overwritten without confirmation if the "Hide Message" option is selected.

F

Α

В

С

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	>		>	>		>	>		>	

6.8.11 EPROM Transfer

This command reads, writes, or collates the program data using the optional EPROM type memory cassette or EPROM IC installed on a ROM writer. This transfer requires a personal computer attached to a generalpurpose ROM writer via a RS-232C cable with the EPROM type memory cassette or EPROM IC install on the ROM writer.

Connection method	: Refer to Sec.1.2.3 "System configuration".
ROM writer	: PECKER-11 manufactured by AVAL, or the equivalent ROM writer (Intel hex used as data transfer format)
• • • • •	

Applicable memory cassette : FX-EPROM-8 (Note that ROM adapter FX-ROM-SOC-1 (optional) is needed.) Data transfer : Program memory clear (parameters, programs, comments, file registers)



Read

Reads out data from the ROM writer, and stores the data in the memory of the PC.

To read data, the appropriate PLC model name should be chosen, then reading can be executed. If the keyword is stored in the ROM, keyword entry dialog box will appear on the display. The operator will need to know the keyword to proceed.

Write

Transfers the data stored in the program memory area of the PC to the memory of the ROM writer.

Verify

Collates the data stored in the memory of the PC with the data stored in the memory of the ROM writer.

- Baud rate: 9600, 4800, 2400, 1200, 900, 600, 300
- Data length: 8, 7
- Parity check: No parity check, odd parity check, even parity check
- Stop bit: 1, 2

The above items determine the communication conditions. The communication settings should be the same as those for the ROM writer.

To transfer the data from the ROM writer, the transfer format should be set to Intel hex.



Regarding ROM writer

Generally, when reading, the ROM writer reads out the data from the ROM and stores the data in it's memory. When writing to the EPROM, the data transferred from the PC is stored in the memory of the ROM writer.

For a detailed description of the ROM writer, refer to the instruction manual of the ROM writer.



OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	✓	~	~	~	~	~	~		~	✓

6.8.12 Font

This command sets the font for the FXGP/WIN-E.

-	Font	
Eont: Arial T Courier New T Symbol T Times New Roman T Times New Roman T Wingdings	Sample	Size: 11 12 14 14 14 16 18 <u>Help</u>
	,	AaBbYyZz

Changing the font size in this window adjusts the size of the symbols displayed in the ladder diagram or the characters displayed in the instruction list.

Also, the Zoom (Z) command changes the size of the displayed symbols and characters.

Initial setting (Recommendation)

Font: MS sans serif Font Style: Bold Size: 7

1
0
2
0
3
4
4
_
5
6
7
8
9
Α
B
С
D
_
E

OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
	1	~	1	1	1	1	1	1	~	~

6.9 Window Menu Functions

The commands on the Window menu, changes the current window display method.



Ledder 0

STL LD ANI OUT

X013

aduer [Program step(144/2000 EEP] - - ۲۰۰۰ [. دیس ا جالت ا

the ladder program edit window and the list program edit window using the cursor as the reference position (see Sec.6.8.10). Example: Overlapped windows

EX2N

-

-

A MITSUBISHI
OPN	L/V	L/M	I/V	S/V	S/M	C/V	R/V	D/L	D/M	SMP
✓	1	~	1	1	~	1	1	1	1	✓

6.10 Help Menu Functions

The Help screen describes how to operate the FXGP/WIN-E and how to use various functions.





1

MEMO

7. Creating Ladder and Instruction Programs

This section describes the main operations necessary for creating or editing ladder programs or instruction programs and the main items necessary for programming.

7.1 Use of Programming Window

Menu command



 To create a ladder program, open the ladder program edit window.

Click File and then New or click View and then Ladder View to open the ladder program edit window.

To create or edit ladder circuit, use the symbols allocated to the function keys, and the Tools command on the Edit menu.

Function keys Ladder program edit window

Instruction list program window

 To create a instruction program, open the instruction list program edit window.
 Click File and then New or click View and the Instruction

Click File and then New or click View and the Instruction View to open the instruction edit window.

To create or edit instruction lists, use the symbols allocated to the function keys, and the commands on the menu bar.



• The main difference between the ladder program edit window and the instruction list program edit window is that the instruction entered in the instruction list will be valid, but the instructions entered in the ladder diagram is not valid until it is converted and set in the instruction list.

The figure to the left shows both the list program edit window and the corresponding ladder program edit window filed to fit one screen.

For the reasons described above, the same program will be shown in both windows at the completion of the ladder circuit conversion, or at the completion of the creation of 1 ladder circuit block. Both windows may be synchronized by clicking the Options, then Preferences, and selecting Synchronous mode, or by clicking Window, followed by Refresh.

ANE OUT OR ANE ANE ANE ANE OUT

7.2 Creating Ladder Programs

7.2.1 Main operations necessary for entering ladder circuit symbols

There are two symbol entry methods. One is to use the symbols allocated to the function keys of the ladder program edit window, and the other is to use the Tools command on the menu. For both methods a dialog box will appear for entering data.

Entering contact symbol (method 1)

Using instruction entry dialog box:

 Entering a-contact symbol (LD, AND, OR) [- -] (F5 or tool pallet) Tools → Contact → - - To use this symbol as the OR, enter additional vertical lines. 	Input device Input device
 Directly entering OR symbol (Shift + F5 or tool pallet) 	Input device HP □Keep this dialog box X1 QK Reference Qancel Help

_	Input device
*	\Box Keep this dialog box
X0 <u>O</u> K	Reference Cancel Help
	Input device □Keep this dialog box
	хо к

Applicable devices:

X (input relay) M (auxiliary relay) Y (output relay) S (state)

T (timer) C (counter)

Data entry method 1

Entering device number

X ↑	0	Ĺ
	Device n	umber
Device	symbol	

Data entry method 2 Entering device name



 $X,\,Y,\,M,\,S,\,T\text{ or }C$



Entering contact symbol (method 2)

Using instruction entry dialog box:





Entering coil symbol

Using instruction entry dialog box:

 -()-) (F7 or tool pallet) Tools →Coil 		- 1)- (SET Y5) (<u>O</u> K	Input instruction
	imer) counter)		
Data entry method 1: Entering device num	ber:		
Y O ← ↑ ↑ ↑ Device symbol (Y, M or S) Device number			(Entering OUT Y0)
T O SPACE K ↑ ↑ Spece Spece Device symbol (T or C) Device number Spece	ecifying constant usir	ng decimal nu	(Entering OUT T0 K1)
T O SPACE D ↑ ↑ Ind Device symbol (T or C) Device number	D C	istant using d	(Entering OUT T0 D0) ata register
Data entry method 2: Entering device nam	<u>e:</u>		
Image: matrix index with the second secon			(Entering OUT RY1)
W A I T 1 * SPACE Device name stored for T or C State State State State	K 1 -	using decimal	(Entering OUT WAIT1 K1) number
W A I T 1 ' SPACE Device name stored for T or C		S (ب (Entering OUT WAIT1 10S) specified data register
Data entry method 3: Entering both device T O SPACE ' 1 O ▲ ▲ ▲ Device name stored Device number Device symbol (T or C)			(Entering OUT T0 10S)
W A I T 1 * SPACE Device name stored for T or C I	D 0 ←	constant usin	(Entering OUT WAIT1 D0) g data register



Entering special coil symbol

Using instruction entry dialog box:

		•
	Input instruction	2
SET YS	Reference Cancel	
		3
Applicable instructions: SET/RST MC/MCR END PLS/PLF STL/RET		4
Data entry method: Entering special coil by specifying device number: S E T SPACE M 0 ← SET/RST instruction ↑ ↑ ↑ ↑	(Entering SET M0)	5
Device number Device symbol SET: Y, M, S RST: Y, M, S, T, C, D		6
P L S PLS/PLF instruction M O ← O Device number	(Entering PLS M0)	7
Device symbol Y, M: M C SPACE N O C	(Entering MC N0 M0)	8
MC instruction Nesting level Device number Device symbol Y, M:		9
M C R SPACE N O MCR instruction Nesting level	(Entering MCR N0)	A
S T L SPACE S O (-) STL instruction Image: A state of the state	(Entering STL S0)	В
Device numbers	(Entoring DET)	С
R E T RET/END instruction	(Entering RET)	

Regarding the data entry method, just described device names can be entered in place of the device symbols.

For the device name entry method, refer to Entering contact symbol or Entering coil symbol described on the previous pages.

Ε

Entering application instruction

Using instruction entry dialog box:

• [-]- (F8 or tool pallet)	- Input instruction
• Tools \rightarrow Function	-CF CKeep this dialog box
	MOV K200 D0

Data entry met	hod 1: Entering	application	instruction by	specifying	instruction word:

M O V SPACE K 1 0 SPACE D 0 Instruction word Operand 1 Operand 2	16-bit MOV instruction
M O V P SPACE K 1 0 SPACE D 0 Instruction word ↑ Operand 1 Operand 2 Operand 2 Specifies the pulse execution type. Image: Compare the pulse execution type. Image: Compare the pulse execution type. Image: Compare the pulse execution type.	16-bit MOV instruction of pulse execution type
D M O ∨ SPACE K 4 0 0 0 SPACE D 0 ↑ Instruction word Instruction word Operand 1 Operand 2 Operand 2 Specifies the 32-bit type. Image: Comparison of the second secon	32-bit MOV instruction
D M O V P SPACE K 4 O O O SPACE D O ↑ Instruction word ↑ Operand 1 Operand 1 Operand 2 Specifies the 32-bit type. Specifies the pulse execution type. Specifies the pulse execution type. Operand 1: Entering application instruction by specifying FNC.number:	32-bit MOV instruction of pulse execution type
F N C 1 2 SPACE K 1 0 SPACE D 0 FNC. No. Operand 1 Operand 2	16-bit MOV instruction
F N C 1 2 P SPACE K 1 0 SPACE D 0 FNC. No.	16-bit MOV instruction of pulse execution type
F N C D 1 2 SPACE K 4 0 0 0 SPACE D 0 Image: Provide the state of th	32-bit MOV instruction
F N C D 1 2 P SPACE K 4 0 0 0 0 SPACE D A FNC. No. A A Operand 1 Operand 1 Operand 1 Operand 1 Specifies the 32-bit type. Specifies the pulse execution type.	0 32-bit MOV instruction of pulse execution type

Regarding the data entry method, just described device names can be entered in place of the operands. For the device name entry method, refer to Entering contact symbol or Entering coil symbol described on the previous pages.





Entering an instruction using the Reference command

Clicking the Reference button in the contact, coil, special coil, or application instruction entry dialog box displays a list of device names and application instructions to choose from. Section 6.3 "Tools" contains a detailed description of operation in each dialog box.



Ε

Entering INV instruction



One of the following operations can be used to enter the INV instruction in the ladder circit.

Operation	Description of motion	
(1) F7 (-/-)		
Tool pallet $(-/-)$	The INV instruction will be	
[Tools] \rightarrow [Wire] \rightarrow [$-/-(\underline{1})$]	entered.	
INV		

Entering horizontal line:



One of the following operations can be used to enter a horizontal line in the ladder circuit.

Operation	Description of motion	
F9 (-)		
-		
Tool pallet (-)	A horizontal line will be entered.	
[Tools] \rightarrow [Wire] \rightarrow [-]	-	
ANB		
(A) (B) (K) (M) (M) (M) (M) (M) (M) (M) (M) (M) (M	The specified number of horizontal lines will be entered. (The number is specified by K.)	
$(Alt) + \rightarrow / \leftarrow$	A continuous line of desired length can be drawn rightward or leftward.	



Entering vertical line



One of the following operations can be used to enter a vertical line in the ladder circuit.

Operation	Description of motion	
(1) (1) (1)		
(↑Shift) + (I)		
Tool pallet (I)	A vertical line will be entered.	
[Tools] \rightarrow [Wire] \rightarrow [I]		
ORB		
$ORB K1 \sim K49$	The specified number of vertical lines will be entered. (The number is specified by K.)	
(Alt) + (↑) / ↓	A continuous line of desired length can be drawn upward or downward.	



Entering special instruction

Entering label P or I:

Position the cursor on the left side of the bus in the ladder diagram, and then press the P or I key. The label entry dialog box will appear on the display.



Manual setting of ladder circuit turning point:

Double-click the turning point number, or position the cursor on the turning point, and then press the K key. The Turning Point dialog box will appear on the display. Using this dialog box, you can change the turning point number automatically sampled.

The turning point number can be set in the range of 0 to 255.



7.2.2 Instruction type ladder circuit entry

In the ladder program edit window, you can enter ladder circuit symbols using the methods described before. In addition, you can also enter instructions, such as AND and OUT, to edit ladder diagrams (Instruction type entry method).

Applicable instructions:

LD, LDI, LDP, LDF	PLS, PLF
and, ani, andp, andf	STL, RET, MC, MCR
OR, ORI, ORP, ORF	INV, ANB, ORB, P, I
OUT, SET, RST	Application instructions (FNC numbers or instruction words)

Data entry method:

Refer to the data entry methods described in Sec.7.3.1.

Ladder diagram drawing concept and symbol conversion:

All the instruction words entered will be converted into ladder circuit symbols, and then displayed on the screen. This method can be used together with the ladder circuit symbol entry method described before.

Instruction word / Keyboard operation	Symbol to be drawn on ladder diagram			
LD,AND	$\neg \vdash$			
LDI,ANI	- / /-			
LDP,ANDP				
LDF,ANDF	\↓			
OR/ORI	└╷└╎└╻╱┙			
ORP/ORF	└┥ャレ┘/└┤џレ┘			
INV	\neq			
ANB				
—	(1 horizontal line)			
ANB K1 to K11	(1 to 11 continuous horizontal lines)			
$\boxed{Alt} + \longrightarrow / \longleftrightarrow$	(Continuous horizontal line of desired length)			
ORB				
(↑Shift) + (I)	(1 vertical line)			
ORB K1 to K49	 (1 to 49 continuous vertical lines)			
(Alt) + ↑ / ↓	 (Continuous vertical line of desired length)			

Inst	Instruction word				Symbol drawn in ladder diagram
OUT	Dev sym		1	vice nber	(Y000) -
OUT	Dev sym			tting lues	(T3 K20) -
SET	Dev sym			vice nber	[SET M3] -
RST	Dev sym		1	vice nber	[RST M3] -
PLS	Device symbol		1	vice nber	[PLS M3] -
PLF	Device symbol				[PLF M3] -
STL S		Devi numl			
RET					[RET] -
MC N* Y/M	MC N* Y/M Devi numb				
MCR N*					[MCR N0] -
p*					P2 -
 *					1001 -
END					[END] -

For application instructions other than the ones above, simply enter the FNC numbers (example: FNC12 K1 D0) or instruction words (example: MOV K1 D0). The application instructions entered will be written in the ladder diagram.

Program conversion:

For a ladder program created with the instruction type entry method, it is necessary to convert the ladder program into a list program.

If the ladder program edit window is closed before completing conversion, the program will be erased (see Sec.7.2.3).

6

7

8

9

Α

В

С

D

Ε



<Example>



The program will be updated at the completion of conversion.

2

3

4

5

6

7

8

9

Α

В

С

 \square

F

7.2.3 Essentials for programming

Use the ladder program edit window to create ladder programs. Editing of the ladder circuit can be done in this window.

Conversion

• The ladder circuit ladder diagrams created in the ladder program edit window should be converted into instruction lists.

If the ladder program edit window is closed before completing conversion, the programs created in the ladder diagram will be erased.

If programs are created in the list program edit window, conversion will not be necessary. This is because instruction words are directly entered in the instruction lists.

• In the ladder diagram, non-converted sections will be dimmed in units of lines. At the completion of conversion, the dimmed section will be changed to normal, and the program step numbers will be determined.

				_	_
-	Ladder			•	٠
			C Y000	C	+
		-[SET	M30	Э	
		-[PLS	M32	З	
			-C Y050	С	
				L	
					÷
ŧ				+	

- Press the <u>CNV</u> (F4) key, or click Tools and then Convert to convert the ladder diagram into an instruction list.
- Conversion can be executed only if the ladder circuit block is complete. If you execute conversion without completing the ladder circuit block or with a wrong program, an error message will appear on the display, and conversion will not be performed.

If an error message appears, correct, the error, and then execute conversion again.

 After determination of the step number, if you delete the entire ladder circuit block by individually deleting the symbols or using the Line Delete command, be sure to enter a new ladder circuit in the null section. If no ladder circuit is entered, a ladder circuit error will be detected at ladder diagram conversion.

To delete the entire block after determination of step number, specify the block range using the ishirt +

 $(\uparrow)/(\downarrow)$, $(\hat{I}shift) + dragging mouse, or Block Select command, and then click the Cut or Delete command or press the Delete key.$

To cancel the ladder diagram delete or edit operation, click the control menu box at the upper left corner of the ladder diagram window. The window will be closed. After that, open the window again to display the ladder diagram.

Be careful! When you close the ladder diagram window, all the edited data will be erased if they are not converted.

- Conversion can be executed for two or more ladder circuit blocks at a time. Note that up to 200 lines can be converted at a time. When the number of non-converted lines exceeds 200, a message will appear on the display, and request conversion.
- Up to 50 lines of ladder circuit blocks can be programmed in the ladder program edit window. This means that the 51st and the subsequent lines cannot be entered. If you use the list program edit window for programming, the 51st and subsequent lines will not be displayed, either. In addition, these excessive lines cannot be printed.

If the DOS version peripheral unit, A6GPP/PHP, or GP-80 is used for printing of ladder diagrams, limit the number of lines to 24 because these peripheral units cannot print more lines.



 After conversion of the MC (master control) or MCR (master control reset) instruction, the ladder display style of the FXGP/WIN-E V1.** will differ from that of the DOS version peripherals and A6GPP. The difference in the display style is shown below:

 SW1PC- FXGPEE (DOS)
 FXGP/ WIN- E (V1.**)

 FXGP/WIN-E (V2.00 or later)
 A6GPP

 N0
 Image: MC N0 M0 Image: MC N0 Image: MC



Use of double coil

It is possible to enter two or more OUT instructions for the same coil. However, operation of the PLC will be more complicated.

At the completion of programming, click Option and then Program Check to check the program for double coils.

Automatic ladder circuit turning

- Up to 11 contacts in series can be entered in each line. The 12th contact will be automatically sent to the next line. If a coil instruction is entered at the 12th point, it will be also sent to the next line. Such ladder circuit turning can be performed up to 20 times, and up to 211 contacts can be entered. Turning point numbers 0 to 255 can be allocated to the turning points. These numbers can be manually changed (see Sec.7.2.1).
- If the STL (step ladder) instruction is in the ladder diagram, ladder circuit turning will not be possible for • the program following the STL instruction. In addition, ladder circuit turning will not be possible for the internal ladder circuit drawn in the SFC program edit window.

Unacceptable ladder diagram

The ladder circuit blocks marked with "x" cannot be converted. Correct such unacceptable blocks into those marked with "O", and then convert them.



9 Α В С

1

2

3

4

5

6

7

8

Ε

MITSUBISHI

Non-converted ladder circuit blocks in the

ladder program will be dimmed.

7.2.4 Creating ladder circuit ladder diagram

Creating new ladder diagram

Click File and then New to create a new ladder circuit ladder diagram. To clear all the ladder circuits stored, click Tools and then All Clear.

Creating ladder diagram

Use the ladder circuit symbols allocated to function keys, tool pallet, list type entry method, or Tools command to create ladder diagrams.

For the ladder circuit symbol entry method, refer to Sec.7.2.1 "Main operations necessary for entering ladder circuit symbols".





 $\begin{array}{c} \text{or} \\ \text{Tools} \rightarrow \text{Convert} \end{array}$



• When conversion is executed, the ladder program displayed on the screen will be converted into an instruction list.

	eð	苏雕		DB	88	9							_
-						Lad	der					•	•
0		,	3002 J/F								—C Y013	7	t
	Y013										K200	2	
			l						 	(05	D20		
10	¥013	_							 [MOV	K1800	D1	3	
										[SET	M30	1	
				 		 				[PLS	M35	3	
13	X002								 DMC	NO	M50	1	
2	×005								 				
25										(MCR	NO		
23	X006									CJ	PO		
31	MSOCO								41 44 +1 44		(Y020]	Ц
									11				Н
									<u>00</u> - 1 _		-C Y022	1	-
T									≠ ka		CTOLL	1+	P

- At the completion of conversion, the program steps will be determined, omitting any empty spaces and adjusting circuit layout. In addition, dimmed blocks will be changed to normal.
- Even after conversion, automatic writing to the memory of the PLC can be executed in the run mode (see Sec. 6.6.6 and Sec. 6.8.10).



Editing



Pressing the Enter key on a contact or output coil, or double-clicking the mouse button on a contact or output coil will cause the instruction change dialog box to appear on the display. If changes need to be made. Enter new data.

Various functions of right mouse button

Click the right button of the mouse on the ladder diagram to perform following operations:

Operation	Operation Description		
Undo	Cancels the last operation.		
Cut	Removes the graphic from the cursor position or specified range.		
Сору	Copies the graphic from the cursor position or specified range.		
Paste	Pastes the Clipboard data (graphic data temporarily stored) on the cursor position.		
Edit cancel	Deletes the ladder diagram just edited.		

1

2

3

Editing (after conversion)

Keys and	l commands	used	for	editing	ladder	diagram
----------	------------	------	-----	---------	--------	---------

Item	Editing	Operation
	Specifies the edit range in unit of	Drag the mouse to specofy the range in units of symbols.
	symbols in the circuit block.	$\boxed{Ctrl} + \longrightarrow / \longleftrightarrow / \Uparrow / \downarrow$
Range specity		$ (\uparrow Shift) + (\uparrow) / (\downarrow) $
	Specifiles the edit range in units of cuicuit blocks.	(↑Shift) + moouse dragging
		Edit \rightarrow Block Select
		Delete
	Deletes the cursor-positioned symbol.	Edit \rightarrow Delete
		$Edit \to Cut$
Symbol delete	Deletes to the left of the cursor.	Back Space
		([↑] Shift) + F8
	Deletes a vertical line.	Tools \rightarrow Wire \rightarrow IDEL
		Tool pallet (IDEL)
	Deletes a line or symbol.	$\boxed{Ctrl} + \boxed{Alt} + \bigcirc / \longleftrightarrow / (\uparrow) / (\downarrow)$
Сору	Copies symbols in the specified range.	Edit → Copy
	Copies symbols in the specified range.	Ctrl) + C
Paste	Pastes data sent from the clip board.	Edit \rightarrow Paste
		Ctrl) + V
	Inserts one line at the cursor position.	Edit \rightarrow Line Insert
Line edit		(1) Shift + (Insert)
	Deletes the cursor-positioned line	Edit \rightarrow Line Delete
	(right side of the cursor).	(↑Shift) + (Delete)
Block delete	Deletes the step-number-determined circuit blocks.	Specify the deit range in units of blocks. Edit \rightarrow Cut or Delete
Edit cancel	Deletes the ladder diagram you edited.	Edit \rightarrow Edit Cancel
Undo	Cancels the last operation, and returns to the previous condition. (Ladder	$Edit \ \rightarrow \ Undo$
	circuit block conversion operation, however, will not be cancelled.)	Ctrl) + Z
		Scroll bar
	Scrolls the screen (or moves the cursor).	$(\uparrow) / \downarrow$
Scroll		Page Up / Page Down
	Jumps to the specified position.	Serch \rightarrow Top, end, device, instruction/coil, step number, tag
Device Change	Changes devices.	Serch \rightarrow Replace, a-contact/b-contact change, convert
Comment	Enter or changes comments.	Edit \rightarrow Device name, device comment, coil comment, block comment

2

3

4

5

6

7

8

9

Α

B

С

D

Ε



Note regarding ladder circuit block delete

	Г	-	Ladder			•	
Step number already -	->			[SET		ÅÅ.	Ladder circuit block Ladder circuit block
determined		↓ M50 		-CT0	K20	+ K	— Ladder circuit block

If step numbers are already determined for the ladder circuit blocks, you can delete the blocks by following the procedure below:

- 1) If you want to delete ladder circuit blocks, specify the delete range using the (Ishift) + (↑/↓), (Ishift) + mouse dragging, or Block Select command.
- 2) Click Cut or Delete, or press the Delete key to delete the specified blocks. When deleted, the corresponding instructions will be also deleted from the instruction list. It is, therefore, not necessary to convert the changed ladder diagram into an instruction list.

After deleting a step-number-determined entire ladder circuit block in units of symbols or using the Line Delete command, be sure to enter a new ladder program in the blank line. If a line remains blank, a conversion error will be detected. To delete an entire ladder circuit block, be sure to follow the above procedure.

To cancel the ladder diagram delete or edit operation, click the control menu box at the upper left corner of the ladder program edit window. The window will be closed. After that, open the window again to display the ladder diagram.

Be careful.

When you close the ladder program edit window, all the edited data will be erased if they are not converted.



7.3 Creating Instruction List Program

7.3.1 Main operations necessary for entering instruction words

Entering contact instruction

Instruction word entry method and ladder program displayed:

[LD] (Load, Initial logical operation contact		normally anons)	
	type NO <	normally open>)	
L D SPACE X O C LD instruction Device number Device symbol	or	F5 (LD) X 0 ← Function key ↑ ↑ ↑ Device number Device symbol	
[AND] (AND, Serial connection of NO <nor< td=""><td>mally oper</td><td>n contact>)</td><td></td></nor<>	mally oper	n contact>)	
A N D SPACE X 1 -	or	F6 (AND) X 1 -	
AND instruction		Function key	
Ladder program displayed:	X000	X001 	,

[LDI] (Load inverse, Initial logical operation	on contact	type NC <normally closed="">)</normally>
LDI SPACE XOC LDI instruction	or	SHIFT + F5 (LDI) X O ← Function key ↑ ↑ ↑ Device number Device symbol
[ANI] (AND inverse, Serial connection of	NC <norr< td=""><td>nally closed contact>)</td></norr<>	nally closed contact>)
A N I SPACE X 1 I ANI instruction Image: Additional system of the system of	or	SHIFT + F6 (ANI) X 1 I Function key
Ladder program displayed:	X000	X001 ────┴┴──── ANI

Devices applicable to LD, LDI, AND, ANI instructions:

X (input relay)	M (auxiliary relay)	T (timer)
Y (output relay)	S (state)	C (counter)



Instruction word entry method and ladder diagram displayed:

[LDP] (Load pulse, Initial logical operation - Rising edge pulse)	1
L D P SPACE X O C LDP instruction	2
Device number Device symbol [ANDP] (AND pulse, Serial connection of Rising edge pulse)	3
A D P SPACE X 1 ANDP instruction	4
Ladder program displayed: X000 X001 Intermediate Intermediate Intermediate LDP ANDP	5
	6
[LDF] (Load falling, Initial logical operation Falling/trailing edge pulse)	7
LDF instruction	8
[ANDF] (AND falling, Serial connection of Falling/trailing edge pulse)	9
ANDF instruction	Α
Ladder program displayed:	В
	С

Device applicable to LDP, LDF, ANDP and ANDF instructions:

X (input relay)M (auxiliary relay)T (timer)Y (output relay)S (state)C (counter)

D E



Instruction word entr	y method	and ladder	program	displayed:

[OR] (OR, Parallel connection of NO <normally open=""> contacts)</normally>			
O R SPACE X OR instruction Image: Construction Image: Construction	or Fu	F7 (OR) X 0 ← Inction key ↑ ↑ Device number Device symbol Device symbol Device symbol	
[ORI] (OR inverse, Parallel connection o	f NC <normally clo<="" td=""><td>osed> contacts)</td></normally>	osed> contacts)	
ORI INSTRUCTION	or	SHIFT + F7 (ORI) X 1 I Function key	
Ladder program displayed:	M0 X000 OR X001 ORI		
		<u>`</u>	
[ORP] (OR pulse, Parallel connection of Rising edge pulse) ORP instruction ORP instruction ORP instruction Device number Device symbol			
[ORF] (OR falling, Parallel connection of	Falling/trailing ed	ge pulse)	
ORF instruction			
Ladder program displayed:	МО		

Device applicable to OR, ORI, ORP and ORF instructions:

X (input relay)	M (auxiliary relay)	T (timer)
Y (output relay)	S (state)	C (counter)



6

7

8

9

Α

В

С

D

Ε

Instruction word entry method and ladder program displayed

[STL] (Step ladder, step ladder operation contact type NO <normally open="">)</normally>	
S T L SPACE S O ← STL instruction ↑ ↑ Device number	2
Device humber Device symbol	3
Ladder program display:	4
	5

Device applicable to STL instructions: S (State)

Instruction word entry method: Specifying device name

To specify the device, enter a device name in place of a device number.

Example:

LD SPACE '	
Instruction word	Device name stored for X, Y, M, S, T and C
F5 (LD) · L	<u>S</u> 1'-
for entering instruction word	

Entering coil instruction

Instruction word entry method and ladder program displayed:

)
[OUT] (Output, Final logical operation type coil dri	ve)	
OUT instruction SPACE YOU CONT or OUT instruction OUT Device number Device symbol Y, M or S	F7 (OUT) Function key Device number Device symbol Y, M or S	
Ladder program display:	(Y000) - OUT	

[OUT] (Output, Final logical operation type coil drive)			
O U T SPACE K 1 0 ← OUT instruction ↑ Specifying decimal type constant Device number Device symbol T, C F7 (OUT) T 0 SPACE K 1 0 ← F7 (OUT) T 0 SPACE K 1 0 0 ← Function key ↑ Specifying decimal type constant Device number Device number Device symbol T, C	When constant is specified How to enter instruction word OUT T0 K100		
O U T SPACE C 1 SPACE D 0 ← OUT instruction	When indirectly specified How to enter instruction word OUT C1 D50		
Ladder program display: ————————————————————————————————————			

Device applicable to OUT instructions:

Y (output relay) S (state) M (auxiliary relay) T (timer) C (counter)



Instruction word entry method: Specifying device name			
To specify the device, enter a device name in place of a device number.			
Example:			
O U T SPACE ' R Y 1 ' I OUT instruction Device name stored for Y, M and S			
F7 (OUT) ` R Y 1 ` ← Function key Device name stored for Y, M and S			
O U T SPACE C 1 ··· 1 0 S ·· ··· OUT instruction			
O U T SPACE W A I T SPACE I I OUT instruction OUT instruction Device name stored for T and C Device name stored for t and c Device name stored for the construction Device name stored for the construction Device name stored for the construction			

Entering single instruction

Instruction word entry method and ladder program displayed

[ANB] (AND block, serial connection of multiple parallel ladder circuits)			
A N B or F8 (ANB) ANB instruction Function key			
Ladder program displayed:			
[ORB] (OR block, parallel connection between blocks)			
O R B			
Ladder program displayed:			
[MPS] (Point Store, Stores the current result of the internal PLC operations) [MRD] (Read, Reads the current result of the internal PLC operations) [MPP] (Pop, Pops [recalls and removes] the currently stored result)			

MPS, MRD or MPP instruction Ladder program displayed:	[MPP] (Pop, Pops [recalls and removes] the currently stored result)
MPP MRD	
	MPP MRD



[NOP] (No operation, No operation or null step)	1
[RET] (Return, Return to step ladder) [End] (End, Forces the current program scan to end)	2
NOP, RET or END instruction Ladder program displayed:	3
S0 - STL	4
(Y001) [RET] END	5
[END] [NOP instructions will not be displayed in the ladder diagram.	6
[P] (Pointer, for Program flow control) Ladder program displayed:	7
P 0	8
[I] (Interrupt Pointer, Interrupt program marker) ₁₃₀₁ X001 I 3 0 1 ←	9
Interrupt pointer number	Α
tering special coil or application instruction:	В

For the special coil and application instruction entry methods in the list program edit window, refer to those used in the ladder program edit window.

Special coil:

[SET] (Set, Sets a bit device permanently ON) [RST] (Reset, Resets a bit device permanently OFF) [PLS] (Pulse, Rising edge pulse) [PLF] (Pulse, Falling/trailing edge pulse) [MC] (Master control, Denotes the start of a master control block) [MCR] (Master control reset, Denotes the end of a master control block)

FNC00- FNC246 Application instruction:

For the above instruction entry method, refer to Sec.7.2.1 "Main operations necessary for entering ladder circuit symbols". Note that in the ladder program edit window, the data entry dialog box will appear so that you can directly enter the instruction word at the cursor position, but in the instruction list program edit window, this dialog box will not appear.

С

D

Ε



Instruction entry using Instruction command

In the instruction list program edit window, you can directly enter instruction words or device numbers in the cursor-positioned line using the keyboard. However, if you want to use lists or dialog boxes for easy programming, use the Instruction command.



Specify the operand or device names.

2

3

F

7.3.2 Creating instruction list

Creating new list

Click File and then New to create a new instruction list. To clear all the ladder circuits stored, click Tools and then All Clear.

Creating/editing instruction list

In the instruction list program edit window, position the cursor and then enter an instruction word directly at the cursor position, or use the ladder circuit symbols allocated to function keys or the Instruction command on the Tools menu to create an instruction list.

For the instruction word entry method, refer to Sec.7.3.1.

For the instructions regarding programming, refer to Sec.7.3.2.



Keys and commands used for editing

Symbol delete:

Delete: Deletes the symbol at the cursor position.

NOP operation:

Edit : All write, all insert, all delete

Scroll:

- Page Up, Page Down, \uparrow , \downarrow , scroll bar
- Search : Top, end, device name, device, instruction, contact/ coil, step number, tag set, tag jump Device change:

Search : Replace, a-contact/ b-contact change, convert

Comment entry:

To enter comments, use the All Entry screen by clicking View and then Comment. Or enter comments in the ladder program edit window.



MEMO

2

3

4

5

6

7

8

9

Α

В

С

Ε

8. Creating SFC Programs

SFC programming is simply entering symbols for devices and instructions in the SFC (Sequential Function Chart) to determine the flow of control.

For a detailed description of SFC control, refer to the instruction manual of the PLC.

8.1 Description of SFC

8.1.1 Types of SFC

The SFC has a single flow.

The single flow type program can control sequences of simple motions. If various input conditions are set or the operator enters various commands, use the selective branch or simultaneous branch type.







8.1.2 **Program types**

When creating programs using the SFC, the SFC sections and ladder diagram sections (inner ladders) should be separately programmed.

The SFC type section consists of ladder blocks and state blocks.

To create programs used for control, enter symbols, such as SET SO (ladder block), Y000 (state), and TRAN (transition). These programs will be controlled as inner ladders.

SFC section:

To use a SFC for programming, enter SFC symbols to determine the flow.

Ladder section:

Create inner ladders in the ladder diagram to specify the control of each process, conditions to transfer to the next state, etc.





When you create a ladder block after the state block, the FXGP/WIN-E will automatically enter the RET instruction.

For this reason, do not enter any RET in the program.

8.1.3 Structure of SFC section The rows, columns, and cursor position in the SFC section are described below: 1 16 Column 0 Column 1 Column 2 columns, 2 maximum Ladder 0 Row Ladder number Row number: block 0 3 Respective states including ladder blocks have their own row numbers. S 0 Row Column number: number Respective states have their own 4 1 column numbers. Up to 16 columns can be used for programming. S 20 Row 5 number 2 S 21 S 22 S 23 Row State 6 number block 3 S 24 Row 7 number 4 Jump 8 S 0 Row number 5 9 Ladder 1 Row Ladder number block 6 Α S 1 Row number State 17 В block S 30 250 rows, maximum С

D

Ε

8.2 Creating SFC Programs

8.2.1 Main operations necessary for entering SFC symbols

Use the function keys to enter these symbols in the SFC program.

Item		Symbol displayed on screen	Function key	Remarks
Ladder block		Ladder m	F8	m=ladder block number, automatically attached
Initial state		Sn	ÎîShift + F4	Sn=S0 to S9. S0 to S9 will be controlled as the initial states, and the initial state depends on the state number.
General state		Sn	Îshift + F4	Sn=S10 to S899.
Jump (loop)		↓ Jump Sn	F6	Sn=S0 to S899.
Jump (reset)		⊥ Reset ✓ Sn	F7	Sn=S0 to S899.
Transition (transition condition)		+	Îshift + F5	The transition condition will be written.
Vertical line			Îshift + F9	Connects a state to another state.
Horizontal line		(Selection branch,junction) (Selection branch,junction)	F9	Automatically recognized as the selective or simultaneous branch line, and the recognition result depends on the symbol-written position.
Combined symbol	State + transition	Sn +	F5	Sn=S10 to S899.
	Branch junction		Îshit + F6	Automatically recognized as the selective or simultaneous branch line, and the recognition result depends on the symbol-written position.
			(îshift) + F7	
			Îshift + F8	
8.2.2 Cursor position and entering symbol

The figure below show the symbols that can be entered in relation to the cursor position on the SFC program edit window.

The selective branches or simultaneous branches will be automatically recognized by the connection position.



- 1) State entry position (creation of inner ladder), entry of jump symbol or creation of ladder block
- 2) Connection position where state and next step are connected to each other
- 3) Junction of selective branches and simultaneous branches
- 4) Transition entry position (creation of transition condition, creation of inner ladder)
- 5) Junction of simultaneous branches or selective branches



8.2.3 Main items necessary for creating SFC

Use of programming window

Open the SFC program edit window to use the SFC for programming. Click View and then SFC to open the SFC program edit window.

Conversion

• After creating a SFC program and inner ladder program in the SFC program edit window, be sure to convert them into an instruction list.

Non-converted programs will be erased when the SFC program edit window is closed.

- Press the CNV (F4) key, or click Tools and then Convert to convert programs.
- Be sure to convert both the inner ladder program and SFC.

Editing inner ladder diagram

- To display, create, or edit a ladder block or inner ladder of a state or transition, position the cursor and then click View and then Inner Ladder View.
- To edit a ladder block or inner ladder of a state.
- When creating a ladder block after the state block, FXGP/WIN-E will automatically enter the RET instruction. For this reason, a RET instruction should never be entered.

Sequence instructions usable for state:

Instruction		LD/LDI/LDP/LDF, AND/ANI/ANDP/ANDF, OR/ORI/ORP/ORF,INV, OUT,SET/RST,PLS/PLF	ANB/ORB MPS/MRD/MPP	MC/MCR	
Initial state, general state		Usable	Usable	Unusable	
Branch,	Output operation	Usable	Usable	Unusable	
junction, state	Transition operation	Usable	Unusable	Unusable	

- The STL instruction cannot be used in interrupt programs or subroutine programs.
- Use of jump instructions in the STL instruction is not inhibited. It is, however, recommended that jump instructions should not be used in the STL instruction. This is because operation will be too complicated.
- When the MC (master control) or MCR (master control reset) instruction is entered, the ladder will be displayed in the same way as the FXGP/WIN-E V1.** (see Sec. 7.2.3).
- When two or more transition conditions exist, these conditions should be programmed in one transition ladder circuit.



2

3

4

5

6

7

8

9

Α

В

С

Ε

- If no inner ladder is created for a state or transition, the state or transition will be marked with "*". When an inner ladder is created, the mark will disappear.
- The SFC can be created without the inner ladders firstly to determine the flow. Afterwards the inner ladders can be added.
 If the SFC without inner ladder program is converted, the instruction list of flow only will be created. This SFC can be displayed again, and then inner ladders can be added to the SFC to complete it.
- Dummy state entry method Follow the procedure below to enter a dummy state in the SFC:
- Enter a state symbol (symbol for dummy state). No program will be processed in this state, therefore it is not necessary to create an inner ladder. On the screen, there is an asterisk "*" to show that there is no inner ladder.
- 2) Create a transition to go to the next state from the dummy state.



3) The dummy state is marked with "*", but convert the program ignoring the asterisk.

Remarks

To jump to the same state in the SFC, do not write the same jump state twice or more. If you write the same jump state in two or more different places, the ladder circuit you created may not function properly.



S10 and S20 are included in the inner ladders of transitions X010 and X020.

Correct the ladder diagram as follows:



In the following ladder circuit, the same problem will occur. Do not write the same jump state twice or more.



Acceptable ladder circuit



2

3

4

5

6

7

8

9

Α

В

С

 \square

Ε

8.2.4 Creating SFC

Creating new SFC

Click File and then New. Select SFC program edit window from the list of programming windows. Using the View menu to open the SFC program edit window is another option.

Creating SFC

Create a SFC using the ladder circuit symbols allocated to the function keys. For the SFC symbol entry method, refer to Sec.8.2.1.

Editing

							PAVIN-E - SFO				v •
File		ols	Search	View	PLC	Remote	Moniter/Test	: <u>O</u> ption	<u>W</u> indow	<u>H</u> elp	
	39 8(<u>8</u> 4									
-	-						SFC				▼ ▲
Ш	\$13										+
Ш	+										
Ш	Reset S13										
	513										
Ш	1 S2										
Ш	T										
Ш	\$21										
			-								
	522		T								
Ш			<u> </u>								
	-	_	526								
Ш	521		526								
Ш	<u>+</u>	_	±								
Ш	\$24		527								
	_+		+ =		-						+
Ŀ	-										
SF		- 1	.CHV . 모		Cur. (I	12,W2)		Write		FX2N	
		1									
			~			l					
		Ľ	C	urs	or (coor	dinate	s			

Press the ment key to change the data entry mode from Overwrite to Insert mode, or vice versa.

Function key (You can also click the mouse button on the desired function key to use the function.)

Keys and commands used for editing

Symbol delete:

- Delete key (cursor position)
- Edit → Delete (cursor position)

Block insert:

Insert an initial state to add an SFC block.

Scroll:

- $\left[\begin{array}{c} Page \\ U_p \end{array}\right], \left[\begin{array}{c} Page \\ D_p \end{array}\right], \left[\begin{array}{c} \uparrow \end{array}\right], or \left[\begin{array}{c} \downarrow \end{array}\right]$ keys, scroll bar (screen scroll)
- Search (top, end, device name, state number)

Inner ladder:

• View \rightarrow Inner Ladder (To edit inner ladders, refer to Sec.7.2 or Sec.7.2.3.)



MEMO

9. Remote Maintenance Using a Telephone Line

The FXGP/WIN-E has the remote access function that enables program transfer and monitoring through a telephone line. For this remote access function, the PC should be connected to the FX1S, FX1N, FX2N and FX2NC PLC via modems.

In addition, the FXGP/WIN-E has the file transfer functions that enables file sending and receiving between PC through the telephone line. As with the remote maintenance function, the PC should be connected to each other via modems. This function allows the user to send and receive programs, comment files, FXGP/WIN-E data files, and other general files to and from another computer.

9.1 System Configuration and Functions

9.1.1 Remote access to PLC

System configuration



- *1: Be sure to use the cable specified by the modem manufacturer to connect the PC to the modem, or connect the PLC to the modem. The transfer speed will be fixed to 9,600 bps.
- *2: The remote maintenance function cannot use on the WindowsNT 4.0, Windows 2000 or Windows XP.

Functions

The PC can be connected to the modem of the PLC through a telephone line.

FXGP/WIN-E can automatically connect to the desired PLC by selecting the corresponding station from those stored in the PC. In addition, FXGP/WIN-E can manually connect the PC to the PLC by manually setting the telephone number of the PLC.

When connected, various commands on the PLC and Monitor/ Test menus will be valid.

These commands are listed below

Commands on PLC menu

- Transfers
- Register Data Transfers
- PLC Memory Clear
- Serial Setting (D8120)
- PLC Keyword Current or Delete
- Runtime Program Changes
- Remote Run/Stop
- Diagnostics
- Sampling Trace

Commands on Monitor/Test menu

- Start monitor
- Stop Monitor
- Dynamic Monitor
- Entry Device Monitor
- Forced Y Output
- Forced On/Off
- Change Current Values
- Change Setting Values

9.1.2 Sending/receiving files

System configuration



- *1: Be sure to use the cable specified by the modem manufacturer to connect the PC to the modem.
- *2: The remote maintenance function cannot use on the WindowsNT 4.0, Windows 2000 or Windows XP.
- *3: The transfer speed will be fixed to 19,200 bps. If the RS-232C port of the PC cannot keep up with such high speed, this file sending/ receiving function cannot be used. (For the transfer speed of the RS-232C port, refer to the operation manual of the PC.)

Functions

One of the PCs should act as the receiver. The other PC should be used to send one or more files specified on the screen to the desired directory of the receiver PC.

If the directory of the receiver PC is not specified, the received files will be stored in sub-directory FILE of the FXGP/WIN-E.

If a new directory is specified, the new directory will be created, and then the received files will be stored in that directory.

Transfer files from the receiver PC to the sender PC to change the transfer direction between the personal PCs.

FXGP/WIN-E files and general files can be sent and received to use this file transfer function.

1

9.1.3 Caution

- 1) If using the remote access system of the PLC frequently, it is recommended that the programs be password-protected. The password will prevent unauthorized users from illegal access to sequence programs. (See Sec.6.6.5 and Sec.6.8.3.)
- 2) During communication via the telephone line, if a call is placed to the number being currently used for communication, a communication error may occur, or the line will be disconnected.
- 3) If the telephone line is connected to the PC via other business phones or private branch exchange, the modem may not function properly due to wrong connector or signal type. For more information, contact the person in charge of the private telephone line.
- 4) Depending on the telephone line conditions or the field intensity of the portable telephone, communication may not be possible, or the line may be disconnected during communication. The file transfer function described in Sec.9.1.2 may not be valid if the wrong modem is connected to the PC, or the wrong PC is used.



2

9.2 Operation Procedures

9.2.1 Remote access to PLC

Connect PC with the FXGP/WIN-E to the PLC using the telephone line in order to enable remote access to the PLC.

Access procedure



- *1: Be sure to turn on the power of the modem first, and then that of the PLC.
 - This is because the PLC will send the modem initialize command to the modem when the power of the PLC is turned on. If you turn on the power of the PLC without turning on the power of the modem, the modem initialize command will not be sent to the modem, therefore communication error will occur when you access to the remote PLC. After turning on the power of both units, if you turn off the power of the modem only, the initialize command will be canceled, therefore communication error will occur. In this case, turn off the power of both units, and then turn on the power of the modem first, and then that of the PLC.
- *2: Be sure to use the To PLC command to connect the PC to the PLC. If you use the To File Transfer command, communication with the PLC will not be possible.
- *3: If you access to the PLC immediately after line connection, a communication error may occur. This is because the modem is still not ready for communication. Wait several seconds, and then access the PLC.

2

9.2.2 File transfer

Connect the FXGP/WIN-E installed PC to another FXGP/WIN-E installed PC using the telephone line in order to enable file sending and receiving between these PCs.

Transfer procedure



*1: Be sure to click To File Transfer to connect to the receiver PC. If you click To PLC, the line will not be connected to the PC.





🙏 MITSUBISHI



9.3 Functions on Remote Menu

9.3.1 Connect



"To PLC" Connects the PC to the PLC through the telephone line to enable remote access to the PLC. Storing and editing data is possible on the remote station (PLC).

"To File Transfer" Connects the PC to the receiver PC to enable file transfer between PCs. Storing and editing data is possible on the remote station (receiver PC).

Line connection screen

Click To PLC and then To File Transfer. The following line connection screen will appear on the display. This screen is common to both commands TO PLC and TO File Transfer, and the data edited on this screen will be controlled in the same way by both commands.

The transfer speed between the PC and the modem depends on the specified command; To PLC or To File Transfer.For this reason, select the right command, and then click the Connect button.

If a wrong connection command is selected for line connection, a communication error will occur during remote access to the PLC or during file transfer between PCs.



Select a remote station.

🖈 MITSUBISHI

Line connection procedure (connection to remote station stored in the list)

Follow the procedure below to connect the telephone line to a remote station stored in the list.

- 1) Select Remote and then Connect. After that, select To PLC or To File Transfer depending on your purpose.
- 2) Click a remote station in the list to specify the station.
- 3) Click Yes.
- 4) The confirmation message "Connect to remote station?" will appear on the display. Select Yes.
- 5) The modem calls the remote station to connect the PC to the remote station.

Line connection procedure (connection to non-stored remote station)

Follow the procedure below to connect the telephone line to the desired remote station by directly entering the phone number in the Manual Connection dialog box.

The data entered in the Manual Connection dialog box will not be stored in the list, but will be logged in the communication log file.

- 1) Select Remote and then Connect. After that, select To PLC or To File Transfer depending on your purpose.
- 2) Click Manual.
- 3) The Manual Connection dialog box will appear on the display. Enter data in this dialog box.

Manual	
**** ELECTRIC	
111-222-3333	
	<u>C</u> ancel
	<u>H</u> elp
	***** ELECTRIC 111-222-3333

Enter the name in this box. The name will be logged in the communication log file as the remote station.

Enter the phone number in this box. The modem will call the phone number.

Click this button to connect the telephone line to the specified remote station.

Set data in these boxes if necessary.

For a detailed description of data items in this dialog box, refer to "Storing new remote station" described later.

- 4) Click the OK button.
- 5) The confirmation message "Connect to remote station?" will appear on the display. Select Yes.
- 6) The modem will call the specified remote station to connect the PC to the remote station.

1

2

3

Storing new remote station

Follow the procedure below to connect the telephone line to the remote station non-stored in the list.

- 1) Select Remote and then Connect. After that, select To PLC or To File Transfer depending on your purpose.
- 2) Click New Entry.
- 3) The Edit Connection Data dialog box will appear on the display. Enter data in this dialog box.

-	Edit Connection Data	
<u>N</u> ame	MITSUBISHI ERECTRIC 1	
<u>P</u> hone	123-1234-1234	
<u>O</u> utside Line		

Enter the name of a remote station in this box. You can use up to 64 characters to set the name of a remote station, however the number of characters displayed in this box is limited.

Enter the phone number of the remote station in this box.

Click this button to store the set data and exit from this screen.

Enter a numeric value in the Outside Line box to specify the extension number. The numeric values allowed in this box are 0 to 9. Only enter data here if an extension number such as "0+phone number" is needed. Enter null data if an extension number is not necessary.

- 4) Click the OK button.
- 5) Check the data set in the dialog box.
- 6) Click the OK button.
- 7) The confirmation message "Connect to remote station?" will appear on the display. Select Yes to connect the line to the specified remote station. Select No not to connect the line.
- 8) The new remote station is now stored in the list.



To store new data or modified data, be sure to follow steps 6) and 7) after entering data in the dialog box. To erase the data set in the dialog box after following steps 1) through 5), click the Cancel button or close the dialog box. The data entered will be erased, and the data currently stored will not be modified.

Changing remote station data

Follow the procedure below to change the remote station data stored in the list:

- 1) Select Remote and then Connect. Then, select To PLC or To File Transfer depending on your purpose.
- 2) Click the remote station to be changed.
- 3) Click the Change button.
- 4) Follow step 3 and subsequent steps described in "Storing new remote station" to change the remote station data.

Copying or deleting destination data

- Follow the procedure below to create another remote station using a remote station data stored in the list.
- 1) Select Remote and then Connect. Then, select To PLC or To File Transfer depending on your purpose.
- 2) Click the remote station to be copied or deleted.
- 3) Click the Copy button. (The data will not be deleted from the currently-displayed position.) Click the Cut button. (The data will be deleted from the currently- displayed position.)
- 4) Click the Paste button.
- 5) The copied or cut data will be pasted at the end line of the list.
- 6) Follow the procedure described in "Changing remote station data" to change the pasted data.
- Follow the procedure below to delete a remote station data from the list.
- 1) Click a remote station to be deleted.
- 2) Click the Delete button.
- 3) Click the OK button. The confirmation message "Connect to remote station?" will appear on the display. Select No.



If the Cancel button is clicked instead of the OK button, or if the dialog box is closed without clicking the OK button, the specified remote station will not be deleted.

Changing remote station order

Follow the procedure below to change the remote station order in the list:

- 1) Click a remote station in the list to change its position in the list.
- 2) Click the Up, Down, Top, or Bottom button to move the remote station data to the desired position.
- 3) Click the OK button. The confirmation message "Connect to remote station?" will appear on the display. Select No.



If you click the Cancel button in place of the OK button, or if you close the dialog box without clicking the OK button, the specified remote station will not be deleted.

9.3.2 Disconnect

Use this function to disconnect the telephone line on the transmitting side.

If a communication or operation error occurs, or if FXGP/WIN-E is closed, the telephone line will be automatically disconnected. To manually disconnect the telephone line, click Disconnect.



Click this button to disconnect the line.

To disconnect the telephone line on the receiving side in the file transfer mode, click Cancel in the To File Transfer dialog box.



How to disconnect the line after abnormal end of Windows or FXGP/WIN-E:

If abnormal end of the Windows or FXGP/WIN-E occurs, turn off the power of the modem to disconnect the telephone line.

When the software ends abnormally, the PC cannot control the telephone line, therefore the line will not be disconnected automatically.

🙏 MITSUBISHI

2

3

4

5

6

7

8

9

Α

В

С

D

E

9.3.3 File transfer

SWOPC-	FXGP/WIN	
<u>R</u> emote <u>H</u> elp		
<u>C</u> onnect ▶		
<u>D</u> isconnect		Use this command to transfer files to the receiver PC in the
<u>F</u> ile Transfer	<u>T</u> ransmit	ready mode.
Enviroment	<u>R</u> eceive	Use this command to set the PC on the receiving side to ready
		mode.

Transmit

When the telephone line is connected, sequentially click Remote, File Transfer, and then Transmit.

1) Select files you would like to send.

	File Transmit		
File Name: new.pmc new.pmw new.ptw modemdat.dat new.cow new.dmw new.pmc new.ptw news.cow news.cow news.dmw	Directories: c:\fxgpwin C:\ C:\ C:\ C:\ C:\ C:\ C:\ C:\ C:\ C:\	+	OK Cancel Help
List Files of <u>T</u> ype: All Files(*.*)	Dri <u>v</u> es:	Ł	

Select files, and then click the OK button. The screen shown in step 2) will appear on the display.

Selection of two or more files:

(shift) + clicking mouse (for selection of continuous files)

Ctrl + clicking mouse (for selection of separate files)

Files that can be sent: Data files created by FXGP/WIN-E and other general files

2) Specify a directory on the receiver PC to send files to the specified directory.

Set destination directory			
Set Destination path ? O NO © YES Destination directory [C:\DATA\PROGRAM	<u>O</u> K <u>C</u> ancel <u>H</u> elp	Click this button to send files.	

- If you select NO, the files will be stored in the sub-directory FILE of the FXGP/WIN-E on the receiving side.
- If you select YES, enter the full directory name in the Destination Directory field. If you specify a nonexisting directory, the confirmation message "Create a new directory?" will appear on the display. If you select YES, a sub- directory will be created in the PC on the receiving side. If you select NO, file transfer will be canceled, and the telephone line will be disconnected.

For a detailed description of file transfer procedure, refer to Sec.9.2.2.



Receive

Sequentially click Remote, File Transfer, and then Receive.



Click this button to cancel file transfer. The telephone line will be disconnected.

The files will be stored in the specified directory, when files are sent from the PC on the sending side to the PC in the ready mode on the receiving side.

For a detailed description of file transfer procedure, refer to Sec.9.2.2.

9.3.4 Environment \rightarrow Modem

Specify and store a modem to be connected to the PC. To specify a modem, sequentially click Remote, Environment, and then Modem.



Use these buttons to change the position of the selected modem data in the list.

Delete button: Deletes the modem data selected.

Selecting modem

Follow the procedure below to select the modem used for telephone line connection:

- 1) Click the modem name in the list. The modem name will be highlighted.
- 2) Click the OK button.

1

Storing new modem

Follow the procedure below to store a new modem in the list.

- 1) Click the New Entry button.
- 2) Enter data in the modem storage screen.

_ <mark></mark>	Modem Setting	
<u>N</u> ame AIWA	, ****	
Initial Setup String	ATEOSO=2Q0V1&M4¥J0¥Q0	
<u>R</u> e-Dial	0 🔶 Times	
Waiting <u>T</u> ime	0 🔶 Sec	
Line Classification-		<u><u>C</u>ancel</u>
● PULSE		<u>H</u> elp
⊂ TONE		

Enter the modem name to be stored in the list. Up to 40 characters can be used for the modem name.

Enter the AT command to initialize the modem. (For a detailed description, refer to the next page.)

Click this button to store the set data and close this window.

Set the type of the telephone line.

Use numeric values 0 to 9 to set the number of re-dialing times. Use numeric values 0 to 999 to set the re- dialing intervals.

- 3) Click the OK button.
- 4) Click the OK button in the Modem dialog box.



If the Cancel button is clicked instead of the OK button, or if the dialog box is closed without clicking the OK button, the set data will be erased.

2

3

4

5

6

7

8

9

Α

В

F

Initialize command setting:

Use the AT command developed by Heiz in the U.S. to initialize the modem. For a detailed description of the AT command, refer to the operation manual of your modem.

The AT commands stored in this software set the modems as shown in the table below.

The set items and data depends on the modem type. For the set items and data, therefore, refer to the operation manual of the modem, if not using one of the modems below.

Set item	PV-AF288(AIWA) ATE0S0=2Q0V1&M4\J0\Q0	PV-AF3360(AIWA) ATE0S0=2Q0V1&M4\J0\Q0	ME3314(OMRON) ATE0S0=2Q0V1S15=8&H0&R1		
Setting of command echo	E0 (not set)	E0 (not set)	E0 (not set)		
Call frequency of auto-answering	S0=2 (twice)	S0=2 (twice)	S0=2 (twice)		
Displaying result code	Q0 (displayed)	Q0 (displayed)	Q0 (displayed)		
Type of result code	V1(characters, words)	V1(characters, words)	V1(characters, words)		
Communication mode	&M4(MNP auto)	&M4(MNP auto)	S15=8(V,42bis)		
Speed of terminal fixed dimension	\J0(fixed) \J0(fixed)		—		
Flow control of transmission data			&H0(not set)		
Terminal flow control scheme	\Q0(not set) \Q0(not set)		&R1(not set)		
Initialize to factory-set value	&F(The FXGP/WIN-E will output the above AT command after outputting AT&F.)				

Changing set modem data

Follow the procedure below to change the modem data stored in the list:

- 1) Click the modem name to change the modem data.
- 2) Click the Change button.
- 3) Follow step 2) and the subsequent steps described in "Storing new modem" to change the modem data.

Copying or deleting set data

- Follow the procedure below to create another modem data using a modem data stored in the list:
- 1) Click on a modem to copy or delete.
- 2) Click the Copy button. (The data will not be deleted from the currently-displayed position.) Click the Cut button. (The data will be deleted from the currently- displayed position.)
- 3) Click the Paste button.
- 4) The copied or cut data will be pasted at the end line of the list.
- 5) Follow the procedure described in "Changing set modem data" to change the pasted data.
- To delete a modem data from the list, select a modem from the list, and then click the Delete button. After that, click the OK button.



If the Cancel button is clicked instead of the OK button, if the dialog box is closed without clicking the OK button, the specified modem data will not be deleted.

Changing modem order

Follow the procedure below to change the modem order in the list:

- 1) Click a modem in the list to change the position in the list.
- Click the Up, Down, Top, or Bottom button to move the specified modem data to the desired position in the list.
- 3) Click the OK button.



If you click the Cancel button in place of the OK button, or if you close the dialog box without clicking the OK button, the modem order will not be changed.



$\textbf{9.3.5} \qquad \textbf{Environment} \rightarrow \textbf{Logging File}$

Use the Logging File command to specify whether the communication log file should be created. Sequentially click Remote, Environment, and then Logging File.



FXGP/WIN-E will automatically create the log file.

To temporarily stop outputting of the log file, select No in the log file output dialog box.

This No command will be valid until you quit the FXGP/WIN-E or cancel the No command.

When the FXGP/WIN-E is activated next time, the log file will be output again.

After temporarily stopping the output, select Yes to output the log file again.

The log file will be stored in the sub-directory LOG of the FXGP/WIN-E installed directory (initial value = \FXGPWIN). The format of log file is shown below:

File name:	9 <u>7</u> 07 year, month,	<u>10</u> . LOG day	
Logged data:	Connect Name Phone Disconnect Time	97/ 07/ 18 ABCDEFG 123-123-123 97/ 07/ 18 00: 18: 54	-

If the line is connected two or more times in one day, the communication data will be sequentially added in the same log file.

Communication data will be logged in text files. A standard text editor can be used to read these files.

2

3

4

5

6

7

8

9

Α

B

С

D

Ε

9.4 Data Setting on PLC Side for Remote Access

To access the remote FX1S, FX1N, FX2N and FX2NC PLC, set the parameters for the modem, and set communication data for the special diagnostic register.

Use the following programming software to enable modem settings for the PLC.

The remote maintenance function cannot use on the WindowsNT 4.0, Windows 2000 or Windows XP.

- FX-PCS/WIN-E, version 1.00 or later (for Windows)
- GX Developer (SWDD5C-GPPW), version SW2 or later (for Windows)

Main setting

- 1) Read out the program from the FX1S, FX1N, FX2N and FX2NC PLC to the peripheral unit.
- 2) Activate the programming software, and click PLC's Mode Setting to select the modem used for communication.

Example: Modem setting using FXGP/ WIN (See Sec.6.8.4.)

	PLC's	mode setting	
Batteryless mode	OON	● OFF	<u>O</u> K
<u>M</u> odem initialize	None		± <u>C</u> ancel
RUN from input RUN terminal input number	AIWA(P)	try mode (- AF288)	<u>H</u> elp
Tour rouning librar inter	UMRON	(ME3314B)	

Select a modem from the Modem Initialize pull-down menu.

If no modem is stored, click User Entry Mode, and set _ the initialize command for the modem.

 Set the D8120 (communication format) value to zero. The D8120 is one of the special diagnostic registers of the PLC.

If the current value of the D8120 is "0":

It is not necessary to rewrite the D8120 value. Go to the next step.

If the D8120 value is written in the program:

Change the program of the D8120 as follows so that the current value of the D8120 is set to "0".



If communication conditions are set by the parameters of the peripheral unit:

The set communication conditions will be written to D8120 at power-on of the PLC. If the communication conditions are to be set by the parameters of the peripheral unit, then clear the original contents of D8120 using the peripheral unit. Use the following communications software to clear the communication conditions.

• FX-PCS/WIN-E

Click Option and then Serial Setting (Parameter). The Serial Setting (Parameter) dialog box will appear on the display. Click the Clear command in this dialog box.

- If you use another programming software to clear the parameters, follow the procedure below:
 - a) Store the parameter data currently set.
 - b) Clear all the parameters.
 - c) Excluding the Serial Setting, set the parameters and keyword to specify the communication conditions.
 - d) Click PLC's Mode Setting, and set the modem as described in step 2).
- 4) Write all the programs and parameters for the FX1S, FX1N, FX2N and FX2NC series PLC.
- 5) If the programs are changed by following step 3), click RUN and then STOP to stop the PLC, and then reset the current values of the D8120 to zero.

Setting user setting mode

Select the User Setting Mode from the Modem Initialize pull-down menu (see step 2) of "Main setting"), and store the appropriate initialize commands for the modem as described below:

Initialize command format:

Use the AT command developed by Heiz in the U.S. The general format of this command is shown below:

А	Т	command+parameter	command+parameter	command+parameter		CR	LF	
---	---	-------------------	-------------------	-------------------	--	----	----	--

For a detailed description of the AT command, refer to the operation manual of the modem.

Storing AT command in PLC:

If User Setting Mode in the PLC's Mode Setting dialog box is selected as described before, the FX1S, FX1N, FX2N and FX2NC series PLC will send the AT&F command (command for resetting to factory-set values) and the data in D1000 through D1059 (modem initialize commands) to the modem via the RS-232C communication adapter/board at power- on of the PLC.

To use the user-set modem initialize commands, previously write the modem initialize commands in data registers D1000 through D1059 using the current value modify function of the peripheral unit or using the program.

Register number	ASCII code	Hexadecimal number	Register number	ASCII code	Hexadecimal number
D1000	A	41	D1013	&	26
D1001	Т	54	D1014	М	4D
D1002	E	45	D1015	4	34
D1003	0	30	D1016	١	5C
D1004	S	53	D1017	Q	51
D1005	0	30	D1018	0	30
D1006	=	3D	D1019	١	5C
D1007	2	32	D1020	J	4A
D1008	Q	51	D1021	0	30
D1009	1	31	D1022	&	26
D1010	&	26	D1023	W	57
D1011	D	44	D1024	CR	0D
D1012	0	30	D1025	LF	0A

Examples of modem initialize commands: ATE0S0 = 2Q1& D0& M4\Q0\JO& W

Up to 60 points of user-set initialize commands can be read out sequentially from the D1000 to D1059. If "00" is read out, reading out of initialize commands will be stopped.

When creating programs, check that the data register area used for the other programs is not the area for these modem initialize commands.



Description of modem setting data previously stored:

The modem setting conditions of the AT command previously stored in the PLC are shown below. The modem setting conditions shown below are for reference. Actual conditions depend on the modem. Please refer to the operation manual of the modem being used.

Setting item	PV-AF288(AIWA) ATE0S0=2Q1&D0&M5\Q0\J0&W	ME3314(OMRON) ^{*1} ATE0S0=2Q1&D0&H0&R1S15=8 &W
Setting of command echo	E0(not set)	E0(not set)
Call frequency of auto-answering	S0=2(twice)	S0=2(twice)
Displaying result code	Q1(not displayed)	Q1(not displayed)
DTR control	&D0(always ON)	&D0(always ON)
Communication mode	&M5(V,42bis)	S15=8(V,42bis)
Terminal flow control method	\Q0(not set)	&R1(not set)
Flow control of transmission data		&H0(not set)
Speed of terminal fixed dimension	\J0(fixed)	
Writing in nonvolatile memory	&W	&W

*1: The AT command setting function for the ME3314 (OMRON) is added to the FX2N and FX2NC series PLC version 2.01 or later (manufacturing number 78**** or later).

Programming data

For a detailed description of the PLC and programs, refer to the following documents:

1

2



MEMO

Appendix A: PLC Device Tables

For further information about device of the FX₀, FX₀s, FX₀N, FX and FX₂c Series PLC can be found in FX Programming Manual (Manual number: JY992D48301), about device of the FX₁s, FX₁N, FX₂N and FX₂Nc Series PLC can be found in FX Programming Manual II (Manual number: JY992D88101).



A-1: FX0/FX0s Series PLC

Latched devices range is fixed by PLC's parameter. So that, these cannot be changed in the Parameter setting menu on FXGP/WIN-E.

Item		Specification	Remarks	
I/O configuration		Max total I/O set by Main Processing Unit		
	General	496 points	M0 to M495	
Auxiliary relay (M coils)	Latched	16 points (EEPROM Backed-up)	M496 to M511	
	Special	56 points	From the range M8000 to M8255	
State relays	General	64 points	S0 to S63	
(S coils)	Initial	10 points (subset)	S0 to S9	
	100 msec	Range: 0 to 3,276.7 sec 56 points	T0 to T55	
Timers (T)	10 msec	Range: 0 to 327.67 sec 24 points	T32 to T55 when special M coil M8028 is driven ON	
Counters (C)	General	Range: 1 to 32,767 counts 14 points	C0 to C13 Type: 16 bit up counter	
Counters (C)	Latched	Range: 1 to 32,767 counts 2 points (EEPROM Backed-up)	C14 to C15 Type: 16 bit up counter	
	1 phase	Range: -2,147,483,648 to +2,147,483,647 counts FXo: Select upto four 1 phase counters with a combined counting frequency of 5kHz or less.	C235 to C238, 4 points Note; C235 is latched (EEPROM backed-up)	
High speed	1 phase c/w start stop input	Alternatively select one 2 phase or A/B phase counter with a counting frequency of 2kHz or less. FXos: When multiple 1-phase counters are used	C241, C242 and C244, 3 points Note; C241 and C244 are latched (EEPROM backed-up)	
counters (C) Max. 4 points	2 phase	the sum of the frequencies must be equal or less than 14kHz. Only 1, 2 phase high speed counter may be used at any one time. When 2 phase counters are in use the maximum counted speeds	C246, C247 and C249, 3 points Note; These are latched (EEPROM backed-up)	
	A/B phase	must be equal or less than 14kHz, calculated as (2 ph counter speed 5 number of counted edges) + 1 ph counter speeds.	C251, C252 and C254, 3 points Note; These are latched (EEPROM backed-up)	
	General	30 points	D0 to D29 Type: 16 bit data storage register pair for 32 bit device	
Data	Latched	2 points (EEPROM Backed-up)	D30 to D31 Type: 16 bit data storage register pair for 32 bit device	
registers (D)	Externally adjusted	Range: 0 to 255 1 point	D8013 Data is entered indirectly through the external setting potentiometer	
	Special	27 points (inclusive of D8013)	From the range D8000 to D8255 Type: 16 bit data storage register	
	Index	2 points	V and Z Type: 16 bit data storage register	
Pointers (P)	For use with CALL	64 points	P0 to P63	
	For use with interrupts	4 points	$100\Box$ to $130\Box$ (rising trigger $\Box = 1$, falling trigger $\Box = 0$)	
Nest levels		8 points for use with MC and MCR	N0 to N7	
Constants	Decimal K	16 bit: -32,768 t 32 bit: -2,147,483,648 t	0 +2,147,483,647	
e eetanto	Hexadecimal H	16 bit: 0000 to FFFF 32 bit: 00000000 to FFFFFFFF		

Table A-1: Device Table for FXo and FXos Series PLC

Γ

A-2: FXoN Series PLC

Latched devices range is fixed by PLC's parameter. So that, these cannot be changed in the Parameter setting menu on FXGP/WIN-E.

 Table A-2:
 Device Table for FXon Series PLC

Item		Specification	Remarks	2
I/O configuration		Max hardware I/O configuration points 128, dependent on user selection (Max. software addressable Inputs 128, Outputs 128)		
	General	384 points	M0 to M383	
Auxiliary relay (M coils)	Latched	128 points (EEPROM Backed-up)	M384 to M511	3
	Special	65 points	From the range M8000 to M8255	
	Latched	128 points (EEPROM Backed-up)	S0 to S127	
State relays (S coils)	Initial (Latched)	10 points (subset)	S0 to S127	
(0 00)		Range: 0 to 3,276.7 sec	001000	
	100 msec	63 points	T0 to T62	
Timers (T)	10 msec	Range: 0 to 327.67 sec 31 points	T32 to T62 when special M coil M8028 is driven ON	5
	1 msec	Range: 0 to 32.767 sec 1 point	Т63	
Counters (C)	General	Range: 1 to 32,767 counts 16 points	C0 to C15 Type: 16 bit up counter	6
	Latched	Range: 1 to 32,767 counts 16 points (EEPROM Backed-up)	C16 to C31 Type: 16 bit up counter	
	1 phase		C235 to C238 4 points	
	1 phase	Range: -2,147,483,648 to +2,147,483,647 counts Select upto four 1 phase counters with a combined	C241, C242 and C244	
High speed	c/w start stop input	counting frequency of 5kHz or less.	3 points	
counters (C) Max. 4points	stop input	Alternatively select one 2 phase or A/B phase	C246, C247 and C249	8
Max. +pointo	2 phase	counter with a counting frequency of 2kHz or less. Note all counters are latched (EEPROM Backed-up)	3 points	
	A/D about		C251, C252 and C254	
	A/B phase		3 points	
	General	128 points	D0 to D127 Type: 16 bit data storage register pair for 32 bit device	
	Latched	128 points (EEPROM Backed-up)	D128 to D255 Type: 16 bit data storage register pair for 32 bit device	ŀ
Data registers (D)	File	1500 points	D1000 to D2499 set by parameter in 3 blocks of 500 program steps Type: 16 bit data storage register	E
registers (D)		Externally Range: 0 to 255 potentiometers to registers adjusted 2 points D8030 and D8031 (D8013 is used when	Data is move from external setting	
	Externally adjusted			
			RTC is fitted)	(
	Special	42 points (inclusive of D8013, D8030 and D8031)	From the range D8000 to D8255 Type: 16 bit data storage register	
	Index	2 points	V and Z Type: 16 bit data storage register	[
Pointers (P)	For use with CALL	64 points	P0 to P63	
	For use with interrupts	4 points	$100\Box$ to $130\Box$ (rising trigger $\Box = 1$, falling trigger $\Box = 0$)	E
Nest	levels	8 points for use with MC and MCR	N0 to N7	
Constants	Decimal K	16 bit: -32,768 to 32 bit: -2,147,483,648 to	9 +2,147,483,647	
Jonotunio		16 bit: 0000 to	FEFE	



A-3: FX Series PLC (Version 2.30 or earlier)

Latched devices range can change in the Parameter setting menu on FXGP/WIN-E. For explanation about operation, refer to section 6.8.2.

 Table A-3:
 Device Table for FX Series PLC (Version 2.30 or earlier)

Item		Specification	Remarks
I/O configuration		Max hardware I/O configuration points 256, dependent on user selection (Max. software addressable Inputs 128, Outputs 128)	
Auxiliany	General	1024 points, Default: 500 points (subset)	M0 to M1023, Default: M0 to M499
Auxiliary relay (M coils)	Latched	1024 points (Battery backed-up), Default: 524 points (subset)	M0 to M1023, Default: M500 to M1023
	Special	256 points	From the range M8000 to M8255
	General	1000 points, Default: 500 points (subset)	S0 to S999, Default: S0 to S499
State relays	Latched	1000 points (Battery backed-up), Default: 500 points (subset)	S0 to S999, Default: S500 to S999
(S coils)	Initial	10 points (subset)	S0 to S9
	Annunciator	100 points	S900 to S999
	100 msec	Range: 0 to 3,276.7 sec 200 points	T0 to T199
Timers (T)	10 msec	Range: 0 to 327.67 sec 46 points	T200 to T245
	1 msec retentive	Range: 0 to 32.767 sec 4 points (Battery backed-up)	T246 to T249
	100 msec retentive	Range: 0 to 3,276.7 sec 6 points (Battery backed-up)	T250 to T255
	General 16 bit	Range: 1 to 32,767 counts 200 points, Default: 100 points (subset)	C0 to C199, Default: C0 to C99 Type: 16 bit up counter
Counters (C)	Latched 16 bit	Range: 1 to 32,767 counts 200 points (Battery backed-up), Default: 100 points (subset)	C0 to C199, Default: C100 to C199 Type: 16 bit up counter
Counters (C)	General 32 bit	Range: -2,147,483,648 to 2,147,483,647 35 points, Default: 20 points (subset)	C200 to C234, Default: C200 to C219 Type: 32 bit up/down counter
	Latched 32 bit	Range: -2,147,483,648 to 2,147,483,647 35 points (Battery backed-up), Default: 15 points (subset)	C200 to C234, Default: C220 to C234 Type: 16 bit up/down counter
	1 phase	Range: -2,147,483,648 to +2,147,483,647	C235 to C240, 6 points
High speed counters (C)	1 phase c/w start stop input	counts General rule: Select counter combinations with a combined counting frequency of 20kHz or less. Note all counters are latched (Battery	C241 to C245, 5 points
Max. 6 points -	2 phase		C246 to C250, 5 points
	A/B phase	backed-up)	C251 to C255, 5 points
Data registers (D)	General	512 points, Default: 200 points (subset)	D0 to D511, Default: D0 to D199 Type: 16 bit data storage register pair for 32 bit device
	Latched	512 points (Battery backed-up), Default: 312 points (subset)	D0 to D511, Default: D200 to D511 Type: 16 bit data storage register pair for 32 bit device
	File	Maximum 2000 points	D1000 to D2999 set by parameter in 4 blocks of 500 program steps Type:16 bit data storage register
	Special	256 points	From the range D8000 to D8255 Type: 16 bit data storage register
	Index	2 points	V and Z Type: 16 bit data storage register

2

Item		Specification	Remarks	
	For use with CALL	64 points	P0 to P63	
Pointers (P)	For use with interrupts	6 input points and 3 timers	l00급 to l50⊒ and l6☆☆ to l8☆☆ (rising trigger ⊒=1, falling trigger ⊒=0, ☆☆=time in msec)	
Nest levels		8 points for use with MC and MCR	N0 to N7	
Constants	Decimal K	16 bit: -32,768 to +32,767 32 bit: -2,147,483,648 to +2,147,483,647		
Constants	Hexadecimal H	16 bit: 0000 to FFFF 32 bit: 0000000 to FFFFFFF		

Table A-3: Device Table for FX Series PLC (Version 2.30 or earlier)



A-4: FX (Version 3.07 or later)/FX_{2C} (All version) Series PLC

Latched devices range can change in the Parameter setting menu on FXGP/WIN-E. For explanation about operation, refer to section 6.8.2. However, fixed latched device cannot be changed by it.

Item			Specification	Remarks	
I/O configuration			Max hardware I/O configuration points 255, dependent on user selection (Max. software addressable Inputs 255, Outputs 255)		
	General		1024points, Default: 500 points (subset)	M0 to M1023, Default: M0 to M499	
Auxiliary relay	Latched	Subset	1024 points (Battery backed-up), Default: 524 points (subset)	M0 to M1023, Default: M500 to M1023	
(M coils)		Fixed	512 points (Battery backed-up)	M1024 to M1535	
	Special		256 points	From the range M8000 to M8255	
	Gen	eral	1000 points, Default: 500 points (subset)	S0 to S999, Default: S0 to S499	
State relays (S coils)	Lato	hed	1000 points (Battery backed-up), Default: 500 points (subset)	S0 to S999, Default: S500 to S999	
(3 colls)	Init	ial	10 points (subset)	S0 to S9	
	Annur	ciator	100 points	S900 to S999	
	100 msec		Range: 0 to 3,276.7 sec 200 points	T0 to T199	
Timers (T)	10 msec		Range: 0 to 327.67 sec 46 points	T200 to T245	
Timers (T)	1 msec retentive		Range: 0 to 32.767 sec 4 points (Battery backed-up)	T246 to T249	
	100 msec retentive		Range: 0 to 3,276.7 sec 6 points (Battery backed-up)	T250 to T255	
	General 16 bit		Range: 1 to 32,767 counts 200 points, Default: 100 points (subset)	C0 to C199, Default: C0 to C99 Type: 16 bit up counter	
Counters (C)	Latched 16 bit		Range: 1 to 32,767 counts 200 points (Battery backed-up), Default: 100 points (subset)	C0 to C199, Default: C100 to C199 Type: 16 bit up counter	
	General 32 bit		Range: -2,147,483,648 to 2,147,483,647 35 points, Default: 20 points (subset)	C200 to C234, Default: C200 to C219 Type: 32 bit up/down counter	
	Latched 32 bit		Range: -2,147,483,648 to 2,147,483,647 35 points (Battery backed-up), Default: 15 points (subset)	C200 to C234, Default: C220 to C234 Type: 16 bit up/down counter	
High speed counters (C) Max. 6 points	1 phase 1 phase c/w start stop input 2 phase A/B phase		Range: -2,147,483,648 to +2,147,483,647 counts General rule: Select counter combinations with a combined counting frequency of 20kHz or less.	C235 to C240, 6 points	
				C241 to C245, 5 points	
				C246 to C250, 5 points	
				C251 to C255, 5 points	

 Table A-4:
 Device Table for FX (Version 3.07 or later) and FX₂c Series PLC
	ltem		Specification	Remarks
	Gen	eral	512 points, Default: 200 points (subset)	D0 to D511, Default: D0 to D199 Type: 16 bit data storage register pair for 32 bit device
	Latched	Subset	512 points (Battery backed-up), Default: 312 points (subset)	D0 to D511, Default: D200 to D511 Type: 16 bit data storage register pair for 32 bit device
Dete	Lateneu	Fixed	488 points (Battery backed-up)	D512 to D999 Type: 16 bit data storage register pair for 32 bit device
Data registers (D)	Fi regis		Maximum 2000 points	D1000 to D2999 set by parameter in 4 blocks of 500 program steps Type: 16 bit data storage register
	RAN regis		2000 points	D6000 to D7999 active when special relay M8074 is active Type: 16 bit data storage register
	Spe	cial	256 points	From the range D8000 to D8255 Type: 16 bit data storage register
	Ind	lex	2 points	V and Z Type: 16 bit data storage register
	For with 0		128 points	P0 to P127
Pointers (P)	For us interr		6 input points, 3timers and 6 counters	I00❑ to I50❑ and I6☆☆ to I8☆☆ (rising trigger ❑=1, falling trigger ❑=0, ☆☆=time in msec)
Ne	st levels		8 points for use with MC and MCR	N0 to N7
	Decir	nal K	16 bit: -32,768 32 bit: -2,147,483,648	
Numbers	Hexade	cimal H	16 bit: 0000 32 bit: 00000000	
	Floa Po	•	32 bit: 0, ±1.175 x 10- (Not directly e	•

Table A-4: Device Table for FX (Version 3.07 or later) and FX₂C Series PLC



A-5: FX1s Series PLC

Latched devices range is fixed by PLC's parameter. So that, these cannot be changed in the Parameter setting menu on FXGP/WIN-E.

	tem	or FX1S Series PLC Specification	Remarks					
	figuration		Main Processing Unit					
	-	-	M0 to M383					
Auxiliary	General	384 points						
relay	Latched	128 points (EEPROM backed-up)	M384 to M511					
(M coils)	Special	256 points	From the range M8000 to M8255					
State relays	General	128 points	S0 to S127					
(S coils)	Initial	10 points (subset)	S0 to S9					
	100 msec	Range: 0 to 3,276.7 sec 63 points	T0 to T62					
Timers (T)	10 msec	Range: 0 to 327.67 sec 31 points	T32 to T62 when special M coil M8028 is driven ON					
	1 msec	Range: 0.001 to 32.767 sec 1 point	Т63					
	Quant	Range: 1 to 32,767 counts	C0 to C15					
	General	16 points	Type: 16 bit up counter					
Counters (C)	Latabad	Range: 1 to 32,767 counts	C16 to C31					
	Latched	16 points (EEPROM backed-up)	Type: 16 bit up counter					
	1 phase	Range: -2,147,483,648 to +2,147,483,647	C235 to C240, 6 points					
	1 phase	counts						
	c/w start	General rule: Select counter combinations	C241to C245, 5 points					
High speed	stop input	with a combined counting frequency of 60kHz or less.						
counters (C)	2 phase	Note all counters are latched (EEPROM	C246 to C250, 5 points					
Max. 6 points	A/B phase	backed-up) If high speed counter is used with the HSCS or HSCR instruction, a combined counting frequency of 30kHz or less.	C251 to C255, 5 points					
	General	128 points	D0 to D127 Type: 16 bit data storage register pair for 32 bit device					
	Latched	128 points (EEPROM backed-up)	D128 to D255 Type: 16 bit data storage register pair for 32 bit device					
Data registers (D)	File registers	Maximum 2000 points	D1000 to D2499 set by parameter in 3 blocks of 500 program steps Type: 16 bit data storage register					
	Externally adjusted	Range: 0 to 255 2 points	D8030 & D8031 Data is entered indirectly through the external setting potentiometer					
	Special	256 points (inclusive of D8013)	From the range D8000 to D8255 Type: 16 bit data storage register					
	Index	16 points	V and Z Type: 16 bit data storage register					
Pointers (P)	For use with CALL	64 points	P0 to P63					
	For use with interrupts	6 points	$100\Box$ to $150\Box$ (rising trigger $\Box = 1$, falling trigger $\Box = 0$)					
Nes	t levels	8 points for use with MC and MCR	N0 to N7					
Constants	Decimal K		1 68 to +32,767 648 to +2,147,483,647					
Constants	Hexadecimal H		00 to FFFF 00 to FFFFFFFF					

 Table A-5:
 Device Table for FX1s Series PLC



2

3

4

5

6

7

8

9

Α

В

С

D

Ε

A-6: FX1N Series PLC

Latched devices range is fixed by PLC's parameter. So that, these cannot be changed in the Parameter setting menu on FXGP/WIN-E.

lte	em	Specification	Remarks
I/O conf	iguration		oints 128, dependent on user selection able Inputs 128, Outputs 128)
	General	384 points	M0 to M383
Auxiliary relay (M coils)	Latched	1152 points EEPROM backed-up: 128 points Capacitor backed-up: 1024 points	M384 to M1535 EEPROM keep: M384 to M511 Capacitor keep: M512 to M1535
	Special	256 points	From the range M8000 to M8255
State relays (S coils)	Latched	1000 points EEPROM backed-up: 128 points Capacitor backed-up: 872 points	S0 to S999 EEPROM keep: S0 to S127 Capacitor keep: S128 to S999
	Initial	10 points (subset)	S0 to S9
	100 msec	Range: 0 to 3,276.7 sec 200 points	T0 to T199
Timers (T)	10 msec	Range: 0 to 327.67 sec 46 points	T200 to T245
	1 msec retentive	Range: 0 to 32.767 sec 4 point, Capacitor backed-up	T246 to T249
-	100 msec retentive	Range: 0 to 3,276.7 sec 6 points, Capacitor backed-up	T250 to T255
	General	Range: 1 to 32,767 counts 16 points	C0 to C15 Type: 16 bit up counter
Counters (C)	Latched	Range: 1 to 32,767 counts 184 points EEPROM backed-up: 16 points Capacitor backed-up: 168 points	C16 to C199 EEPROM keep: C16 to C31 Capacitor keep: C32 to C199 Type: 16 bit up counter
	General	Range: -2,147,483,648 to 2,147,483,647 20 points	C200 to C219 Type: 32 bit bi-directional counter
	Latched	Range: -2,147,483,648 to 2,147,483,647 15 points, Capacitor backed-up	C220 to C234 Type: 32 bit bi-directional counter
	1 phase	Range: -2,147,483,648 to +2,147,483,647	C235 to C240, 6 points
Llich or and	1 phase c/w start stop input	counts General rule: Select counter combinations with a combined counting frequency of	C241to C245, 5 points
High speed - counters (C)	2 phase	– 60kHz or less. – Note:	C246 to C250, 5 points
Max. 6 points	A/B phase	all counters are latched (EEPROM backed- up) If high speed counter is used with the HSCS or HSCR instruction, a combined counting frequency of 30kHz or less.	C251 to C255, 5 points

 Table A-6:
 Device Table for FX1N Series PLC

A MITSUBISHI

lt	em	Specification	Remarks							
	General	128 points	D0 to D127 Type: 16 bit data storage register pair for 32 bit device							
	Latched	7872 points EEPROM backed-up: 128 points Capacitor backed-up: 7744 points	D128 to D7999 EEPROM keep: D128 to D255 Capacitor keep: D255 to D7999 Type: 16 bit data storage register pair for 32 bit device							
Data registers (D)	File registers	Maximum 7000 points	D1000 to D7999 set by parameter in 14 blocks of 500 program steps Type: 16 bit data storage register							
	Externally adjusted	Range: 0 to 255 2 points	D8030 & D8031 Data is entered indirectly through the external setting potentiometer							
	Special	256 points (inclusive of D8030 and D8031)	From the range D8000 to D8255 Type: 16 bit data storage register							
	Index	16 points	V and Z Type: 16 bit data storage register							
Pointers (P)	For use with CALL	128 points	P0 to P127							
	For use with interrupts	6 points	$100\Box$ to $150\Box$ (rising trigger $\Box = 1$, falling trigger $\Box = 0$)							
Nest	levels	8 points for use with MC and MCR	N0 to N7							
Constants	Decimal K		2,768 to +32,767 33,648 to +2,147,483,647							
Constants	Hexadecimal H		0000 to FFFF 0000 to FFFFFFFF							

Table A-6: Device Table for FX1N Series PLC

2

Ε

A-7: FX2N/FX2NC Series PLC

Latched devices range can change in the Parameter setting menu on FXGP/WIN-E. For explanation about operation, refer to section 6.8.2. However, fixed latched device cannot be changed by it.

Table A-7: Device Table for FX2N and FX2NC Series PLC

	ltem		Specification	Remarks
I/O co	onfiguration		Max hardware I/O configuration points 255 (Max. software addressable Inpu	
	Gen	eral	1024 points (Battery backed-up), Default: 500 points (subset)	M0 to M1023, Default: M0 to M499
Auxiliary relay (M coils)	Latched	Subset	1024 points (Battery backed-up), Default: 524 points (subset)	M0 to M1023, Default: M500 to M1023
		Fixed	2048 points (Battery backed-up)	M1024 to M3071
	Spe	cial	256 points	From the range M8000 to M8255
	Gen	əral	1000 points, Default: 500 points (subset)	S0 to S999, Default: S0 to S499
State relays	Latcl	ned	1000 points (Battery backed-up), Default: 500 points (subset)	S0 to S999, Default: S500 to S999
(S coils)	Init	al	10 points (subset)	S0 to S9
	Annun	ciator	100 points	S900 to S999
	100 n	nsec	Range: 0 to 3,276.7 sec 200 points	T0 to T199
Timoro (T)	10 m	sec	Range: 0 to 327.67 sec 46 points	T200 to T245
Timers (T)	1 ms reten		Range: 0 to 32.767 sec 4 points (Battery backed-up)	T246 to T249
	100 n reten		Range: 0 to 3,276.7 sec 6 points (Battery backed-up)	T250 to T255
	Gen 16		Range: 1 to 32,767 counts 200 points, Default: 100 points (subset)	C0 to C199, Default: C0 to C99 Type: 16 bit up counter
Countors (C)	Latcl 16		Range: 1 to 32,767 counts 200 points (Battery backed-up), Default: 100 points (subset)	C0 to C199, Default: C100 to C199 Type: 16 bit up counter
Counters (C)	Gen 32		Range: -2,147,483,648 to 2,147,483,647 35 points, Default: 20 points (subset)	C200 to C234, Default: C200 to C219 Type: 32 bit up/down counter
	Latel 32		Range: -2,147,483,648 to 2,147,483,647 35 points (Battery backed-up), Default: 15 points (subset)	C200 to C234, Default: C22 to C234 Type: 32 bit up/down counter
	1 ph	ase	Range: -2,147,483,648 to +2,147,483,647 counts	C235 to C240, 6 points
High speed	1 ph c/w s stop i	tart	General rule: Select counter combinations with a combined counting frequency of 20kHz or less. Note all counters are latched (Battery backed-up)	C241 to C245, 5 points
counters (C)	2 ph	ase	If high speed counter is used with the HSCS or HSCR instruction, a combined counting frequency of	C246 to C250, 5 points
lax. 6 points	A/B pl		11kHz or less. If high speed counter is used with the HSZ instruction, a combined counting frequency of 5.5kHz or less.	C251 to C255, 5 points

Table A-7:	-									
	ltem		Specification	Remarks						
	Gen	eral	512 points, Default: 200 points (subset)	D0 to D511, Default: D0 to D199 Type: 16 bit data storage register pair for 32 bit device						
	Latched	Subset	512 points (Battery backed-up), Default: 312 points (subset)	D0 to D511, Default: D200 to D511 Type: 16 bit data storage register pair for 32 bit device						
Data		Fixed	7488 points (Battery backed-up)	D512 to D7999						
registers (D)	File reç	gisters	Maximum 7000 points	D1000 to D7999 set by parameter in 14 blocks of 500 program steps Type: 16 bit data storage register						
	Spe	cial	256 points	From the range D8000 to D8255 Type: 16 bit data storage register						
	Ind	ex	16 points	V0 to V7 and Z0 to Z7 Type: 16 bit data storage register						
	For with C		128 points	P0 to P127						
Pointers (P)	For us interr		6 input points, 3 timers, 6 counters	I00❑ to I50❑ and I6☆☆ to I8☆☆ (rising trigger ❑=1, falling trigger ❑=0, ☆☆=time in msec)						
Ne	st levels		8 points for use with MC and MCR	N0 to N7						
	Decin	nal K	16 bit: -32,768 to + 32 bit: -2,147,483,648 to +							
Numbers	Hexade	cimal H	16 bit: 0000 to FFFF 32 bit: 0000000 to FFFFFFF							
	Floating	g Point	32 bit: 0, ±1.175 x 10 ⁻³⁸ , ±3.403 x 10 ³⁸ (Not directly enterable)							

 Table A-7:
 Device Table for FX2N and FX2NC Series PLC

Appendix B: PLC Instruction Tables

For further information about instruction of the FX₀, FX₀s, FX₀s, FX₀s, FX₂s and FX₂c Series PLC can be found in FX Programming Manual (Manual number: JY992D48301), about the FX₁s, FX₁s, FX₂s and FX₂s Series PLC can be found in FX Programming Manual II (Manual number: JY992D88101).

B-1: Basic Program Instruction Table

Instruction (Mnemonic)	FXo/ FXos	FXON	FX/FX2C	FX1S	FX1N	FX2N/ FX2NC						
LD	~	~	~	~	~	~						
LDI	~	~	~	~	~	~						
LDP				~	~	~						
LDF				~	~	~						
AND	~	~	~	~	~	~						
ANI	~	~	~	~	~	~						
ANP				~	~	~						
ANF				~	~	~						
OR	~	~	~	~	~	~						
ORI	~	~	~	~	~	~						
ORP				~	~	~						
ORF				~	~	~						
OUT	~	~	~	~	~	~						
SET	~	~	~	~	~	~						
RST	~	~	~	~	~	~						
MC	~	~	~	~	~	~						
MCR	~	~	~	~	~	~						
PLS	~	~	~	~	~	~						
PLF	~	~	~	~	~	~						
ANB	~	~	~	~	~	~						
INV				~	~	~						
ORB	~	~	~	~	~	~						
MPS	~	~	~	~	~	~						
MRD	~	~	~	~	~	~						
MPP	~	~	~	~	~	~						
NOP	~	~	~	~	~	~						
END	~	~	~	~	~	~						

Table B-1: Basic Program Instructions

B-2: SFC (STL) Program Instruction Table

Table B-2: SFC (STL) Program Instructions

Instruction (Mnemonic)	FXo/ FXos	FXON	FX/FX ₂ C	FX1S	FX1N	FX2N/ FX2NC
STL	~	~	~	~	~	~
RET	~	~	~	~	~	~

1
2
3
4
5
6
7
8
9
Α
В
С
D
Е



B-3: Applied Instruction Table



Note

Some instructions is supported differently by CPU version of an PLC. For further information about instruction of the FX₀, FX₀S, FX₀N, FX and FX₂C Series PLC can be found in FX Programming Manual (Manual number: JY992D48301), about the FX₁S, FX₁N, FX₂N and FX₂NC Series PLC can be found in FX Programming Manual II (Manual number: JY992D88101).

✓: Supports from first product

m

- ✓(□.□): The instruction is supported from "(□.□)" or later. "(□.□)" is CPU version of PLC on the Table A-3.
- $X(\square.\square)$: The instruction is not supported from "(□.□)" or later. "(□.□)" is CPU version of PLC on the Table A-3.

m

EV

	Instruction		F	X0/FX	0S	FXon				FX 2.30 earlie		FX(late	(V3.07 er)/F)	7 or K2C		FX1S			FX1N		FX2N/FX2NC			
	FNC No.	Mnemonic	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	
	000	CJ		~			~		~	~		~	>		~	>		~	>		~	<		
	001	CALL							~	~		~	>		~	>		~	>		~	<		
	002	SRET								<			~		~	>		~	>		~	<		
Ň	003	IRET		~			~			~			~			~			~			~		
E L	004	EI		~			~			~			~			~			~			~		
Program Flow	005	DI		~			~			<			~			>			>			<		
Pro	006	FEND		~			~			~			~			~			~			~		
	007	WDT		~			~		~	~		~	~		~	~		~	~		~	~		
	008	FOR		~			~			~			~			~			~			<		
	009	NEXT		~			~			~			>			~			~			~		
	010	CMP		~	~		~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	
	011	ZCP		~	~		~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	
ē	012	MOV		~	~		~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	
npa	013	SMOV							~	~		~	~								~	~		
Cor	014	CML							~	~	~	~	~	~							~	~	~	
Move And Compare	015	BMOV					~		~	~		~	~		~	~		~	~		~	~		
ove ,	016	FMOV							~	~		~	~	~							~	~	~	
Ĭ	017	XCH							~	~	~	~	~	~							~	~	~	
	018	BCD		~	~		~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	
	019	BIN		~	~		~	~	~	~	~	~	~	~	~	>	~	~	>	~	~	~	~	

Table B-3: Applied Instructions

Table B-3: Applied Instructions

	Instruction		F	FX0/FX0S			FXON	I	•	FX 2.30 arlie			(V3.07 er)/F)			FX1S		FX1N			FX	2N/FX	2NC	1
	FNC No.	Mnemonic	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	2
	020	ADD	<u>م</u>	-	۲ 3	4	-	3	۵ ۲	-	3	⊿	-	33	⊿	-	۲ 3	۵ ۲	-	۲ 3	4 V	-	3	3
ions	021	SUB		~	~		~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	
oerat	022	MUL		~	~		~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	
al O _l	023	DIV		~	~		~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	4
ogic	024	INC		~	~		~	~	~	~	~	~	~	~	~	~	~	~	~	>	~	~	~	
nd L	025	DEC		~	~		~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	-
Arithmetic And Logical Operations	026	WAND		~	~		~	~	~	~	~	~	~	~	~	v	~	~	~	~	~	~	~	5
hmet	027	WOR		~	~		~	~	~	~	~	~	~	~	~	v	~	~ ~	~	~	v	~	~	
Arith	028 029	WXOR NEG		~	~		~	~	~ ~	~ ~	~ ~	~ ~	~ ~	ン ン	~	~	~	~	~	~	~ ~	~ ~	~ ~	
	029	ROR							~	~	~	~	~	~		-					~	~	-	6
	031	ROL							~	~	~	~	~	~							~	~		
	032	RCR							~	~	~	~	~	~							· ·	~		-
Rotation And Shift	033	RCL							~	~	~	~	~	~							~	~		7
And 9	034	SFTR		~			~		~	~		~	~		~	~		~	~		~	~		
ion 4	035	SFTL		~			~		~	~		~	~		~	~		~	~		~	~		•
otat	036	WSFR							~	~		~	~								~	~		8
E C	037	WSFL							~	~		~	~								~	~		
	038	SFWR							~	>		~	~		~	~		~	~		~	~		•
	039	SFRD							~	~		~	~		~	~		~	~		~	~		9
	040	ZRST		~			~		~	~		~	~		~	~		~	~		~	~		
	041	DECO		~			~		~	~		~	~		~	~		~	~		~	~		•
c	042	ENCO		~			~		~	~		~	~		~	~		~	~		~	~		Α
ration	043 044	SUM BON							~	~	~	~	レ レ	~							~	~ ~	~	
Opei	044	MEAN							~	~	~	~	~	v v							~	~	~	-
Data Oper	045	ANS					-		ľ	~	-		~	-		-						~	-	В
	047	ANR							~	~		~	~								~	~		
	048	SQR										~	~	~		-					~	~	~	~
	049	FLT										~	~	~							~	~	~	С
	050	REF		~			~		~	~		~	~		~	~		~	~		~	~		
	051	REFF							~	~		~	~								~	~		Ĺ
sing	052	MTR								~			~			~			~			~		D
ces	053	HSCS			~			~			~			~			~			~			~	
High Speed Processing	054	HSCR			~			~			~			~			~			2			~	_
peed	055	HSZ									~			~									~	Ε
th S _t	056	SPD					<u> </u>			~			~			~			~			~		
Hig	057	PLSY		~	~		~	~		~	~		~	~		~	~		~	~		~	~	
	058	PWM		~			~			~			~			v			~			v		
	059	PLSR														~	~		~	~		~	~	

Table B-3: Applied Instructions

	Instruction		F)	Ko/FX	los		FXON			FX 2.30 arlie		FX(lat	(V3.0) er)/F)	7 or X _{2C}		FX1S			FX1N	l	FX2N/FX2NC			
	FNC No.	Mnemonic	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	
	060	IST		~			~			~			~			~			~			~		
	061	SER							~	~	~	~	~	~							~	~	~	
s	062	ABSD								~			~	~		~	~		~	~		~	~	
tion	063	INCD								~			~			~			~			~		
struc	064	TTMR								~			~									~		
/ Ins	065	STMR								~			~									~		
Handy Instructions	066	ALT		~			~		~	~		~	~		~	~		~	~		~	~		
Ĥ	067	RAMP		~			~			~			~			~			~			~		
	068	ROTC								~			~									~		
	069	SORT											~									~		
	070	ТКҮ								~	~		~	~								~	~	
	071	HKY								~	~		~	~								~	~	
ces	072	DSW								~			~			~			~			~		
External FX I/O Devices	073	SEGD							~	~		~	~								~	~		
0	074	SEGL								~			~			~			~			~		
ΕX	075	ARWS								~			~									~		
nal	076	ASC								~			~									~		
Exter	077	PR								~			~									~		
ш	078	FROM					~	~	~	~	~	~	~	~				V	~	~	~	~	~	
	079	TO					~	~	~	~	~	~	~	~				V	~	~	~	~	~	
	080	RS					✓ (1.2)			~			~			~			~			~		
s	081	PRUN							~	~	~	~	~	~	~	~	~	~	~	~	~	~	~	
Devices	082	ASCI					✓ (1.2)					~	~		~	~		~	~		~	~		
	083	HEX					✔ (1.2)					~	~		~	~		~	~		~	~		
External FX Serial	084	CCD					✔ (1.2)					~	~		~	~		~	~		~	~		
xter	085	VRRD							~	~		~	~		~	~		~	~		~	~		
Ш	086	VRSC							~	~		~	~		~	~		~	~		~	~		
	088	PID											✔ (3.3)			~			~			~		

Table B-3: Applied Instructions

	Instruc	tion	F	Ko/FX	los		FXON	l		FX 2.30 arliei		FX(lat	(V3.07 er)/F)	7 or (2C		FX1s			FX1N	I	FX	2N/FX	2NC		1
	FNC No.	Mnemonic	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation		2
	090	MNET	Pu	16	32	Pu	16	32	► Pu	16	32	n × (3.3)	9 X (3.3)	32	Pu	16	32	Pu	16	32	Pu	16	32		3
	091	ANRD							~	~		x	×												
S	092	ANWR							~	~		(3.3) X	(3.3) X												4
Unit		RMST										(3.3)	(3.3)												
I F2	093 094	RMWR							~	~ ~	~	~	~ ~	~											5
External F2 Units	094	RMRD							~	~	~	~	~	~											5
Ext	096	RMMN							~	· ·	•	~	~	·											
	097	BLK							~	V		X (3.3)	X (3.3)												6
	098	MCDE							~	~		X (3.3)	X (3.3)												
	110	ECMP																			~		~		7
	111	EZCP																			~		~		
2	118	EBCD																			~		~	l r	
anc	119	EBIN																			~		~		8
oint1	120	EADD																			~		~		•
g Pc	121	ESUB																			~		~	r	
Floating Point1 and 2	122	EMUL																			~		~		9
Ē	123	EDIV																			~		~		U
	127	ESQR																			~		~		
	129	INT																			~		~		Α
ometry	130	SIN																			~		~		$\boldsymbol{\Lambda}$
mom	131	COS																			~		~		
Trigond	132	TAN																			~		~		В
Data Operations 2	147	SWAP																			~	~	~		С
Positioning Control	155	ABS															~			~			✔ (3.0)		D
g Cc	156	ZRN														~	~		~	~					
ninc	157	PLSV														~	~		~	~				[-
ositic	158	DRVI														~	~		~	<					Е
ط	159	DRVA														~	~		~	~					

Table B-3:	Applied Instructions

	Instruc	tion		Ko/FX			FXON			FX 2.30 arlie		FX(V3.07 er)/F)	7 or K2C		FX1S			FX1N		FX	n∕FX	2NC
	FNC No.	Mnemonic	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation	Pulse Operation	16bit Operation	32bit Operation
_	160	TCMP													~	~		~	~		~	~	
ntro	161	TZCP													~	~		~	~		~	~	
Real Time Clock Control	162	TADD													~	~		~	~		~	~	
Clock	163	TSUB													~	~		~	~		~	~	
ne (166	TRD													~	~		~	~		~	~	
l Tir	167	TWR													~	>		~	~		~	~	
Rea	169	HOUR														>	5		~	2		✔ (3.0)	✔ (3.0)
	170	GRY																			~	~	~
es	171	GBIN																			~	~	~
Gray Codes	176	RD3A																~	~		✔ (3.0)	✔ (3.0)	
G	177	WR3A																~	~		✔ (3.0)	✔ (3.0)	
Additional Functions	180	EXTR																			v (3.0)	✔ (3.0)	
	224	LD=														>	~		~	<		<	~
	225	LD>														~	~		~	~		~	~
	226	LD<														~	~		~	~		~	~
	228	LD⇔														~	~		~	~		~	~
	229	LD≤														~	~		~	~		~	~
	230	LD≥														~	~		~	<		<	~
S	232	AND=														~	~		~	~		~	~
isor	233	AND>														~	~		~	~		~	~
Inline Comparisons	234	AND<														>	~		~	~		~	~
Cor	236	AND<>														~	~		~	~		~	~
line	237	AND≤														~	~		~	<		<	~
<u> </u>	238	AND≥														>	~		~	~		<	~
	240	OR=														~	~		~	<		<	~
	241	OR>														>	~		~	٢		<	~
	242	OR<														>	~		~	~		<	~
	244	OR<>														>	~		~	~		~	~
	245	OR≤														~	~		~	<		<	~
	246	OR≥														~	~		~	~		~	~

2

3

4

5

6

7

8

9

Α

В

С

D

Ε

Appendix C: Error Cord of PLC

For further information about error cord of the FX₀, FX₀s, FX₀N, FX and FX₂C Series PLC can be found in the each series Hardware Manual and FX Programming Manual (Manual number: JY992D48301), about the FX₁s, FX₁N, FX₂N and FX₂NC Series PLC can be found in the each series Hardware Manual and FX Programming Manual II (Manual number: JY992D88101).

Error Detection Device	Stored Error Number	Associated Meaning	Action			
	0000	No error				
	6101	RAM error	Check the cable connection between the extension unit/block and the PLC			
	6102	Operation circuit error				
D8061	6103	I/O bus error (M8069 = ON)				
PLC Hardware error	6104	Extension unit 24V failure (M8069=ON)				
	6105	Watch Dog Timer error	Scan time has exceeded the WDT time value set in D8000. Check user program.			

Table C-1: PLC Hatdware Error

Table C-2: PC/HPP Communication Error

Error Detection Device	Stored Error Number	Associated Meaning	Action
	0000	No error	
D8062	6201	Parity/ overrun/ framing error	Check the cable
PC/HPP communication	6202	Communications character error	connection between the
error	6203	Communication data sum check error	programming device and
(Not FX1s, FX1N)	6204	Data format error	the PLC
	6205	Command error	

Table C-3: Serial Communication Error

Error Detection Device	Stored Error Number	Associated Meaning	Note				
	0000	No error					
	6301	Parity/ overrun/ framing error					
	6302	Comms character error	Check communication				
	6303	Comms data sum check error	settings, parameters and				
D8063	6304	Comms data format error	applicable devices. (Computer link, N:N				
Serial communication errors	6305	Command error Computer link - received command other than GW (global) when station number was FF	network, Parallel link etc.) Refer to FX Communication Users				
	6306	Watchdog timer error	Manual for wiring				
	6312	Parallel link character error	techniques				
	6313	Parallel link data sum check error	1				
	6314	Parallel link data format error					



Error Detection Device	Stored Error Number	Associated Meaning	Action			
	0000	No error				
	6401	Program sum check error				
Dooo (6402	Memory capacity setting error	STOP the PLC, check			
D8064 Parameter	6403	Latched device area setting error	parameter, if incorrect			
error	6404	Comment area setting error	change to a suitable			
	6405	File register area setting error	value			
	6406 - 6408	Reserved				
	6409	Other setting error				

Table C-4: Parameter Error

Table C-5: Syntax Error

Error Detection Device	Stored Error Number	Associated Meaning	Action
	0000	No error	
	6501	Incorrect instruction/ device symbol/ device number combination	
	6502	No timer or counter coil before setting value	
	6503	 No setting value following either a timer or a counter coil Insufficient number of operands for an applied instruction 	
D8065 Syntax error	6504	 The same label number is used more than once The same interrupt input or high speed counter input is used more than once 	During programming, each instruction is checked as it is entered. If a syntax error is
,	6505	Device number is outside the allowable range	detected, re-enter the instruction correctly
	6506	Invalid applied instruction	
	6507	Invalid Pointer device [P] assignment for Jump or Call instruction	
	6508	Invalid Interrupt pointer device [I] assignment	
	6509	Other error	
	6510	MC nesting (N) number error	
	6511	The same interrupt input or high speed counter input is used more than once	

Table C-6: Cil Error Detection	<i>cuit Error</i> Stored Error			1
Device	Number	Associated Meaning	Action	
	0000	No error		
	6601	LD and LDI is used continuously 9 or more times in succession		2
		 No LD/ LDI instruction. The use of LD/LDI or ANB/ORB instruction is incorrect. The following instructions are not proposed to the active by a line. 		3
	6602	connected to the active bus line: STL, RET, MCR, (P)ointer, (I)nterrupt, EI, DI, SRET, IRET, FOR, NEXT, FEND and END 3) When MPP is missing		4
	6603	MPS is used continuously more than 12 times	-	5
	6604	The use of MPS, MRD, MPP instruction is incorrect.		
		 The STL instruction is continuously used 9 times or more MOD instruction (Untermuted) 	A circuit error occurs if a	6
D8066 Circuit error	6605	MC, MCR instruction, (I)nterrupt pointer or SRET instruction is used within an STL program area RET has not been used in the	combination of instructions is incorrect or badly specified.	7
		program or is not connected to an STL instruction	Select programming mode and correct the identified error.	8
		 No (P)ointer, (I)nterrupt pointer No SRET/ IRET 		
	6606	 An (I)nterrupt pointer, SRET or IRET has been used within the main program 		9
		 STL, RET, MC or MCR have been used within either a subroutine or an interrupt routine 		Α
	6607	 The use of FOR and NEXT is incorrect The following instructions have been used within a FOR -NEXT loop: STL, RET, MC, MCR, IRET, SRET, FEND or END 		В
	6608	 The use of MC/ MCR is incorrect Missing MCR N0 SRET, IRET instruction or an 		С
		(I)nterrupt pointer has been used within an MC/ MCR instruction area		D
	6609	Other error		

Table C-6: Circuit Error

Е

Error Detection Device	Stored Error Number	Associated Meaning	Action
201100		LD, LDI is used continuously 9 or more	
	6610	times in succession	
	6611	Number of LD/LDI instructions is more than ANB/ORB instructions	
	6612	Number of LD/LDI instructions is less than ANB/ORB instructions	
	6613	MPS is used continuously more than 12 times	
	6614	MPS instruction missing	
	6515	MPP instruction missing	
	6616	Unauthorized use of the MPS/ MRD/ MPP instructions; possible coil missing	
	6617	One of the following instructions is not connected to the active bus line: STL, RET, MCR, (P)ointer, (I)nterrupt pointer, EI, DI, SRET, IRET, FOR, NEXT, FEND and END	
	6618	STL, RET, MC or MCR programmed within either a subroutine or an interrupt routine	
D8066 Circuit error	6619	Invalid instruction programmed within a FOR - NEXT loop: STL, RET, MC, MCR, (I)nterrupt pointer, IRET and SRET	A circuit error occurs if a combination of instructions is incorrect or badly specified.
	6620	FOR - NEXT instruction nesting levels (5) exceeded	Select programming mode and correct the
	6621	The number of FOR and NEXT instructions does not match	identified error.
	6622	NEXT instruction not found	
	6623	MC instruction not found	
	6624	MCR instruction not found	
	6625	The STL instruction is continually used 9 times or more	
	6626	Invalid instruction programmed within an STL - RET program area: MC, MCR, (I)nterrupt pointer, IRET and SRET	
	6627	RET instruction not found	-
	6628	(I)nterrupt pointer, SRET and IRET incorrectly programmed within main program	
	6629	(P)ointer or (I)nterrupt pointer label not found]
	6630	SRET or IRET not found]
	6631	SRET programmed in invalid location]
	6632	IRET programmed in invalid location	

Table C-6: Circuit Error

Error Detection Device	Stored Error Number	Associated Meaning	Action	1
	0000	No error		
	6701	 No jump destination (pointer) for CJ or CALL instructions (P)ointer is designated in a block that comes after the END instruction An independent label is designated in 	These error occur during the execution of an operation.	2
		a FOR-NEXT loop or a subroutine 6 or more CALL instruction nesting levels	When an operation error occurs, STOP the PLC enter programming code	3
	6702	have been used	and correct the fault. Note: operation errors	
D8067	6703	3 or more interrupt nesting levels have been used	can occur even when the syntax or circuit design	4
Operation error	6704	6 or more FOR - NEXT instruction nesting levels have been used	is correct, e.g. D500Z is a valid	5
	6705	An incompatible device has been specified as an operand for an applied instruction	statement within an FX1N PLC. But if Z had a	
	6706	A device has been specified outside of the allowable range for an applied instruction operand	value of 10000, the data register D10500 would be attempted to be accessed. This will	6
	6707	A file register has been accessed which is outside of the users specified range	cause an operation error as there is no D10500	7
	6708	FROM/ TO instruction error	device available.	
	6709	Other error, i.e. missing IRE/ SRET, unauthorized FOR - NEXT relationship		8
	6730	Sampling time Ts (Ts<0 or >32767)	The identified parameter	
	6732	Input filter value α (α <0 or >=101)	is specified outside of its	
	6733	Proportional gain KP (KP<0 or >32767)	allowable range	9
	6734	Integral time constant TI (TI<0 or >32767)	Execution ceases PID instruction must be reset	
	6735	Derivative gain KD (KD<0 or >=101)	before execution will	
	6736	Derivative time constant TD (TD<0 or >32767)	resume	A
	6740	Sampling time TS is less than the program scan time.	TS is set to program scan time - Execution will continue.	E
D8067	6742	Current value Δ exceeds its limits	Data affected resets to	
PID	6743	Calculated error ε exceeds its limits	the nearest limit value. For all errors except	C
Operation error	6744	Integral result exceeds its limits	6745, this will either be a	C
51101	6745	Derivative gain over, or differential value exceeds allowable range	minimum of -32768 or a maximum of +32767.	
	6746	Derivative result exceeds its limits	Execution will continue,	
	6747	Total PID result exceeds its limits	but user should reset PID instruction.	
	6750	SV - PV _{nf} < 150, or system is unstable (SV - PV _{nf} has wide, fast variations)	The error fluctuation is outside the normal	E
	6751	Large Overshoot of the Set Value	operation limits for the PID instruction.	
	6752	Large fluctuations during Autotuning Set Process	Execution ceases. PID instruction must be reset.	

 Table C-7:
 Operation Error and PID Operation Error



MEMO

2

3

4

5

6

7

8

9

Α

В

С

D

Ε

Appendix D: File converter

By using this file converter, a MELSEC MEDOC file (.TMP) can be converted into an FXGP/ WIN-E file.

- SWOF	C-FXGP/WIN-E(File converter)		
<u>F</u> ile ⊻iew <u>H</u> elp			
Conversion Process	MELSEC MEDOC> FXGPWIN 보	Options	Selection of PLC
Source File name	*.tmp	<u>B</u> rowse	
Destination File name	*.pm w	B <u>r</u> owse	
	Convert		
	PLC:FX2N		

Listings for printout

- 1) "INSTR" and "PARAMETER" must always be selected in the printout menu of MELSEC MEDOC.
- 2) When the device names and the device comments should be converted, "NAME" must be selected in the printout menu of MELSEC MEDOC.
- 3) When the circuit comment is converted, "ONLY COM" must be selected in the printout menu of MELSEC MEDOC.
- 4) When the value of the data register is converted, "DWR" must selected in the printout menu of MELSEC MEDOC.

File name

- 1) File names up to eight characters are allowed. (Do not use the long file names of WINDOWS95)
- 2) The extension of the text file for MELSEC MEDOC should to be "TMP".

Limitation and rule of the data when converting.

- Device name conversion NAME which exceeds eight characters are cut to eight characters. Characters other than alphabet characters,numbers or the effective signs stated below are replaced with '_'(under bar).
 (Effective sign: +-*/=.? #\$%&: ; _)
- 2) Circuit comment

Up to 500 characters in total are converted when there are two or more comment lines for one circuit block.

3) Value of data register

When two or more sets of SETUP data of the data, link and file registers (DWR) are converted, only the last SETUP data can be converted.



MEMO

2

Appendix E: Associated Manuals

Further information on Hardware (Specification, configuration, Installation and wiring) programming, optional modules or special function units/blocks can be found in each respective manual.

Table	E-1: Associated M			,
	Manual name	Manual No.	Description	
	FX2N Series Programmable Controllers Hardware Manual	JY992D66301	This manual contains hardware explanations for wiring, installation and	
	FX2NC (DSS/DS) Series Programmable Controllers Hardware Manual	JY992D76401		
	FX2NC (D/UL) Series Programmable Controllers Hardware Manual	JY992D87201		
	FX1N Series Programmable Controllers Hardware Manual	JY992D89301		
	FX1s Series Programmable Controllers Hardware Manual	JY992D83901	specification, etc.	
	FX Series Programmable Controllers Hardware Manual (FX/FX2c)	JY992D47401		
Main	FX0/FX0N Series Programmable Controllers Hardware Manual	JY992D47501		
	FXos Series Programmable Controllers Hardware Manual	JY992D55301		
	FX Programming Manual II (for FX1S, FX1N, FX2N, FX2NC Series PLC)	JY992D88101	This manual contains instruction explanations.	
	FX Programming Manual II (for FX0, FX0s, FX0N, FX, FX2c Series PLC)	JY992D48301		
	FX1N-4EX-BD Input Expansion Board User's Manual	JY992D95001	This manual contains hardware explanations for wiring, installation and specification, etc.	
	FX1N-2EYT-BD Input Expansion Board User's Manual	JY992D95201		
	FX I/O Terminal Block User's Guide	JY992D560401		
Analog	FX1N-1DA-BD Analog Output Expansion Board User's Guide	JY992D96401	This manual contains explanations for wiring, installation and specification.	
	FX1N-2AD-BD Analog Input Expansion Board User's Guide	JY992D96201		
	FX2N-2DA Special Function Block User's Guide	JY992D74901	This manual contains explanations for wiring, installation, specification and BFM allocation, etc.	
	FX2N-2AD Special Function Block User's Guide	JY992D74701		

Table E-1: Associated Manuals



Table	E-1: Associated M		
	Manual name	Manual No.	Description
	FXoN-3A Special Function Block User's Guide FX2N-4DA	JY992D49001	
	Special Function Block User's Guide	JY992D65901	
	FX2N-4AD Special Function Block User's Guide	JY992D65201	
	FX2N-4AD-PT Special Function Block User's Guide	JY992D65601	
	FX2N-4AD-TC Special Function Block User's Guide	JY992D65501	
Analog	FX2N-8AD Special Function Block User's Manual	JY992D86001	This manual contains explanations for wiring, installation, specification
Ani	FX-2DA Special Function Block User's Guide	JY992D52801	and BFM allocation, etc.
	FX-4DA Special Function Block User's Guide	JY992D61001	
	FX-4AD Special Function Block User's Guide	JY992D52601	
	FX-2AD-PT Special Function Block User's Guide	JY992D55701	
	FX-4AD-TC Special Function Block User's Guide	JY992D55901	
	FX2N-2LC Special Function Block User's Manual	JY992D85801	
Counter	FX2N-1HC Special Function Block User's Guide	JY992D65401	This manual contains explanations for wiring, installation, specification
High Speed	FX-1HC Special Function Block User's Guide	JY992D53001	and BFM allocation, etc.
	FX-1PG-E/FX2N-1PG-E Pulse Generation Unit User's Manual	JY992D65301	
g Control	FX2N-10PG Pulse Generation Unit User's Manual	JY992D93401	This manual contains explanations for wiring, installation, specification
Positioning Control	FX2N-10GM, FX2N-20GM Hardware/Programming Manual	JY992D77801	and BFM allocation, etc.
	FX-10GM, FX-20GM Hardware/Programming Manual	JY992D60401	

 Table E-1:
 Associated Manuals

 Table E-1:
 Associated Manuals

	Manual name	Manual No.	Description	4
1 Switch	FX2N-1RM-E-SET Programmable Cam Switch User's Manual		This manual contains explanations for wiring, installation, specification and BFM allocation, etc.	
Programmable Cam Switch		JY992D71101		2
Progra				3
	FX _{2N} -232IF RS-232C Interface Block Hardware Manual	JY992D73501	This manual contains explanations for wiring, installation, specification and BFM allocation, etc.	4
	FX _{2N} -232-BD Communication Board User's Guide	JY992D66001		-
	FX _{2N} -422-BD Communication Board User's Guide	JY992D66101		5
	FX2N-485-BD Communication Board User's Guide	JY992D73401		6
	FX2NC-232ADP Installation Manual	JY997D01101	- This manual contains explanations for specification	L
	FX _{0N} -232ADP Communication Adapter User's Guide	JY992D51301	This manual contains explanations for specification	7
ork	FX-232ADP Communication Adapter User's Guide	JY992D48801		8
Netwo	FX2NC-485ADP Installation Manual	JY997D01201		
ation and	FX/FX0N-485ADP Communication Adapter User's Guide	JY992D53201	1	9
Communication and Network	FX Communication (RS232C, RS485) User's Manual	JY992D69901	This manual contains explanations for N:N network, parallel link, computer link and no protocol communication (RS instruction and FX2N-232IF RS232C interface block) about FX family PLC.	A
O	FX2N-16CCL-M CC-Link System Master Block User's Manual	JY992D93101		B
	FX2N-32CCL CC-Link Interface Block User's Manual	JY992D71801		
	FXon-32NT-DP Profibus-DP Interface User's Manual	JY992D61401	This manual contains explanations for wiring, installation, specification and BFM allocation, etc.	C
	FX2N-64DNET DeviceNet Interface User's Manual	JY992D86301		D
	FX2N-32ASI-M AS-interface Master Block User's Manual	JY992D76901	1	E
	FX2N-16LNK-M MELSEC I/O Link System Master Block User's Manual	JY992D73701	This manual contains explanations for wiring, installation, specification setting and I/O allocation, etc.	

	Manual name	Manual No.	Description
Vetwork	FX0N-16NT MELSEC NET-MINI (-S3) Interface User's Guide	JY992D48901	This manual contains explanations for wiring, installation and specification.
ation and	FX-16NT/NP MELSEC NET-MINI (-S3) Interface User's Guide	JY992D56201	This manual contains explanations for wiring, installation, specification
Communication and Network	FX-16NT/NP-S3 MELSEC NET-MINI-S3 Interface User's Manual	JY992D34601	and BFM allocation, etc.
	FX2N-CNV-BD Manual	JY992D63601	
Others	FX1N-CNV-BD Manual	JY992D84701	This manual contains explanations for installation.
Oth	FX2N-8AV-BD Manual	JY992D62501	
	FX1N-8AV-BD Manual	JY992D84601	

Table E-1: Associated Manuals

Introduction	1
Installing FXGP/WIN-E	2
Main Operation	3
Work Windows and their Functions	4
Guidance of Basic Operations	5
Menu Bar Commands and their Functions	6
Creating Ladder and Instruction Programs	7
Creating SFC Programs	8
Remote Maintenance Using Telephone Line	9
Appendix A: Device Lists	Α
Appendix B: Instruction List	В
Appendix C: Error Code Tables	С
Appendix D: File Convertor	D
Appendix D: Associated Manuals	Е
	Installing FXGP/WIN-EMain OperationWork Windows and their FunctionsGuidance of Basic OperationsGuidance of Basic OperationsMenu Bar Commands and their FunctionsCreating Ladder and Instruction ProgramsCreating SFC ProgramsRemote Maintenance Using Telephone LineAppendix A: Device ListsAppendix B: Instruction ListAppendix C: Error Code TablesAppendix D: File Convertor

1	Introduction
2	Installing FXGP/WIN-E
3	Main Operation
4	Work Windows and their Functions
5	Guidance of Basic Operations
6	Menu Bar Commands and their Functions
7	Creating Ladder and Instruction Programs
8	Creating SFC Programs
9	Remote Maintenance Using Telephone Line
Α	Appendix A: Device Lists
В	Appendix B: Instruction List
С	Appendix C: Error Code Tables
D	Appendix D: File Convertor
E	Appendix D: Associated Manuals



SOFTWARE MANUAL FX-PCS/WIN-E



HEAD OFFICE: TOKYO BUILDING, 2-7-3 MARUNOUCHI, CHIYODA-KU, TOKYO 100-8310, JAPAN HIMEJI WORKS: 840, CHIYODA CHO, HIMEJI, JAPAN

MODEL	FX-PCS/WIN-SW-E
MODEL CODE	09R908